

## Best Available Copy

Printed in the United States of America. Available from National Technical Information Service U.S. Department of Commerce 5285 Port Royal Road, Springfield, Virginia 22161 Price: Printed Copy \$8.00; Microfiche \$2.25

This report was prepared as an account of work sponsored by the United States Government. Neither the United States nor the Energy Research and Development Administration/United States Nuclear Regulatory Commission, nor any of their employees, nor any of their contractors, subcontractors, or their employees, makes any warranty, express or implied, or assumes any legal liability or responsibility for the accuracy, completeness or usefulness of any information, apparatus, product or process disclosed, or represents that its use would not infringe privately owned rights. And a second second

τ

ORNL-5041 UC-41 - Health and Safety

Contract No. W-7405-eng-26

4

;

HEALTH PHYSICS DIVISION Emergency Technology Section

#### SUMMARY

#### SURVIVAL OF THE RELOCATED POPULATION OF THE U.S. AFTER A NUCLEAR ATTACK

Final Report

Ъy

Carsten M. Haaland Conrad V. Chester Eugene P. Wigner

for

Defense Civil Preparedness Agency Washington, D. C. 20301

Interagency Agreement AEC 40-31-64 and DCPA 01-74-C-0227, Work Unit 3539A

#### DCPA Review Notice

This report has been reviewed in the Defense Civil Preparedness Agency and approved for publication. Approval does not signify that the contents necessarily reflect the views and policies of the Defense Civil Preparedness Agency.

APPROVED FOR PUBLIC RELEASE; DISTRIBUTION UNLIMITED

OAK RIDGE NATIONAL LABORATORY Oak Ridge, Tennessee 37830 operated by UNION CARBIDE CORPORATION for the ENERGY RESEARCH AND DEVELOPMENT ADMINISTRATION

#### SURVIVAL OF THE RELOCATED POPULATION OF THE U.S. AFTER A NUCLEAR ATTACK

C. M. Haaland, C. V. Chester, and E. P. Wigner

#### SUMMARY

The feasibility of continued survival after a hypothetical nuclear attack is evaluated for people relocated from high-risk areas during the crisis period before the attack. The attack consists of 6559 MT, of which 5951 MT are ground bursts, on military, industrial, and urban targets. Relocated people are assumed to be adequately protected from fallout radiation by shelters of various kinds. The major problems in the postattack situation will be the control of exposure to fallout radiation, and prevention of severe food shortages to several tens of millions of people. A reserve of several million additional dosimeters is recommended to provide control of radiation exposure. Written instructions should be provided with each on their use and the evaluation of the hazard. Adequate food reserve exists in the U.S. in the form of grain stocks, but a vigorous shipping program would have to be initiated within two or three weeks after the attack to avoid large scale starvation in some areas. If the attack occurred in June when crops on the average are the most vulnerable to fallout radiation, the crop yield could be reduced by about one-third to one-half, and the effects on crops of possible increased ultraviolet radiation resulting from ozone layer depletion by nuclear detonations may further increase the loss. About 80% of the U.S. crude refining capacity and nearly all oil pipelines would be either destroyed or inoperative during the first several weeks after an attack. However, a few billion gallons of diesel fuel and gasoline would survive in tank storage throughout the country, more than enough for trains and trucks to accomplish the grain shipments required for survival. Results of a computer program to minimize the ton-miles of shipments of grain between Business Economic Areas (BEAs) indicate that less than 2% of the 1970 rail shipping capacity, or less than 6% of the 1970 truck shipping capacity would be adequate to carry out the necessary grain shipments. The continuity of a strong federal government throughout the attack and postattack period is essential to coordinate the wide-scale interstate survival activities.

ORNL-5,041 Health and Safety

Contract No. W-7405-eng-26 HEALTH PHYSICS DIVISION Emergency Technology Section



for

Defense Civil Preparedness Agency Washington, D. C. 20301

Interagency Agreement AEC 40-31-64 and DCPA 01-74-C-0227, Work Unit 3539A

#### DCPA Review Notice

This report has been reviewed in the Defense Civil Preparedness Agency and approved for publication. Approval does not signify that the contents necessarily reflect the views and policies of the Defense Civil Preparedness Agency.

APPROVED FOR PUBLIC RELEASE; DISTRIBUTION UNLIMITED ø JUNE: 97 OAK RIDGE NATIONAL LABORATORY

263050

Oak Ridge, Tennessee 37830 operated by UNION CARBIDE CORPORATION for the ENERGY RESEARCH AND DEVELOPMENT ADMINISTRATION

#### ACKNOWLEDGMENTS

The authors are grateful to Dr. David Bensen, COTR<sup>\*</sup> from DCPA, for guidance and valuable assistance during the course of this project; to T. J. Byram, Statistical Reporting Service, USDA, for data on grain production, particularly for the computer tape on U.S. 1973 grain production by county; to Henry Dickson, National Flight Data Center, Federal Aviation Administration, Department of Transportation, for a copy of the Airport Data Base on computer tape; to Phillip Coleman for expeditious preparation of computer graphics and computations; to Sarah A. Brown for the compilation and presentation of data on airports; to R. R. Davis for data on airlift capabilities of commercial aircraft; to Marjorie Fish for obtaining and graphing data on grain storage; and to Gary Westley for preparation of the linear programming for minimizing the transport of grain.



"A Glossary of Acronyms for this report is given on page xi.

CONTENTS

v

**n** - - -

. .

and the second second second

......

	rage
ACKNOWLEDGMENTS	iii
LIST OF FIGURES	vii
LIST OF TABLES	ix
GLOSSARY OF ACRONYMS	xi
ABSTRACT	1
1. INTRODUCTION	3
2. SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS	7
2.1 Summary	7
2.2 Conclusions	11
2.3 Recommendations	12
3. THE ATTACK SCENARIO	17
3.1 Introduction	17
3.2 The Attack	20
3.3 Crisis Relocation	24
4. RADIOLOGICAL HAZARDS AND DEFENSES	37
4.1 Introduction	37
4.2 Fallcut from the CRP-2B Attack	37
4.3 The U.S. RADEF Program	52
4.4 Radiological Exposure Control Guidelines	61
4.5 The Basis for Radiological Exposure Control	
	81
4.6 Shelter Survival Conditions in a Hazardous Radiation Field	. 89
5. COMMUNICATIONS	95
5.1 AM Radio Broadcast	95
5.2 Two-way Communications	96
6. THE FOOD AND WATES SITUATION	. 105
6.1 Water Supply	105
6.2 The Food Situation	105
6.3 The Effects of Dust and Depletion of the Ozone	100
Layer Due to Nuclear Attack	129

1

### PRECEDING PAGE BLANK-NOT FILMED

7.	TRANS	SPORTATION FOR POSTATTACK SURVIVAL	133
	7.1	Introduction	133
	7.2	Oil Pipelines	133
	7.3	Inland Waterways	137
	7.4	Railroads	138
	7.5	Commercial Aircraft and Airports	139
	7.6	Trucks	142
	7.7	Postattack Shipment of Grain Stocks	145
8.	PETRO	DLEUM	153
9.	THE 3	MEDICAL LOAD	157
10.	GOVER	RYMENT AND THE ECONOMY	161
ANN	OTATEI	D BIBLIOGRAPHY	163
APP	ENDIX	A. TRANSPORT BY COMMERCIAL AIRLINES	181
	A.1	Introduction	181
	A.2	Existing Programs	181
	A. 3	Maintenance and Material Requirements	190
	A.4	Description of Aircraft	191
	A.5	Alternate Airfields	191
APP	ENDIX	E. MINIMIZATION OF TRANSPORT OF GRAIN	195
APP	ENDIX	C. SURVIVAL OF PETROLEUM REFINERIES	199

vi

#### LIST OF FIGURES

Page

3.1	CRP-2B Attack Pattern on the U.S	22
3.2	Number of People Exposed to Blast	23
3.3	Two-psi Blast Circles on the Northeastern U.S	25
3.4	Fifteen-psi Blast Circles on the Northeastern U.S.	26
3.5	Relocated Population, Northeast	29
3.6	Relocated Population, Central and East	30
3.7	Relocated Population, Southeast	31
3.8	Relocated Population, Midwest	32
3.9	Relocated Population, South Central	33
3.10	Relocated Population, Northwest	34
3.11	Relocated Population, Southwest	35
4.1	CRP-2B Fallout, Unit-Time Reference Dose-Rates	38
4.2	Area of U.S. as a Function of Unit-Time Reference Dose-Rate	41
4.3	Four-Day Dose and Peak ERD as a Function of Arrival Time of Fallout	43
4.4	Percent of Population vs Unit-Time Reference Dose-Rate	47
4.5	Estimation of PF for Shelters at Edgemont, S.D	51
4.6	Fallout Radiation Levels One Year After	53
4.7	Entry Times into Contaminated Areas for $100 \le R_0 \le 1000$	74
4.3	Entry Times into Contaminated Areas for $1000 \le R_0 \le 10,000$	77
4.9	Entry Times as a Function of Protection Factor of Shelters	79
4.10	Acute Hematopoietic Syndrome as a Function of Dose	88
4.11	Life-Shortening vs Radiation Exposure	90
4.12	Incidence of Leukemia vs Radiation Exposure	91
5.1	Citizens Band and Amateur Radio Transmitters, 1960-1973	98
5.2	Industrial and Transportation Radio Transmitters, 1960-1973	99
5.3	Total Authorized Radio Transmitters, 1960-1973	100
6.1	Total Production of Corn, Wheat, and Soybeans as Percent of U.S. Population Minimum Survival Requirement	111

#### vii

T

viii

ないないのであるというないであるとうないのであるというで

in the state in the state in the state in the state of the

6.2	Corn, Wheat, and Soybeans Stored On and Off Farms as	
	Percent of Total Production	112
6.3	Corn, Wheat, and Soybean Stocks, January and April	113
6.4	Corn, Wheat, and Soybean Stocks, July and October	114
6.5	Wheat Production Exports and Price, 1910-1975	116
6.6	Grain Yield as a Function of Unit-Time Reference	
	Dose Rate	119
6.7	Grain Production, 1973, Northeast	122
6.8	Grain Production, 1973, Central and East	123
6.9	Grain Production, 1973, Southeast	124
6.10	Grain Production, 1973, North Central	125
6.11	Grain Production, 1973, South Central	126
6.12	Grain Production, 1973, Northwest	127
6.13	Grain Production, 1973, Southwest	128
7.1	Volume of Domestic Intercity Traffic	134
7.2	Commercial Airports in the U.S. with Runways 500 ft or	
	Longer	140
7.3	Distribution of Airports by Runway Lengths	141
7.4	Distribution of Fuel Storage Capacity	143
7.5	Cumulative Fuel Storage Capacity	144
7.6	Food-Days from Grain for Various Populations	147
7.7	Distribution of People, Relocated According to	
	ADAGIO, Shown by BEAs	148
7.8	Distribution of Production of Grain in 1973 by BEAs	149
8.1	Refineries in U.S	154
8.2	Illustrative Operating Conditions for the Trans- Alaska Pipeline	156

#### APPENDICES

A.1	Field Length Requirements for the L-1011 Lockheed TriStar Aircraft	187
A.2	Fuel Requirements for the L-1011 Lockheed TriStar Aircraft	188
A.3	Runway Pavement Thickness for L-1011 Lockheed TriStar	192
C.1	Survival of North American Refineries as Function of Number of Warheads and Reliability	201

#### LIST OF TABLES

2.1	Summary of Assessment of Postattack Survival of Sectors for Support for the Relocated Population	9
3.1	Summary of CRP-2B Attack	21
3.2	States with Relocation Hosting Factors Greater than Three. 2	28
4.1	Areas Covered by Fallout	iC
4.2	Reduction Factors for Radiation Intensity Due to Decay 4	12
4.3	Population in Areas Covered with Fallout	ie
4.4	Forty Counties with Heaviest Fallout	34
4.5	Area Dose-Rates One Year After A Hypothetical Attack 5	54
4.6	Radiation Detection Instruments Currently Distributed to States	57
4.7	Distribution of Monitoring Kits by States 5	59
4.8	Guidelines for Shelter and Operational Activities 6	53
4.9	The "Penalty" Table	56
4.10	Permissible Exposure Time in an Area Contaminated by Fallout Resulting from a Nuclear Blast (Soviet)6	59
4.11	Permissible Entry Times and Doses for Daily Exposure to Fallout Radiation, with No Medical Care Required According to the "Penalty" Table	72
4.12	Exposures in Shelters of Low PF	77
4.13	Environmental Radiation Protection Factors Provided by Civilian Vehicles	30
4.14	Accumulated Estimated Exposures for 50% Incidence of Physiological Symptoms	34
4.15	Some Clinical and Statistical Estimates of Human Total-Body Radiation Tolerance	36
5.1	Transmitters Authorized by FCC in 1973	)1
6.1	Distribution of the Food Dollar in Northeastern USA, 1965-66	)7
6.2	Estimating Surviving Yield of U.S. Crops 11	18
6.3	Total U.S. Grain Production for Three Different Sensitivities to Fallout Irradiation	2:
7.1	Number of Privately and Publicly Owned Transport Units 13	35
7.2	Postattack Grain Shipments to Deficit Food Areas 15	51

Page

の意識が変換したが、「ないない」ないのではない。ここに、ここに、ここに、ここになったいではないないないないないないないないないないではないのないないないないないないないないないないないないないないです。

tillitä 19580 storrädes verkäätällen einen kirkkäätäksi tehten einen on on on an an

ix

8.1 Minimum Operating Inventories, Petroleum Supplies . . . 155
9.1 Reported Cases of Spr ified Notifiable Diseases . . . . 158

#### APPENDIX

A.1	Annual Passenger Seat-Mile and Additional Cargo Ton-Mile Capacities of WASP Passenger Aircraft, Calendar Year 1975, International Fleet	182
A.2	Annual Passenger Seat-Mile and Additional Cargo Ton- Mile Capacities of WASP Passenger Aircraft, Calendar Year 1975, Domestic Fleet	184
A.3	Characteristics of Representative Aircraft in the Commercial Fleet	186
A.4	Performance Characteristics of Representative Aircraft	189

х

#### GLOSSARY OF ACRONYMS

BE	A	-	Business Iconomic Area
CA	Р	-	Civil Air Patrol
CB		-	Citizens Band
CE	P	-	Circular Error Probable
C0	TR		Contracting Officer's Technical Representative
CR	AF	-	Civil Reserve Air Fleet
CR	Р	-	Crisis Relocation Planning
CS	LWEMND	-	Committee to Study the Long-Term Worldwide Effects of Multiple Nuclear-Weapons Detonations
DC	PA	-	Defense Civil Preparedness Agency
EB	S	-	Emergency Broadcast System
EM	P	-	Electromagnetic Pulse (from nuclear detonations)
EO	С	-	Emergency Operating Center
ER	D	-	Equivalent Residual Dose
FA	A	-	Federal Aviation Administration
FC	C	-	Federal Communications Commission
FP	A	-	Federal Preparedness Agency
GS	A	-	General Services Administration
HO	В	-	Height of Burst
IC	BM	-	Intercontinental Ballistic Missile
LA	S	-	Leo A. Schmidt
LE	T	•••	Linear Energy Transfer
NA	S	-	National Academy of Science
NA	SA	-	Nationsi Aeronautics and Space Administration
NC	RP	-	National Council on Radiation Protection
OH	VM	-	Other High Value Military
OR	NL	-	Oak Ridge National Laboratory
PF		-	Protection Factor
RA	DEF	-	Radiological Defense
RA	M	-	Ralph A. Mason
RB	E	-	Relative Biological Effectiveness
RD	0	-	Radiological Defense Officer
RD	POG	-	Radiological Defense Planning & Operations Guide

の日本の時代のないなどのためを

i

	Reference Equivalent Space Exposure
RES	- Kelerence Equivalent offer ant
RFMSF	- Radiological Factors in Manned Space Flight
SAC	- Strategic Air Command
SLBM	- Submarine Launched Ballistic Missile
SSBN	- Ballistic Missile Submarine (nuclear)
UI	- Urban-Industrial
USDA	<ul> <li>United States Department of Agriculture</li> </ul>
WASP	- War Air Services Program
WSEG	- Weapons Systems Evaluation Group

and the second secon

AND IN A SUPPORT OF THE

÷

**EXAMPLE 1** 

xii

SALANDI LUZIA T

ስታል<u>ሽ ተኳስ አስተዋሪ አስተዋሪ የተቀላቁ</u>ት የአካላት የሚሰራ የአካሪ አስተለ የሚሰራት ዓመር በቸውን እንስታለት የአካሪ አስትት የሆኑ የመንግር የአካሪ ከተመሰራት የተቀላ እ

#### SURVIVAL OF THE RELOCATED POPULATION OF THE U.J. AFTER A NUCLEAR ATTACK

「たいかいないのないないないで、このようななないないないないと

C. M. Haaland, C. V. Chester, and E. P. Wigner

#### ABSTRACT

The feasibility of continued survival after a hypothetical nuclear attack is evaluated for people relocated from high-risk areas during the crisis period before the attack. The attack consists of 6559 MT, of which 5951 MT are ground bursts, on military, industrial, and urban targets. Relocated people are assumed to be adequately protected from fallout radiation by shelters of various kinds. The major problems in the postattack situation will be the control of exposure to fallout radiation, and prevention of severe food shortages to several tens of millions of people. A reserve of several million additional dosimeters is recommended to provide control of radiation exposure. Written instructions should be provided with each on their use and the evaluation of the hazard. Adequate food reserve exists in the U.S. in the form of grain stocks, but a vigorous shipping program would have to be initiated within two or three weeks after the attack to avoid large scale starvation in some areas. If the attack occurred in June when crops on the average are the most vulnerable to fallout radiation, the crop yield could be reduced by about one-third to one-half, and the effects on crops of possible increased ultraviolet radiation resulting from ozone layer depletion by nuclear detonations may furthe. increase the loss. About 80% of the U.S. crude refining capacity and nearly all oil pipelines would be either destroyed or inoperative during the first several weeks after an attack. However, a few billion gallons of diesel fuel and gasoline would survive in tank storage throughout the country, more than enough for trains and trucks to accomplish the grain shipments required for survival. Results of a computer program to minimize the ton-miles of shipments of grain between Business Economic Areas (BEAs) indicate that less than 2% of the 1970 rail shipping capacity, or less than 6% of the 1970 truck shipping capacity would be adequate to carry out the necessary grain shipments. The continuity of a strong federal government throughout the attack and postattack period is essential to coordinate the wide-scale interstate survival activities.

#### 1. INTRODUCTION

This report describes research performed at ORNL<sup>\*</sup> for DCPA Work Unit 3539A titled <u>Postattack Survival Planning</u>. The objective and scope of work for the study are quoted from the Task Order as follows:

"SCOPE: Given the assumptions that:

(1) At a time in the not too distant future an international nuclear crisis <u>has</u> occurred;

(2) That U.S. Crisis Relocation Plans in accordance with currently conceived elements <u>have</u> been implemented;

(3) That radiological protection <u>has</u> been provided and used, again according to currently conceived ideas; and

(4) That a nuclear attack on the U.S. of a magnitude within that considered consistent with current SALT weapons limitations has occurred.

"Define the nature and scope of plans for caring for the survivors of the attack, concentrating on those personnel who have been relocated into the host areas.

"The time of consideration for this research effort will be from about one day following initiation of the nuclear attack until the end of the so-called survival period--a few weeks to a few months later.

"This survival plan shall consider two possibilities:

(1) The attack occurs in the quite near future, therefore only today's resources (i.e., radiological instruments, current state of trained personnel, warning capabilities, and the like) exist; and

(2) The attack occurs after a few years of modest civil preparedness effort, and therefore today's resources could have been augmented accordingly.

"The cost/benefits of such augmentation shall be examined.

"The constituents of these postattack survival plans shall include, but not be limited to, the following:

(1) Radiological exposure control;

"A Glossary of Acronyms for this report begins on page xi.

(2) Housing, feeding, medical, and public health services;

(3) Organization and constitution of an emergency labor force;

(4) Expedient (and perhaps temporary) social and political reorganization.

"In addition to an overall report, an explicit output of the study shall be a definition of areas where additional research is required (including suggested Scopes of Work)."

This report consists of ten chapters, briefly outlined as follows: Chapter 2 summarizes important results, conclusions and recommendations. Chapter 3 describes the attack scenario as specified by DCPA, including the details of the nuclear attack and the location of people under relocation planning. Chapter 4 discusses an idealized fallout pattern from the attack, briefly reviews the current U.S. RADEF program, and surveys radiological monitoring and control, including that of the Soviet Union. Chapter 5 presents a brief survey of current communications capabilities in the U.S. followed by a rough estimate of what might survive a nuclear attack, the effect of EMP, and what communications will be essential for postattack survival. In Chapter 6 the status of food and water in the U.S. is reviewed as it might exist before and after the attack, with emphasis on the location and quantities of surviving grain stocks in relation to the distribution of the surviving relocated people. Transportation capabilities before and after the attack are discussed in Chapter 7 with emphasis on the capability of the trucking industry to transport grain in order to avert starvation. Petroleum reserves, refining and shipping capabilities, before and after an attack, are discussed in Chapter 8 primarily in regard to needs for survival, principally food transport and fuel for heating. Estimates of the additional medical load brought about by relocation of people followed by an attack are given in <u>Chapter 9</u> in addition to a review of status of drug supplies. Requirements for governmental functions and social structure are discussed in Chapter 10 especially in relation to survival capability. The last chapter is followed by an annotated bibliography

of about 150 reports relating to postattack survival and three appendices which contain material too detailed and/or technical to include in the main body of the report.

10.0723-024

testing to an a star of the started

į.

and the states of the second

#### 2. SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS

#### 2.1 Summary

A hypothetical nuclear attack of 6559 MT, of which 5951 MT are ground bursts, is assumed to strike industrial and population targets in the U.S. in the not-too-distant future. If the people in high-risk areas are relocated during the crisis period preceding the attack to rural shelters where they are remote from the direct effects of nuclear weapons and protected against fallout, can they survive through the first few weeks after the attack, until recovery operations are well under way?

The two major threats to individual survival in the early postattack situation under these circumstances are: (1) excessive exposure to fallout radiation through improper monitoring and control of dose; and, (2) shortages of food.

In regard to the first threat, this report reviews established exposure control guidelines for fallout radiation, including the "penalty" table recently published by the Mational Council on Radiation Protection, and assesses the capability for following them.

In regard to the second threat, the location of people as planned with the current program and the location of grain stocks indicates a possibility that millions may perish from food shortages unless food shipments are begun within two or three weeks after the attack. Sufficient grain to feed the entire population of the U.S. for several months to more than a year, depending on the season, exists in storage in the local areas where it is produced. When the quantity of grain in storage is less than a year's supply, there is adequate grain growing in the fields, much of which can be harvested with little radiation hazard to agricultural workers if appropriate precautions are taken. If the attack occurs in June, when crops are on the average most vulnerable to fallout radiation, about one-third to one-half of the annual crop yield could be destroyed. Additional crop failure could occur due to increased ultraviolet radiation resulting from depletion of the ozone layer. しんべきになれたい かくだいまたであいまし ちょうちょうかい ちょうかんしかい ちまたいない ちょうちょう ちょう いちょう

しい しろんで キャン えんでいい かんなまっとうたんへんちものちちちょう ちょうしていろう

The shipment of grain to deficit areas will require not more than 2% of the 1970 rail shipping capability <u>nor</u> more than 6% of the trucking capability, and sufficient reserve petroleum will survive to accomplish this shipment.

The survival of communications, transportation, petroleum, electrical power, and the requirements for medical aid are surveyed and assessed only as far as they are necessary to assure survival of the relocated population through the first few weeks after the nuclear attack. The results at this assessment are summarized in Table 2.1.

Citizens' Band radios will probably survive in numbers adequate for critical civilian communications. In 1973, there were approximately 4 million CB transmitters in the U.S., and this number may double by 1977. Shipment of food and other crucial supplies will use primarily trucks and trains, of which at least 60% may be expected to survive because of relocation measures taken during the crisis period. Ships and barges may not be very useful in the first few weeks after the attack because of fallen bridges and destroyed locks and docks, although the vessels themselves may survive because of crisis period action. Oil pipeline terminals wil' be damaged or destroyed in crucial locations, and most of the refineries will be destroyed. However, about two billion gallons of diesel and about three billion gallons of gasoline would survive in tank storage outside of the major risk areas, which would be more than adequate for the trains and trucks to carry out survival missions during the first few weeks after the attack.

At least 20 million gallons of aircraft fuel will survive in tanks at lesser airports, which may be available to light aircraft of the Civil Air Patrol. First priorities should be given for reconnaissance of transportation routes, surveying blockages by debris and fallen bridges, and monitoring radiological hazard with aerial survey meters.

Very few large interconnected power plants are expected to be operating in the first few weeks after an attack because of disruption of the transmission grid by blast and fire. It is anticipated that electrical power will not be essential for basic survival in the first few weeks after the attack and will gradually be restored during the recovery period.

## Table 2.1

# Summary of Assessment of Postattack Survival of Sectors for Support for the Relocated Population

Sector	ltem	Estimated Survival
Communications	Telephones AM Radio Broadcast	Classified information At least two operating stations receivable at any location in U.S., and at least 100 mill <sup>4</sup> on portable AM receivers
	FM Radio Broadcast CB Radios Public Safety Transmitters	Not estimated 4-6 million mobile units (1977) 0.5 to 1.5 million mobile units, 80-100 thousand fixed units
	Industrial Services Transmitters Land Transportation Transmitters Amateur and Disaster Radio Services	<pre>0.5 to 1.5 million mobile units, 90-110 thousand fixed units 0.8 to 1.2 million mobile units 80 to 120 thousand fixed units</pre>
Transportation	Automobiles Buses (includes school buses) Trucks Train Locomotives Freight Train Cars Aircraft (single engine)	<pre>At least 60 million (over 60%) 200-240 thousand (50-60%) 11-16 million (50-75%) 15-22 thousand (50-75%) 0.9 to 1.3 million (50-75%) 59-88 thousand (50-75%) (at least 1000 airports with runways of 4000 feet or less will survive)</pre>

9

11.14

2013 มีของสามารถสาว เป็นสายารถสาว

.

Table 2.1 (cont'd)

Pipelinesable to function because of unction because of the point of the poin	Ships 3-4 hundred (50-72%) Ships pipelines will survive but will no able to function because of destroyed termin controls	Z-3 thousand (50-75%) Towboats	insportation (con'd) II-17 thousand (50-75%) (usefulness limited bec of destroyed locks and port facilities)	Aircraft (twin engine or larGor) 11-17 thousand (50-75%)	Estimated Survival Sector	l suivris horizon	<pre>11-17 thousand (50-75%) (usefulness limited because of destroyed locks and port facilities) of destroyed locks and port facilities) 2-3 thousand (50-75%) 3-4 hundred (50-75%) Most of the pipelines will survive but will not be able to function because of destroyed terminals a controls 3.0-4.4 billion gallons (40-60%) 3.0-4.4 billion gallons (40-60%) 2.3-3.4 billion gallons (40-60%) 15-25 million gallons 80-100 refineries, having 20% pre-attack crude refining capacity 1.4-2.1 thousand (40-60%) 1.4-2.1 thousand (40-60%) 1.4-2.1 thousand (40-60%) 1.4-2.1 thousand (40-60%) 1.4-2.1 thousand (40-60%) 1.4-2.1 thousand miles (90-95%) Most of the power transmission net will survive but will not funct transmission net will survive but will not funct</pre>	Item Aircraft (twin engine or larcor) Barges Towboats Ships Pipelines Pipelines Gasoline, automotive Gasoline, automotive Mviation gasoline Aviation gasoline Refineries Refineries Power generation stations Transmission networks	Sector insportation (con'd) troleum troleum
---	--	-----------------------------------	--	--	------------------------------	-------------------	---	---	---

10

A WARDAN SHARE BEACH AND A SHARE AND A SHARE AND A

1.14

#### 2.2 Conclusions

Under the assumptions specified for this research program in the Scope of Work statement, we are to assume that people in high-risk areas are relocated "in accordance with currently conceived elements," one of which was the ADAGIO computer program which assigns 89.6 million people to host areas. According to this assignment, about 90% of the U.S. population would be located remote from the blast and fire effects of the nuclear weapons of the 6559 MT attack, and would therefore survive through the attack period.

Also, according to assumptions specified in the Scope of Work statement, we are to assume that "radiological protection has been provided and used during and after the attack," again according to currently conceived ideas. The fallout radiation from the specific 6559 MT CRP-2B attack is more severe than most attacks which have been considered in the past, and it may be necessary to increase the protection factor requirements of shelters to cope with the increased threat.

If we assume that the currently conceived ideas provide adequate fallout procection, then we conclude that most of those people in host areas why survive the attack will also survive through the early postattack period, <u>provided that</u>: (1) guidelines for control of exposure to radiation are generally known, equipment is available to enable shelterees to adhere to these guidelines, and control measures are established to ensure that shelterees adhere to these guidelines in actuality; and (2) a vigorous shipping program of grain stocks is inaugurated in the first two or three weeks after the attack to prevent food shortages in those areas which do not have an adequate food reserve. Survival of adequate grain stocks and the emergence of a more-thanadequate transportation capability shortly after the attack is virtually certain.

and the second second

STATE AND AND A

Proper equipment in shelters includes adequate food and water for the anticipated period of confinement, adequate ventilation, two or more radiation survey meters and/or dosimeters for shelters of 50 or more occupants, two-way communications either by portable radio transmitterreceiver or telephone, and at least one portable AM radio receiver per

shelter. At the present time there are not enough dosimeters or survey meters to provide two for every large shelter. However, there are more than enough portable AM radio receivers, and broadcasts from AM stations should provide adequate general information concerning fallout in their vicinity to prevent casualties due to people leaving shelters too soon after the attack.

Additional modest Civil Defense efforts carried on through several more years could result in a significant increase in the number of survivors over the number resulting from the current situation by developing the full potential of currently conceived Crisis Relocation Plans.

#### 2.3 Recommendations

2.3.1 - The increased threat from fallout radiation posed by current force levels indicates that protection factors of shelters may have to be increased to reduce fatalities, and also to permit the control of dose accumulation so that it occurs primarily outside the shelters in the postattack situation. If the radiation protection of "currently conceived ideas" is adequate, as we are to assume for this study, then the additional requirement indicated by the increased threat is an increase of the number of dosimeters in reserve storage. During the shelter confinement period it may be necessary for one or more volunteers to perform an urgent mission or errand into the external environment, and at later times of the confinement, an increasing number of people may be required to leave the shelters for various tasks. If the shelter occupancy averages 100 people, there should be at least three dosimeters per shelter. Two dosimeters should be reserved for external use to provide a fairly good indication of accumulated dose among those who leave the shelters to work during the later periods of shelter confinement, that is, during the period of gradual emergence from shelters. An additional dosimeter should remain within the shelter to determine the accumulated exposure of the shelterees. If 180 million people are sheltered in the host areas, then, at 100 people per shelter

12

「おおおいろうろうないないとなっていろうないというないないないないである」

Construction of the second

on the average, and three dosimeters per shelter, 5.4 million dosimeters would be required for shelters. Actually a number of people may be located in small shelters, averaging perhaps 5 to 10 people per shelter. These shelters should have at least one dosimeter--an urgent requirement for the first emerging person and possible subsequent emergents, depending on the local situation. If dosimeters are not available in the shelters, some people may refuse to emerge, even though they are told by the local AM radio broadcasts that the radiation fields have decayed to safe levels.

Dosimeters should also be provided for workers in critical industry, who may be asked to volunteer to return to work in areas contaminated with fallout. Each person who must enter alone into a contaminated area should have a dosimeter, and every small group of two to five people who work together in a contaminated area should have at least two dosimeters. If a group working in a contaminated area has only one dosimeter and that dosimeter becomes damaged during the working time, then there is no record of the dose received, and it must then be assumed that the group was exposed to the maximum existing radiation in that area as determined by radiation survey meters. This procedure, which is similar to current practice with radiation survey crews, will usually result in the grounding of the work crew for a period, the length depending on their exposure history and the intensity of radiation in the hottest spots in the area in which they were working. If we assume that 20% of the workers in nonagricultural industry, of which there were 72 million in 1972, are involved in a critical industry, and that a dosimeter is required on the average for every other person, then 7.2 million dosimeters would be required for these workers. Perhaps as many as half of the dosimeters provided for shelters could be used for the critical labor force. These would reduce the number of dosimeters required for the critical labor force to 4.5 million, if 5.4 million are provided for shelters (although more dosimeters may be required for shelters when those with low occupancy are counted).

An additional number of dosimeters may be necessary for farmers, in the eventuality that their fields are contaminated with radioactive

Sugar Sec

fallcut during planting or harvest time. Because of the nature of farm labor, a dosimeter should be provided for each farm laborer. In 1970, there were 2.3 million people employed in agriculture, including selfemployed, wage and salary, and unpaid family workers, indicating that about 2.3 million dosimeters should be made availetle for agricultural workers.

The total number of dosimeters required according to this very rough estimate is 12.2 million, compared with 2.7 million dosimeters on hand at the present time. A more detailed survey should be made to determine with greater cercainty the number of additional dosimeters which may be required to cope with the greater threat of the current nuclear force levels. From our rough estimates, it seems certain that a large number of additional dosimeters would be required. If these additional dosimeters cannot be supplied, then information on the construction of improvised fallout meters should be widely disseminated.

THE REAL PROPERTY AND A DESCRIPTION OF A

2.3.2 - The Radiological Defense Planning and Operations Guide should be rewritten to adopt the guidelines of the "penalty" table, and the ERD concept should be dropped. The mathematical formulation of the ERD requires calculations in order to determine the ERD and these calculations are beyond the capabilities of the average layman, whereas the "penalty" table is very simple to understand and provides equally effective, if not better, guidelines.

2.3.3 - Alternate relocation plans should be considered which would minimize not only exposure to fallout, but also reduce the requirements for grain shipments after the attack. For example, heavy relocation from the cities of New York and Boston into the New England states results in a difficult food supply situation, whereas, if part of this population drove a greater distance during relocation into Ohio or Virginia, for example, the quantity of grain shipped could be greatly reduced.

2.3.4 - Further research on food crops is required to determine the effects of fallout radiation <u>combined</u> with additional ultraviolet

radiation which may result from depletion of the ozone layer. It must be emphasized that the gamma and beta radiation to which the plants are exposed <u>must</u> simulate the <u>decaying</u> intensity of the fallout radiation at early times after the detonation. A gamma radiation field of unvarying intensity has been used in the past, and the extrapolation of these results to the case of a field of decaying intensity is doubtful because of the self-repair mechanisms which occur in plants.

2.3.5 - Additional maps of the U.S. should be prepared similar to Fig. 4.1, which would show contours of fallout radiation in terms of the unit-time reference dose rates, but for different wind conditions. These maps would have general usefulness for planning and for damage estimation. The computer program which generated Fig. 4.1 was designed to permit easy modification of entry parameters, and additional maps can be generated for different input conditions at relatively low costs.

2.3.6 - According to Bull and Sobin (1970) the animal feed industry has considerable potential for rapid conversion to the processing of grain for human consumption in a national emergency. A survey of feed mill location and capacity was made by USDA in 1971. This data base should be analyzed to determine the role these mills may have in connection with CRP, the CRP-2B attack, and postattack survivel.

2.3.7 - A detailed estimate should be made on a county-by-county basis of the quantity and condition of non-grain food resources surviving the attack, including an assessment of the capability of postattack food-processing plants to supply the surviving population.

2.3.8 - A survey should be made of caves and mines which are suitable for human occupancy with special attention to the prevalence of histoplasmosis in the caves and mines. ይዲስ ለተየቆደሴ ውስት መቆን ያስያስፈልን ያይህ እሱስት ልቅት ብሃት ድርስ የሚሰራ የሚሰራ የስት የስት በማስት የሆኑ በማስት የአስት መስት የሰላ ላይ በባት በመጠር መጠር መ

SUSAN SALA

#### 3. THE ATTACK SCENARIO

17

#### 3.1 Introduction

The general scenario to set the stage for research on this project was outlined for us in the first four statements of the Scope of Work specified by DCPA, as quoted in Chapter 1. Briefly, it consists of (1) an international crisis; (2) relocation of people in the U.S.; (3) protection against fallout; and (4) a nuclear attack. In this chapter, further details will be presented and examined on the location and distribution of the relocated people, and on the magnitude of the nuclear attack. Before getting into those details, it may be useful for the general reader to review the evolution of policy which prescribes this scenario.

From studies throughout the 1960's on how to protect people from the blast and fire effects of nuclear weapons, DCPA (then OCD) reached the following basic conclusions in the early 1970's, as quoted from a fact sheet on crisis relocation planning released by Information Services of DCPA on May 20, 1974:

"(1) If an attack should occur, the primary enemy targets probably would be U.S. missile sites. military installations, and centers of industry and population (i.e., metropolitan areas).

- "(2) An attack very likely would be preceded by a period of international tension or crisis. This could constitute 'strategic warning,' and provide time for protective actions to be taken.
- "(3) A great deal of protection against radioactive fallout (i.e., fallout shelter) already exists in the United States, and more is being identified (mostly in new buildings) as time goes on. Attention should not be given to protection against nuclear blast and fire.
- "(4) Blast and fire would endanger mainly people living or working near military targets and in large metropolitan areas. These two types of location may therefore be called 'high-risk' areas.

"(5) It is not financially feasible to build special underground blast-and-fire shelters in these high-risk areas.

#### PRECEDING PAGE BLANK-NOT FILMED

"(6) It may be feasible, however, when an international crisis threatens to result in a nuclear attack, for residents of high-risk areas to be temporarily relocated in snall-town and rural areas, where nuclear weapons probably would not be targeted, provided these people could be protected against radioactive fallout."

The conclusion underlined in the sixth statement above has led to a comprehensive investigation by DCPA into the practicality of evacuating high-risk areas when nuclear attack threatens. This program of crisis relocation has been adopted as a civil defense option by the Department of Defense, as described in the following words by former Secretary of Defense James R. Schlesinger in a budget report to Congress on February 5, 1975:

"The Soviet Union for many years has given a great deal of attention to civil defense, including not only the construction of shelters and the training of civilians but also the preparation of plans for evacuation of the bulk of the population from its major cities in the event of a crisis. Thus, the Soviet leaders have the option to evacuate the cities or to shelter the population in place, depending upon their assessment of the situation at the time.

"We believe that the United States should have a similar option for two reasons: (1) to be able to respond in kind if the Soviet Union attempts to intimidate us in a time of crisis by evacuating the population from its cities; and (2) to reduce fatalities if an attack on our cities appears imminent.

"Similarly, this nation should have the option in the event of an intense crisis to evacuate the civilian population from high risk areas near such military installations as SAC bases, ICBM fields, SLBM support facilities, etc., to less hazardous areas while protecting the rest of the population against fallout. As noted last year, a Soviet counterforce attack which deliberately avoids our cities would still produce a large amount of nuclear fallout which could drift over areas that are downwind from strategic military installations. This civil defense option would complement the military response options that we are now introducing into our planning to strengthen deterrence against a Soviet counter-force attack.

"Accordingly, we propose to continue our efforts, within the limits of the resources available, to improve いいろうないないがっ ちょういろん たいきい ちょうしょう たいたちのちょう たまま たちちょう たいしょ

our ability to protect the population in place against fallout and to develop in an orderly way two major options for the relocation of the population in a crisis. The first option, which would be designed against the threat of a Soviet counterforce attack, would involve the relocation of the population from high-risk areas near key military installations and the protection of the rest of the population against fallout. This option could reduce nationwide fatalities due to fallout from a limited Soviet counterforce attack to relatively low levels -- well under one million -provided that the people in the communities that would be most exposed to fallout from such an attack make effective use of the shelters available.

"The second option, which would be designed against an all-out Soviet nuclear attack, would involve the evacuation of the population from cities, as well as from areas near key military installations. Repeated studies have shown that the evacuation of the bulk of the population from our major metropolitan areas could save some 70 million lives in an all-out Soviet attack on the United States, over and above those saved by inplace protection options.

"Pilot-project work undertaken in areas near some of our important military installations during fiscal years 1974-75, has established the feasibility of developing plans to allocate risk area populations to surrounding host areas, including the development of standby public information (for publication during a crisis) on 'where to go and what to do' should relocation be implemented. Public officials at state and local levels in the pilot areas accepted the need for this type of contingency planning but pointed out that federally-supported planning assistance would be needed.

"Other studies indicate that it would be feasible to relocate population from cities over a period of several days, and to provide for their reception and care in host counties for a period of up to two weeks. Specially-tailored solutions, however, would have to be developed for the most densely-urbanized parts of the U.S., such as the Northeast. It would also be feasible to redirect the distribution of food and other essentials to support evacuees in host areas, provided adequate state-level planning is done with industries concerned. Pilot-project experience with a 'host area survey' indicates that local plans in host areas can provide for protecting evacuees from fallout radiation by use of best available existing protection, plus crisis action to improve fallout protection in existing buildings and to construct expedient shelters."

ころにないたいないないないないないないないないないできたいとうないないないないないない

LIE There where a start and a

Given the assumptions concerning relocation as stated above, and that 90% of the U.S. population survives the attack, it is our task to assess their continuing survivability throughout the postattack period.

#### 3.2 The Attack

If there were an all-out nuclear war between the U.S. and the Soviet Union, how would the Soviet Union distribute weapons on the United States? It is generally believed among defense planners in the U.S. that the principal targets, in order of priority, would be:

- (1) U.S. military installations
- (2) Military-supporting industrial, transportation and logistics facilities
- (3) Other basic industries and facilities which contribute to the maintenance of the U.S. economy
- (4) Population concentrations of 50,000 or greater.

These target priorities, combined with public projections of Soviet capabilities (circa 1980) under existing strategic arms limitations, can be used to generate possible Soviet weapon assignments which are useful for planning purposes. A specific hypothetical attack, prepared by DCPA and other defense planners, was given to us for our study. This attack, titled CRP-2B Attack, is summarized in Table 3.1, and consists of 1444 detonated weapons, for a total of 6559 megatons, of which 5051 megatons are ground bursts. The map of the U.S. in Fig. 3.1 shows circles within which the overpressure from blast exceeds 2 psi. Target points on ICBM fields do not designate actual locations of silos in order to retain an unclassified status for this attack, but they are sufficiently accurate to indicate possible fallout patterns for CRP. Other military targets were located partly with the aid of the February 1973 map titled "Major Army, Navy, and Air Force Installations in the United States," prepared by the Defense Mapping Agency. and the second destines desired and an all of a second and the second and

ikutski a konoriatiski konosiski usen alia osebilasi to analasi osebilasi in one ta teosiski topistoje osebilas

If people remained in place throughout the attack, about 125 million (1970 Census) would be located within the 2 psi circles, and about 58 million would be inside the 15 psi region, as shown in Fig. 3.2. The coverage on Northeastern U.S.A. for these two overpressures, chosen for

Та	b	1	е	3	•	1

Summary of CRP-2B Attack

Target Type	Numbers of Weapons				Type of Burst	Total Megatons
	Megatons					
	<u>1</u>	2	3	20		
ICBM Fields	0	0	0	127	Ground	2540
SAC AFB or SSBN Support	46	1	0	0	Ground	48
CRP/OHVM	183 <sup>.</sup>	0	0	1	Ground	203
CRP/UI	0	0	0	113	Ground	2260
CRP/UI	614	0	0	0	Air <sup>*</sup>	614
CRP/UI	0	183	1	0	Air <sup>*</sup>	369
CRP/UI	0	0	175	0	Air <sup>*</sup>	525
TOTALS	843	184	176	241		6559

NARKAN KARANA KARANGAN NARKAN KARANA KARAN

starke betale at hallen versk ville betale taken in it or on one on a subset of the sold

\* Height of burst chosen to maximize extent of 10 psi overpressure.

out on a subscript of a subscript of the birth of the bir



CRP-2B Attack Pattern on the U.S. Fig. 3.1 and a survey of the second second and the second second second second second

Sour and and a source and the result of the

10115



a salahari

SALAN AND A



23

いいいちんちんちょう ちょうちょうちんちょうちょう

and the second second

A DESCRIPTION OF A DESCRIPTION OF

illustrative purposes only, are shown in Figs. 3.3 and 3.4. It is seen that bursts overlap on some targets. Due to unreliability of missile systems, the attacker cannot be certain that all of his weapons will reach their targets and explode. Hence, it is assumed that additional weapons would be assigned to some targets which the attacker especially wishes to destroy. The population numbers given above indicate that millions of people could survive the attack without relocation if they were provided with blast and fallout shelters of modest hardness. However, if relocation is an available option, then the act of remaining in shelters in high-risk areas would seem to many people to be playing Russian roulette, in view of the uncertainties as to the actual assignment of weapons by the Soviets and the inaccuracies in placing the weapons. It may be necessary for some people to take such risks if they are involved in critical industry, or are maintaining a manufacturing operation for which shutdown may be either disastrous or require many days.

The fallout resulting from this attack will be discussed in Chapter 4, Radiological Hazards and Defenses. Another aspect of this attack, concerning the survival of petroleum refineries (30% survive), will be discussed in Chapter 8 on petroleum. Other unclassified weapon assignment programs of similar yield have been developed by the Federal Preparedness Agency, GSA, and are designated as UNCLEX 73-Charlie and UNCLEX 73-Mike, according to whether the attack concentrates on civilian (Charlie) or military (Mike) targets.

#### 3.3 Crisis Relocation

The hypothetical attack described above can be used to define highrisk areas from which relocation of population should be planned. In order to define these areas, the attack was modified in two ways, first, to maximize direct effects (blast) and second, to maximize fallout. For the first modification, all weapons were assumed to be airburst, systems reliability was 90% and the CEP of weapons was 0.5 nautical miles. Counties in which the geographical centroid was subject to a 50% or greater probability of receiving blast pressure of 2 psi or more were

「「「「ない」」というないで、「いう」




considered to be high risk areas due to direct effects. For the second modification, all weapons were assumed to detonate at ground level, and all counties with a 50% or greater probability of 10,000-roentgen, 4-day cumulative dose or more at the centroid were assumed to be at high risk, under certain fallout arrival assumptions.

With these definitions of high risk areas combined with the ADAGIO-S computer program (Schmidt, 1974), the DCPA computer facility at Olney, Maryland, was used to develop a preliminary nationwide allocation of population. It was assumed that 80% of the population in all highrisk areas would be relocated, involving a total of 89.6 million people. Travel distance was limited to 200 miles on a straight line basis, except for California and the New England states. A hosting factor of 3 (ratio of evacuees to residential population) was used for all states except as indicated in Table 3.2. A hosting factor of 9 was used in California because the non-target areas have such a low population compared with the population in target areas, due to desert and mountain terrain. This hosting factor could be reduced by requiring that California sea coast residents drive even further inland.

「ある」では、ため、ころのないのないないないとなって

ber fester site and the second strategies and second to be been as a second one of the second second second se

A copy of the computer tape generated by this program was sent to us by DCPA, which made available, among other data, the total population per county after relocation. These data are displayed graphically in Figs. 3.5-3.11, in which the U.S. is broken into seven regions, and the number of people per county is indicated by the grid of lines drawn within the county boundary. A fairly complete evacuation is indicated for those counties for which all or nearly all of the area of the county is designated as a high-risk area, such as Nassau and Suffolk counties in New York (Fig. 3.5); De Kalb county in Georgia (Fig. 3.7); Hennepin and Ramsey counties in Minnesota, containing Minneapolis and St. Paul (Fig. 3.8); Chambers and Galveston counties in Texas (Fig. 3.9); and San Francisco and San Mateo counties in California (Fig. 3.11). In some areas these maps indicate the presence of people in high-risk areas, or in areas which are impossible to live in, such as desert areas. Usually these areas are large counties in which the people are actually located only in a part of the county, such as Cook county in Illinois, San Bernadino county in California, and Coconino county in Arizona.

Tab	le	3.	2

States with Relocation Hosting Factors Greater than Three

Arizona	5	Maine	6	Ohio	4
California	9	Maryland	4	Pennsylvania	6
Colorado	3.5	Massachusetts	6	Rhode Island	6
Connecticut	6	Michigan	4	Vermont	6
Delaware	4	New Hampshire	6	Virginia	4
Florida	5	New Jersey	6	Washington	4
Illinois	4	New York	6	West Virginia	4
Indiana	4				

Contractor and

a - and an and the sector of the events and a sector of the sector of th

ા ૬ કે આ પ્રેસ્ટ્રેસ્ટ્રિસ્ટ્રેસ્ટ્રિસ્ટ્રેસ્ટ્રેસ્ટ્રિસ્ટ્રેસ્ટ્રિસ્ટ્રિસ્ટ્રિસ્ટ્રિસ્ટ્રિસ્ટ્રેસ્ટ્રિસ્ટ્રેસ્ટ્રિસ્ટ્રેસ્ટ્રિસ્ટ્રેસ્ટ્રિસ્ટ્

ALL DARGENESS



Fig. 3.5 Relocated Population, Northeast. The legend to the right indicates the number of people per county.

ORNL DWG 76-5867







and the second second strated and a strategy and a second s

Marine Street





a set of the second second

ىرىدىغۇر مەركىم مەركىم ئەركىمىڭ قىرىخار - (مەر مەردىكى ئەركىمىكە كەركىمىكە ئەركىمىكە ئەركىمىكە مەرد

BARYANA MAN

X STREET



2月、午秋代が生まれるから







ORNL DWG 76-5872

ner vering her not beidel automatic veringen beiden heligen in der en beschräften die

33

NAME AND ADDRESS OF THE PROPERTY OF THE PROPER

(CER ) .





and the second of the second second

. ....

and fr

di anti

34

CRN1 DWG 76-5871





and the second second

35

Same and the state of the second s

The state of the second se

The movement and care of relocated populations are under continuing study. For the purposes of this investigation, the preliminary distribution reported herein will be used to study the effects of radiation from fallout and to assess the availability of food. As we shall see from a study of food distribution based on grain stocks, a different relocation plan could be considered which would reduce the requirements for grain shipments in the postattack situation.

AD STRACT AND STRATE A REAL ON STRACT STRATE STRATE OF STRATES AND ADDITION AND A

ધાન્યત્રાથિકોઈટેસ્ટનેન્ડસીઇપ્રિંગ્લ્પેન્ડસ્ટોર્સિઝીઇપિકો સ્ટીઝિઝીઈસ્ટ સ્પાદ સ્થાત પ્રાત્ય કાર્યકાઈક્<mark>ડિડીઇ</mark>ક્સિડીડી

#### 4. RADIOLOGICAL HAZARDS AND DEFENSES

#### 4.1 Introduction

Of all the disastrous effects of nuclear weapons, the hazards of radioactive fallout from ground-burst detonations cover the largest area. The Soviet arsenal of nuclear weapons has the capability of covering over half the area of the United States with radioactive fallout which would be lethal to unprotected humans. In this study, we assume that fallout protection has been provided, hence our concern with fallout will be with the manner in which it impedes survival activities after the attack. We will construct a hypothetical pattern of fallout from the attack described in Chapter 3 and use it to indicate the general magnitude of the problem. Radiological mapping capabilities as they currently exist in the United States will then be described, followed by a discussion of radiological control guidelines, including those of the Soviet Union.

#### 4.2 Fallout from the CRP-2B Attack

A number of computer programs have been developed for the purpose of predicting fallout from ground-burst nuclear weapons (Polan, 1966). Our patterns for fallout from the CRP-2B attack, for which the unit-time reference dose rates are shown in Fig. 4.1, are based on the WSEG-10 NAS/RAM/LAS model, as transmitted to us by Leo A. Schmidt (1974). For simplicity we have assumed that all weapons are detonated simultaneously, and that the wind is uniformly from the west with an effective velocity of 25 mph. The wind shear is assumed to be 0.2 mph per kilofoot of vertical cloud thickness, the terrain factor and fission-fusion ratio are both unity, and the normalization factor, also called the K factor (Advisory Committee on Civil Defense, 1973), is 1930 roentgens per hour per kiloton per square mile. Although these conditions and assumptions are somewhat idealized, the results should be adequate for gaining an insight into the magnitude of the problem, and to provide a basis for generating plans to cope with the problem. In an actual situation, the



the state of the second of

•

こと、これによっているという。 これ、これにはないないないないないないないないないないないないないないないないない

winds and weather will be highly variable, resulting in extremely complex patterns of fallout such that the determination of the extent of fallout radiation must be determined by actual measurements by radiation survey instruments. The capability for such measurement will be discussed after discussion of this hypothetical model.

The darkest areas in Fig. 4.1 indicate : reas in which the unit-time reference dose rate is 10,000 R/hr or greater, and the total area at this level is about 60,000 sq mi, as summarized in Table 4.1. The cumulative area covered by fallout as a function of unit-time reference dose rate is shown graphically in Fig. 4.2. If we assume that the radiation intensity decays uniformly throughout the country, and that the fallout remains fixed in place after being deposited on the ground, then the dose rate isopleths in Fig. 4.1 represent the relative levels of radiation intensity at various times after the attack. According to the standard radiation decay model for fallout from nuclear weapons, the intensity of radiation decreases with time according to  $R = R_{1}t^{-1.2}$ , where R is the intensity of radiation in roentgens/hour at time  $\underline{t}$  in hours after the attack, and R is the unit-time reference dose rate. Factors by which the radiation intensity is reduced according to this law are listed in Table 4.2. These factors may be applied to the contours in Fig. 4.1 to indicate the radiation intensity levels at time t (in hours) after the attack. As an example, the contours in Fig. 4.1 for 10,000 R/h at unit time (H+1) become contours of 100 R/hr in two days after the attack, and the same contours indicate levels of 10 R/hr in thirteen days after the attack, and after 90 days, the same contours represent radiation levels of 1 R/hr.

Because the biological effects of radiation depend on <u>accumulated</u> dose, the contours in Fig. 4.1 are not immediately useful to a defense planner. In order to increase the usefulness of these contours, the graphs in Fig. 4.3 were developed, which show the four-day dose or peak ERD (Equivalent Residual Dose) as a function of the arrival time of the fallout. In an actual situation, the fallout will arrive at different times at different locations, due to varying wind conditions. In this case, the unit-time reference dose rate can be estimated by measuring

-
•
4
e
-
<u>a</u>
đ
H

0
Ξ.
ğ
1
-
ŝ
4
ຍິ
÷.
et l
Ē.
B.
ğ,
Ŧ
ц
ut
lout
llout
Fallout
r Fallout
by Fallout
d by Fallout
red by Fallout
ered by Fallout
overed by Fallout
Covered by Fallout
s Covered by Fallout
as Covered by Fallout
reas Covered by Fallout

Unit-Ti Dose Rat	me Reference te (R/hr)	Area sq mi	Cumulative Area sq mi	Cumulative Percent of Area of Coterminous U.S.
≥10,000		60,000	60,000	2.0
3000 to	10,000	295,000	355,000	12.0
1000 to	3000	446,000	801,000	27.0
300 to	1000	648,000	1,449,000	48.9
100 to	300	493,000	1,942,000	65.5
30 to	100	437,000	2,379,000	80.3
10 to	30	170,030	2,549,000	86.0

I



Table	4.	. 2
-------	----	-----

#### Reduction Factors for Radiation Intensity Due to Decay

Time After Attack (Days)	Reduction Factor
1.9	0.01
13.2	0.001
89.8	0.0001

ì

and Design of South

9 - 18 - 19 E J

TUNE LAURE



Fig. 4.3 Four-Day Dose and Peak ERD as a Function of Arrival Time of Fallout.

the radiation dose rates at various times, plotting these values on a graph, and extrapolating back to one hour after detonation.

Because the radiation is decaying while it is enroute to the final deposition area, the dose at the area will be reduced considerably, depending on how far downwind the deposition area is located from the detonation. The dose to unprotected humans as a function of the fallout arrival time can then be estimated by the factors given in Fig. 4.3.

The fallout patterns in Fig. 4.1 were calculated with the assumption that the mean wind velocity for fallout transport is 25 mph from the west, as mentioned previously. Suppose, as an example, that we wish to estimate the four-day dose to unprotected humans at Marshall, Minnesota, under these wind conditions, and for the CRP-2B attack. Marshall is located at approximately 44.4° N latitude and 95.8° W longitude. The upwind nuclear bursts which will produce fallout on Marshall, under the wind and attack assumptions used here, are located in western South Dakota. The latitude scale on the left side of Fig. 4.1 can be used as a distance scale; one degree of latitude corresponds to 60 nautical miles or 69.1 statute miles. The distance from Marshall, Minnesota, to the nuclear burst sites in South Dakota varies between approximately  $4.5^{\circ}$  and  $5.5^{\circ}$  of latitude, or between 311 and 380 miles. With a mean effective wind for fallout transport of 25 mph, the time of arrival of fallout in Marshall will range between about 12 hours for the earliest arrival and 15 hours for the latest. From Fig. 4.3, the factor for estimating the four-day dose (the dashed line) is about 1.07 for 12 hours arrival time, and about 0.96 for 15 hours arrival time. We will use F = 1. From Fig. 4.1, the unit-time reference dose rate,  $R_0$ , is near the 3000 R/br isopleth, so we estimate it to be about 2500 R/hr. The four-day dose at Marshall for these conditions for unprotected humans is then 2500 roentgens, obtained from the product of F, the factor obtained from Fig. 4.3 and  $\rm R_{_{O}},$  the unit-time reference dose rate obtained from Fig. 4.1.

Exposure to 400-450 roentgens of whole-body radiation is considered to be fatal to 50% of the people exposed, as discussed in sections 4.4 and 4.5. It is evident that fallout shelters would be required in Marshall, Minnesota, under these conditions. Factors which enter the

considerations in selecting the degree of protection are discussed on page 75.

If a person remained an entire lifetime in an area contaminated by radioactive fallout from a single attack by nuclear weapons, the total dose, if unprotected, would be approximately four times the unit-time reference dose rate. However, because of biological recovery from much of the radiation damage, the Equivalent Residual Dose (ERD), as defined by the equation in Fig. 4.3, will reach a peak at about five days after the attack in the case where the fallout reaches the area in one hour after the detonations, about 25 miles downwind for the hypothetical case we are considering, and the peak ERD will be about 2.8 times the unittime reference dose rate.

The number of people in the areas of fallout are listed in Table 4.3 for two cases, the 1970 residential population and the population relocated according to the ADAGIO program. A more complete display of the data obtained is shown in Fig. 4.4, which plots the percent of people, plotted on the ordinate, which are located in areas having a unit-time reference dose rate equal to or greater than that shown along the abscissa. The names of forty counties which have the heaviest fallout from this attack are listed in Table 4.4.

It is often useful to have a single index by which the relative effectiveness of one situation can be compared with another. The effectiveness of the relocation in avoiding fallout can be roughly indicated by a single index,  $I_f$ , which we call the Fallout Avoidance Index. If the relocation is the best possible with regard to avoiding fallout, the index will have the value of unity. If the relocation makes no change, the index will be zero, and if conditions are worse, the index will be negative. We define  $G_e = \sum_i P_{ie} R_i$ , the sum over all U.S. counties of the product, in each county, of the population in the county after relocation, and  $R_i$ , the unit-time reference dose rate in the county. Similarly, we define  $G_r = \sum_i P_{ir} R_i$ , involving the residential in situ population. Finally, we define  $G_o = 1.94 \times 10^{-5} P$ , where P is the total population of the country, and the constant is the dose rate per hour based on the tolerable background radiation of 170 mr/yr, as

Table 4.3

TAL NAMES IN THE REAL PROPERTY OF THE REAL PROPERTY

With the

Population in Areas Covered with Fallout (Hypothetical Situation)

Unit-Time Reference Dose Rate (R/hr)	Population (19 (millions)	<u>70 Residential)</u> (percent)	Population (millions)	(relocated) (percent)
10,000	18.55	9.17	8.74	4.32
3000 to 10,000	48.65	24.07	30.60	15.13
1000 to 3000	46.86	23.18	51.43	25.43
300 to 1000	45.06	22.29	50.62	25.03
100 to 300	18.24	9.02	25.50	12.61
30 to 100	10.95	5.42	15.50	7.67
10 to 30	3.17	1.57	3.81	1.88
0 to 10	10.66	5.28	15.94	7.92

and the second





a seaso

2.4日間に設計した。1.4月間になって、1.4月間に

THE REAL PROPERTY OF THE PARTY OF THE PARTY

TABLE 4	٠	4	
---------	---	---	--

A COLUMN AND A COLUMN AND

and a manufacture of the state of

1.1.1.1.

TO A STREET STREET STREET ST

	County	State	Unit-Time Reference Dose Rate (R/hr)
1.	Queens	New York	32 000
2.	Nassau	New York	30,000
3.	New York	New York	20,600
4.	White	Arkansas	19,800
5.	Cooper	Missouri	17,800
6.	Moniteau	Missouri	17,300
7.	Deuel	Nebraska	17,000
8.	Chevenne	Nebraska	16,600
9.	Bronx	New York	16,600
10.	Suffolk	New York	16,400
11.	Boone	Missouri	15,400
12.	Keith	Nebraska	15,300
13.	Kent	Maryland	15,300
14.	Cole	Missouri	15,100
15.	Clinton	Illinois	15,000
16.	Kings	New York	15,000
17.	Callaway	Missouri	14,900
18.	Woodruff	Arkansas	14,500
19.	Walsh	North Dakota	14,100
20.	Cross	Arkansas	14,100
21.	Pettis	Missouri	13,800
22.	Grand Forks	North Dakcta	13,300
23.	Osage	Missouri	13,200
24.	Ramsey	Minnesota	13,100
25.	Petroleum	Montana	12,900
26.	Sedgwick	Cclorado	12,900
27.	Shelby	Tennessee	12,800
28.	Bristol	Massachusetts	12,700
29.	Madison	Illinois	12,700
30.	Cochise	Arizona	12,400
31.	Warren	Missouri	12,300
32.	Washington	Minnesota	12,200
33.	Marion	Illinois	12,100
34.	Marshall	Minnesota	12,000
35.	Wilson	Kansas	11,900
36.	Montgomery	Missouri	11,900
37.	Fayette	Tennessee	11,700
38.	Pennington	Minnesota	11,600
39.	Contra Cosa	California	11,500
40.	Queen Annes	Maryland	11,500

# Forty Counties with Severe Fallout Radiation from the CRP-2B Attack (Hypothetical Situation)

a management of the

set by the NCRP. The Fallout Avoidance Index, I<sub>f</sub>, may then be defined as:

$$I_{f} = (G_{r} - G_{e})/(G_{r} - G_{o})$$

If the relocation results in placing people such that the radiation in their new locations is equal to the specified tolerable background radiation, then the value of the index is unity, indicating a good solution to the problem. If the relocation results in the same sum of products of population and radiation as the residential location, then the index will be zero, indicating a poor solution to the problem. For this attack,  $G_e = 4.0 \times 10^{11}$  and  $G_r = 6.8 \times 10^{11}$  for the 1970 residential population, and  $G_o = 3922$ . The value of  $I_f$  is 0.41. The ADAGIO relocation was based on a different fallout pattern, hence the index is not as high as it would have been if the relocation had been based on the fallout pattern used here, as shown in Fig. 4.1.

The fallout areas shown in Fig. 4.1 can be used as a rough indicator for requirements for fallout shelters under fairly general wind conditions. Transport of fallout from megaton-yield weapons takes place primarily in the stratosphere. In the winter, stratospheric winds blow predominantly from the west, but in the summer over much of the U.S., the stratospheric winds blow from the east (Crutcher, 1959). Because any wind direction is possible (the probability is not relevant if alternatives are available unless the shelterees wish to engage in Russian roulette), the areas of possible heavy fallout around high-risk areas is represented by a circle which can be generated by placing a compass point on the western edge of the dark area, and inscribing a

Alternately, the fallout protection factor (PF) for a shelter in a given area can be estimated as follows: First, locate the geographical position of the shelter on the map; second, place a compass point on the western edge of the nearest and darkest fallout pattern (the point of detonation); third, with radius equal to the distance from the compass point to the shelter location, inscribe a circle which passes through the downwind portion of the fallout pattern; forth, estimate the maximum value of the unit-time reference dose rate where the circle intersects the fallout pattern resulting from the weapon detonation location at the center of the circle. This procedure is repeated with all the neighboring fallout patterns which appear to have a significant effect.

For some winter wind conditions, the effective wind velocity may be twice that used to generate Fig. 4.1 (25 mph) and if these conditions are to be used, the length of the downwind patterns would be increased and the width reduced. For more detailed evaluations, several maps should be used, corresponding to several wind speeds, and more detailed contours should appear on the maps. The time of arrival is estimated by calculating the distance from the shelter location to the upwind edge of the pattern, as provided for the unit-time reference dose rate, and then dividing that distance by the effective wind velocity. Figure 4.3 can then be used to calculate the peak ERD or 4-day dose, from which the desired PF can be estimated.

An example of application of the procedure described above may be helpful. Suppose we are going to build a fallout shelter in the vicinity of Edgemont, South Dakota. In Fig. 4.5, two circles are drawn which pass through Edgemont and have their origins in the western edges of two of the darkest (heaviest fallout) regions in the vicinity of Edgemont. If the effective wind velocity were 25 mph and blew from the southsouthwest at the time of the attack, the fallout pattern to the south of Edgemont in Fig. 4.5 would be rotated so that Edgemont would lie within the region of heaviest fallout of this pattern, and the fallout from detonations to the north would not affect Edgemont at all. Similarly, if the wind blew from the north-northeast at the time of the attack, the fallout pattern to the north of Edgemont would be rotated so that Edgemont would lie on the tip of the darkest portion of that pattern, and the fallout from detonations from the south would have no effect on Edgemont. The radii of the two circles are approximately 100 and 130 miles, and the unit-time reference dose rates are estimated to be 10,000 R/hr and 13,000 R/hr respectively, and the fallout arrival times are approximately 4 and 5.2 hours respectively.

and the state from the state of the books of the state of





From Fig. 4.3, the peak ERD factors are respectively 1.7 and 1.53, resulting in estimated peak ERDs to unprotected humans of 17,000 R and 19,900 R respectively. The latter number would be the preferred number to use for specifying the shelter  $^{\rm p}$ F, which would be about 400 if a peak ERD of 50 R were to be allowed in the shelter. If several maps were used, corresponding to different wind velocities, several values of the peak ERD would result, and the highest value obtained would be the value chosen for specifying the PF of the shelter.

Radiation levels are shown in Fig. 4.6 which would exist at one year after the attack if there were no decontamination measures taken and if no leaching by rain occurred. These results were estimated with the use of detailed analysis of the decay of the various isotopes (R. Chester, 1974) because the standard decay law does not apply after about five or six months after the detonation. At one year after the attack the radiation intensities are decaying slowly, and the one year dose starting at one year after the attack can be estimated approximately by multiplying the hourly dose-rate shown in Fig. 4.6 by by 8760, the number of hours in a year. The lowest contour in Fig. 4.6 is 0.0001 R/hr, and encompasses an area of 2.4 million square miles, 80% of the U.S. area, as listed in Ta 4.5. The long-range effect of this radiation on humans will be div  $x_{1000}$  later in this chapter, and the effect on crops will be discussed in chapter 6.

#### 4.3 The U.S. RADEF Program

Radiological Defense (RADEF) is a first-priority emergency preparedness program at federal, state, and local levels of government, designed to enhance the survival of citizens from fallout threats in the event of a nuclear attack. Protection of the people from radiation hazards and early implementation of survival measures can be accomplished only through an organized capability of detecting, monitoring, reporting, and analyzing the fallout situation at each affected locality. Radiation measuring and detection instruments in the hands of trained personnel are the only means of gaining reasonably accurate information on the fallout radiation level at a given time and place, because the levels of



### Table 4.5

# Area Dose-Rates One Year After a Hypothetical Attack

Dose-rate, (R/hr)	Area (sq mi)	Cumulative Area (sq mi)	Percent of Area of Coterminous U.S.
0.1	108	108	0.004
0.01	347,000	347,000	11.7
0.001	1,089.000	1,436,000	48.5
0.0001	944,000	2,380,000	80.3
0.00001	274,000	2,654,000	89.5

54

A CARLES AND A C

Ê

and the second second

alexand Mississificity

 $R_{2}^{0}(A)$ 

radiation will vary in time due to local climatic conditions, during and after the fallout deposition, and the variations may be considerable over short distances.

Federal guidance relating to RADEF is given in Part E, Chapter 5, of the Federal Civil Defense Guide (FCDG), which is currently being updated by DCPA to include the changing strategic threat, crisis relocation, and various peacetime nuclear threats.

Five basic components of the RADEF program are:

- the provision of radiological monitoring capability for each shelter;
- (2) the establishment of a network of appropriately dispersed centers of monitoring and reporting capability called "monitoring stations;"
- (3) the development of capabilities at EOCs (Emergency Operating Centers) to process the raw radiological data into readily usable form, and to provide staff support through interpretation of data, provision of technical guidance, and recommendations of possible courses of action;
- (4) the provision of instrumentation and training for the radiological self-help protection of emergency service and vital facility personnel; and
- (5) the provision of dosimeters for postattack exposure control for civil defense workers engaged in recovery operations.

The provision of radiological monitoring in each shelter and in its immediate vicinity is essential in order to maintain a record of exposure for each individual, especially those who must make excursions out of the shelter for survival purposes. Radio broadcasts will be able to provide only general rough approximations of radiation levels in the environs of shelters.

The network of "monitoring stations" will serve as a base from which: (a) monitors will perform on-station monitoring during the period when the radiation hazard is great, and (b) detailed mobile monitoring will take place during the period when radiation rates will permit limited field operations on a controlled risk basis. Aerial monitoring can effectively supplement, but not replace, the detailed monitoring station functions, especially for monitoring transportation routes, agricultural

lands, etc. Data obtained by the monitoring stations will be communicated to EOCs for processing, preferably by radio, in view of the threats to telephone lines from EMP, blast and fire.

The system of EOCs and monitoring stations requires trained RADEF officers (RDOs) plotters, analysts, recorders, and radiological monitors. As of June 30, 1975, there were 3500 qualified RDOs in the U.S., with a goal of 10,000, and 177,000 trained radiological monitors, with a goal of 380,000.

The basic radiological instruments necessary for measuring dose rates (survey meters) and accumulated dosages (dosimeters) are provided by the federal government to the states and through them to local government for use in community shelters and state and local monitoring stations. Additional and more specialized equipment has also become part of the total instrumentation relating to civil defense, such as aerial monitoring instruments, remote sensor radiation meters for EOCs, etc. To insure operational readiness and reliability, a system to control, maintain, repair, and calibrate equipment is also provided through federally-funded state shops.

This RADEF capability is deteriorating in some states due to shortage of state and federal funds, and from lack of interest in others.

A brief description of the DCPA radiological instruments for operational use is given in Chapter 7, Annex 1, of the <u>Radiological</u> <u>Defense Planning and Operations Guide</u>, SM-11.23.2, revised March 1967, Department of Defense. Office of Civil Defense. The number and general distribution of instruments in the U.S., as of July 1974, is listed in Table 4.6. The number of monitoring kits and other survey instruments are listed by state in Table 4.7. The monitoring kit contains one V-700 radiation survey meter (low range, high sensitivity), two V-715 high range gamma survey mecers, two V-742 high range dosimeters, and one V-750 dosimeter charger. Many of the 138,000 shelter kits listed in Table 4.6 are located in areas which are considered to be high-risk areas under CRP. These kits should be moved to the relocation areas during the crisis period, and the plans for this movement should be made in detail at the local level before a crisis occurs.

i de la companya de l

## Table 4.6

# Radiation Detection Instruments Currently Distributed to States (figures rounded off)

Source: DCPA-Operations & Planning

		Number	Approximate	Cost
Α.	Instruments			·
	Survey Meters			
	Low Range (V-700) 05 mr/hr 0-5 mr/hr 0-50 mr/hr	425,000	\$ 60.90	
	Remote Sensor (V-711) High Range (V-715) 05 r/hr 0-5 r/hr 0-50 r/hr 0-500 r/hr	400 530,000	100.00 60.00	
	Remote Reading (V-717) Chamber remotable to 25	80,000	100.00	
	High Range (V-720) 0-5 r/hr 0-50 r/hr 0-500 r/hr	95,000	80.00	
	Dosimeters (Self-reading)			
	Training (V-138) 0-20 Roentgen (V-730) 0-100 Roentgen (V-740) 0-200 Roentgen (V-742)	190,000 130,000 155,000 2,700,000	25.00 20.00 20.00 20.00	
	Chargers			
	For all dosimeters (V-750)	500,000	15.00	

• • • • •

ator inter a var a tor to

Contraction R. Str.

n en reen in de la service e

and the state of the state

Strain and

Table 4.6 (cont'd)

Number Distribution Β. The total of instruments in states is about 5,100,000 In monitoring stations - 61,000 kits (1 ea. V-700, 2 ea. V-715, 2 ea. V-742, 1 ea. V-750) 366,000 In shelters - 138,000 kits (same except only one V-715 690,000 Emergency worker dosimeters and chargers in state buildings (e.g. national guard armories) 2,100,000 900,000 In state maintenance (one location per state) In federal warehouse (Richmond, Virginia) 600,000 In training (high schools, etc. throughout states) 400,000

atta (Adual adual a sua a a a sua substanti

recommended siting: at 1-3 miles centers for urban areas. at 7-10 miles centers for rural.

# Table 4.7

# Distribution of Monitoring Kits (July 1974)

State	Total Number of So (Monitoring Kits)	ets Other Survey ) Instruments
Alabama	1927	1
Alaska	371	õ
Aricona	555	Ő
Arkansas	1187	ĩ
California	6063	18
Colorado	524	0
Connecticut	1155	1
Delaware	185	121
Dist. of Col.	62	25
Florida	912	0
Georgia	1708	Ő
Hawaii	238	Ő
Idabo	699	82
Illinois	1426	127
Indiana	1142	10
Iowa	967	0
Kansas	1123	Ő
Kentucky	1447	Ő
Lousiana	1102	Ő
Maine	881	107
Maryland	1112	595
Massachusetts	1519	0
Michigan	1448	67
Minnesota	1504	1
Mississippi	867	ō
Missouri	1201	Ō
Montana	992	0
Nebraska	1138	0
Nevada	424	0
New Hampshire	1	1
New Jersev	2126	0
New Mexico	704	Ő
New York	3952	2
North Carolina	990	0
North Dakora	811	22
Ohio	2154	0
Oklahoma	1581	0
Oregon	707	0
Pennsylvania	2346	1

ALTER ANT ALTER

Table 4.7 (con'd)

State	Total Number of Sets (Monitoring Kits)	Other Survey Instruments
Rhode Island	488	0
South Carolina	1083	0
South Dakota	796	0
Tennessee	1405	164
Texas	3321	0
Utah	592	5
Vermont	211	0
Virginia	893	15
Washington	1018	0
West Virginia	807	8
Wisconsin	2111	0
Wyoming	436	49

and the state of the state

Stell's Station and Station

1 A A
In addition to these instruments, 1100 V-781 aerial survey meters have been distributed to the states for use in the Civil Air Patrol aerial radiological monitoring program. The availability of aircraft, fuel, and runways for carrying out this program will be discussed in Chapter 7.

### 4.4 Radiological Exposure Control Guidelines

The basic approach of DCPA to control of radiological exposure is described in considerable detail in the Radiological Defense Planning and Operations Guide, SM-11.23.2, revised March 1967, (RDPOG) which is a reprint of published and draft materials from the Federal Civil Defense Guide. A more recent guide is given in NCRP Report No. 42, Radiological Factors Affecting Decision-Making in a Nuclear Attack, November 15, 1974. The underlying philosophy in both documents is to provide a description of the hazards of radiological exposure, how to detect and protect against it, and how to keep track of cumulative doses without attempting, for practical reasons, to specify precise exposure limits for all people in all kinds of situations which may occur in a postattack situation.

One anticipated situation, for example, is that some shelters may be poorly stocked with food or water, and it may be necessary for someone in the shelter to volunteer to take a calculated risk and deliberately expose himself to radiation in order to procure supplies for survival. Another situation may occur after the radiation hazard has diminished to some extent, when it may become necessary for some shelterees to participate in radiological monitoring surveys, or in rescue work, or in the shipping of vital supplies to less fortunate areas. In all these cases it is extremely important to have detailed information on the radiological condition in the immediate vicinity of the shelter. This information could be obtained either by radiation survey meters in possession of the shelterees, or by radio communication (two-way) with someone external to the shelter who has conducted a survey of the shelter environment. If neither survey meters nor two-way communications with an external surveyor are available or possible, the shelterees may

have to rely only on AM broadcasts for the radiological situation. which may be grossly in error for their particular location. The possibility of "hot spots" due to climatic conditions can result in radiation intensities which may be a hundred times stronger in one location than in another which is only a mile away. A more detailed discussion of shelter survival in hazardous radiation fields is given in Section 4.6, in this chapter.

During the early periods following a nuclear attack (the first week or two) before a complete evaluation of the hazards can be determined, the recommendations shown in Table 4.8, taken from RDPOG, p. 3-27, may be used as a guide for directing shelter and operational activities. In order to benefit from this guidance, there must be some kind of radiation survey meter in the shelter. As soon as information becomes available as to the age of the fallout, the guidelines in Table 4.8 should be modified. If the fallout is relatively young (2 or 3 hours old) at the time of measurement of the dose rate, the radioactivity is decaying rapidly, and relaxation of control of exposure to radiation in the shelter can be tolerated to some degree. However, if fallout is several days or weeks old at the time of measurement, then the radiation intensity is decaying slowly, and more rigid control of exposure is necessary.

The guideline for exposure of emergency personnel to radiation as given in RDPOG is that "to the extent practicable the ERD of emergency personnel should always be kept well below 200 R." For workers in critical areas (non-emergency) the comparable maximum ERD is 100 R. An extreme total exposure for emergency personnel during a 12 month period which would keep the ERD below 200 R during the first year, would permit no more than 200 R in the first month, no more than 25 R per week in the next 5 months, and no more than 10 R per week in the next 6 months. The maximum exposure for the entire year under this schedule is about 1000 R, which is extreme. There is one example in history so far where a person has received about 1000 R in a period of 106 days and survived (Lushbaugh, in Tobias and Todd, 1974, pp. 502-503). According to Lushbaugh, "If the experience of the one Mexican survivor can be used as a criterion, normal man may only be able to achieve similar tolerance to

appressive and the second

7.7.

# Guidelines for Shelter and Operational Activities (Taken from RDPOG)

If Outside Dose Rate Has Fallen to: (in R/hr)	Activities That May Be Tolerated
Less than 0.5	No special precautions necessary for performance of essential tasks, except to keep fallout particles from contaminating people ar sleep in the shelter.
0.5 to 2	Outdoor activity (up to a few hours per day) tolerable for essential purposes, which include fire fighting, police action, rescue, repair, securing necessary food, water, medicine and blankets, important communications, disposition of waste, exercise and obtaining fresh air. Eating, sleeping, and all other activities should be conducted in the best available shelter.
2 to 10	Very short periods (less than an hour per day) of outdoor activity are colerable for the most essential purposes. Shelter occupants should rotate outdoor tasks to minimize total doses. Outdoor activities of children should be limited to 10 to 15 minutes per day. Rescue, repair, communications and exercise may safely take place in less than optimum shelter.
10 to 100	Time outside of shelter should be held to a few minutes and limited to those few activities that cannot be postponed for at least one more day. Insofar as possible, all people should remain in the best available shelter no matter how uncomfortable.
Greater than 100	Outdoor activity of more than a few minutes may result in sickness or lethality. The only occasions which might call for moving are (1) risk of death or serious injury in present shelter from fire, collapse, thirst, etc., and (2) present shelter is greatly inadequate

274 372 AV

an average marrow dose of about 6 rads/day if the irradiation is by high-energy, low LET photons." If this rate of exposure were tolerated for a year, the total dose would be about 3300 R from fallout-type radiation. In consideration of possible long-range effects of exposure to radiation, to be discussed in the pext section, it is decidedly to the advantage of the individual to be exposed to the minimum quantity of radiation which is compatible with the accomplishment of an emergency task.

A comprehensive study of the effects of exposure to radiation was sponsored by NASA to establish guidelines for astronauts on spaceflights of long duration (Space Radiation Study Panel, 1967). In the last chapter on Evaluations and Recommendations, it is assumed that an acceptable reference-equivalent space exposure  $(RES_m)$ , established on the basis of the risk-versus-gain philosophy, was 250 refere •••••.valent units (reu) for a one-year space mission. For fallout- adiation, 250 reu is approximately equivalent to 250 rads, midline absorbed dose, or about 375 R, whole-body exposure. If this dose were absorbed according to the schedule given in Table 33 of Radiological Factors in Manned Space Flight (RFMSF hereafter) (Space Radiation Study Panel, 1967) the ERD, calculated according to the formula in RDPOG, at the end of the year would be about 30 rad, or an exposure ERD to fallout-type radiation of about 45 R. If this recommendation were applied to the extent that the guideline for exposure of emergency personnel to fallout radiation were lowered from 200 R to a maximum ERD of 50 R, then an exposure rate which would keep the ERD below 50 R during the first year would be as follows: less than 100 R in the first month; less than 15 R per week in the next five months; and less than 3 R per week in the remaining six months, for a total maximum yearly dose of about 475 R.

The ERD concept of human response to radiation has been discrewited by a number of people (Sacher, 1958; Sacher and Grahn, 1964; Storer, 1959; Langham, 1967; Steward, 1974) yet it has provided the basis for extensive guidelines for control of exposure to radiation throughout the world, partly because none of the detractors cited proposed an alternative scheme for planning purposes in the event of large-scale radiation

hazards presented by nuclear war. A recent effort to provide an alternate guideline is the "Penalty" table, reproduced in Table 4.9 from Appendix B of NCRP Report No. 42. It relates three categories of exposure rate conditions (columns a,b,c) with three categories of expected consequences (rows A,B,C), depending upon total accumulated exposure. Examples of the use of this table are given as follows, as quoted from the NCRP Report No. 42:

Example 1:

A MARCAN STATES A CONTRACT OF A CONTRACT OF

analysi (India's Starfie Sciller and I may the second

Purpose: To limit exposure to low medical risk. (Refer to row A.) To achieve this purpose, it would be necessary to limit the total radiation exposure of individuals to less than 150 R in any one week (column a); 200 R in any one month (column b); and 300 R in any four-month period (column c).

For example, if individuals receive the one-week limit of 150 R (column a) within the first week, then the limit for additional exposure during the ensuing three weeks of the first month, to keep within the onemonth limit (column b), would be 200 R - 150 R = 50 R. This additional exposure of 50 R could be received in any period of time, ranging from one day to three weeks of the ensuing three weeks of the first month, without exceeding the one-week or one-month limits in the "Penalty" Table. However, if this additional exposure of 50 R were received, for example, within the second week, then the individuals would have to be kept free of further exposure during the remainder of the first month to keep within the one-month limit for row A (200R). Similarly, if the individuals have received the limit of 200 R in the first month, without exceeding 150 R in any one week of that month, the limit of additional exposure for the ensuing three months of the first four months (column c) would be 100 R for a total of 300 R (200 R + 100 R) in four months.

Example 2.

Purpose: Operations at the intermediate level of significant medical risk (row B), justified by highly critical emergency situations.

In this case, the de^ision-maker may find it necessary to allow greater exposure than one or another of the limits indicated in row A, but would be constrained whenever possible by other limits in row A, and always by limits in row B of the Penalty Table.

and the second second

and a subject of the and a stread of the subject of

The "Penalty" Table

		Accumulat (R)	ted Radiation ) in Any Per	n Exposures iod of
	Medical Care Will Be Needed By	а	Ъ	с
		One Week	One Month	Four Months
A	NONE	150	200	300
B	SOME (5 percent may die)	250	350	500
С	MOST (50 percent may die)	450	600	

For example, if individuals who have received 150 R within the first week are required in some emergency to receive an additional 200 R during the remainder of the first month (for a total of 350 R in the first month), it is desirable, if possible, that the one-week constraint for row A (column a ) be observed by allowing, no more than 150 R of this additional exposure during any one week within that month, even though the onemonth limit (200 R) and four-month limit (300 R) for row A will have been exceeded and the one-month limit (350 R) for row B will have been reached. If it is not possible to keep within any of the constraints for row A, then the row E constraints have to be applied, in an attempt to keep exposure in any one week as far as possible below 250 R, to limit the exposure during the first month to 350 R. Any additional exposure after this first month must be kept as far as possible below the additional 150 R which would attain the four-month limit of 500 R (row B).

As in Example 1, the decision-maker could schedule exposures in a variety of ways within the constraining limits to meet the work required by the problem at hand.

Example 3.

Enter a la contra de la contra de

Purpose: Operations at the high levels of medical risk (row C), justified only by extremely critical emergency situations.

In extreme emergencies, situations could arise that might justify operating at the high risk level (row C). Those activities that could result in saving a significant number of lives may call for the deliberate exposure of some persons at the highest constraint levels where radiation sickness and a 50 percent probability of death are expected (row C). If such situations arise, the decision-makers would use for guidance row C of the Penalty Table in a manner similar to that discussed for the low or intermediate risk rows (A and B) in Examples 1 and 2 above.

According to Example 1, it would be necessary to limit the total radiation exposure to less than "300 R in any four-month period," if medical care were to be avoided. This criterion indicates that a total exposure of 900 R could be tolerated over a one-year period, without requiring medical care, as long as the total radiation exposure does not exceed 300 R in any four-month period. One of the conclusions given in RFMBS may have some bearing on establishing a guideline for maximum exposure rates during emergency missions of short duration (less than two days): (p. 256) "For bonemarrow responses, doses delivered at dose rates of 50 rads/day and above are assumed to produce maximum injury per rad, while exposures at vates of 1 rad/day and below are assumed to produce minimum injury per rad accumulated." It is implied here that 50 rads or more is delivered in the maximum case. For fallout-type radiation, 50 rads/day corresponds to about 75 R/day whole-body exposure. This RFMSF conclusion could therefore lead to the criterion that emergency personnel should not be exposed to more than 75 R in any one day.

The Soviet Civil Defense Manual (Egorov et. al., 1970) defines fairly specific categories of radiation exposure rates, and relates them to the mode of transportation of the reconnaissance teams which undertake radiological surveys, as follows: (p. 163) "The terrain is considered contaminated if it has a dose rate of 0.5 R/hr or higher. As a rule reconnaissance on foot is continued to a dose rate not higher than 30 R/hr; in automobiles to a dose rate of not more than 100 R/hr. Reconnaissance of regions with higher radiation levels is carried out by reconnaissance groups (teams) only on special order by the chief who ordered the reconnaissance. Localities with higher dose rates, up to 200 R/hr, can be reconnoitered only in tanks or in armored transports, and higher than 200 R/hr in helicopters or in airplanes; such reconnaissance is conducted by higher CD staff officials."

The Soviet guidelines on exposure to fallout radiation are oriented strongly toward rescue work and getting factories into production as quickly as possible. Equations and tables in their CD manuals are based on the same standard radiation decay formula used in this country, i.e.  $R=R_{o}t^{-1.2}$ , where R is the dose rate at time <u>t</u> in hours after the detonation, and R<sub>o</sub> is the unit-time reference dose rate. An interesting and useful table from the 1970 Soviet Civil Defense manual (Egorov et al., 1970), which shows permissible exposure times in an area contaminated by fallout, is reproduced in Table 4.10. An example is given of the use of this table in the Soviet manual, which will be related here in the following paragraph.

and a star has been and a low marked by a fingure data to see the second star where the second starter

AV Kershaddin

SAME AND

เห็นแหละวิจีที่มีสารมีการคล

and the second second

Permissible Exposure Time in an Area Contaminated by Fallout Resulting from a Nuclear Blast (Soviet) Time of Entry Into the Contaminated Area (From the Time of the Blast) (hr)

D/R Value <sup>a</sup>	0.5	Ч	7	Ś	4	ŝ	9	7	80	6	10	12	24
		Expc for	osure Time Different	e (in Hc : Times	ours and of Entry	Minutes Into t	t for Wh he Cont	nich the caminate	e Deterr ed Area,	nined Va: , Referre	lue D/R ed to th	is Obtai e Blast	rned Time.
0.2	0-15	0-14	0-13	0-12	0-12	0-12	0-12	0-12	0-12	0-12	0-12	0-12	0-12
0.3	0-22	0-22	0-20	0-19	0-19	0-19	0-19	0-18	0-18	0-19	0-18	0-18	0-18
0.4	0-42	0-31	0-27	0-26	0-26	0-25	0-25	0-25	0-25	0-25	0-25	0-21	0-24
0.5	1-02	0-42	0-35	0-34	0-32	0-32	0-32	0-31	0-31	0-31	0-31	0-31	0-30
0.6	1-26	0-54	0-44	0-41	0-39	0-39	0-38	0-38	0-28	0-37	0-37	0-37	0-37
0.7	205	1-08	0-52	0-49	0-47	0-46	0-45	0-45	0-44	0-44	0-44	0-44	0-43
0.8	2-56	1-23	1-02	0-57	0-54	0-53	0-52	0-51	0-51	0-51	050	050	0-49
0.9	4-09	1-42	1-12	1-05	1-02	1-00	0-59	0-58	0-58	0-57	0-57	0-57	0-55
1.0	5-56	2-03	1-23	1-14	1-10	1-08	1-06	1-05	1-05	1-04	1-04	1-03	1-02
2.0	1562-00	11-52	4-06	3-13	2-46	2-35	2-29	2-24	2-20	2-18	2-16	2-13	2-06
2.5		31-00	6-26	4-28	3-48	3-28	3-16	3-08	3-03	2-59	2-55	2-51	2-40
3.0		96-39	9-54	60-9	5-01	4-28	4-10	3-58	3-49	3-43	3-38	3-30	3-14
4.0	•· 1	3124-00	23-43	11-05	8-12	6-57	6-16	5-50	5-33	5-19	5-10	4-58	4-26
6.0			193-19	35-35	19-48	14-43	12-19	10-55	10-02	9-24	8-57	8-19	7-01
10.0				728-49	124-00	59-18	39-34	30-39	25-42	22-35	21-32	17-52	13-08

<sup>a</sup>D/R equals permissible dose in roentgens divided by the dose rate R/hr at the moment of entry into the contaminated region.

Suppose the nuclear explosion occurred at 6 o'clock, and the working crew is to enter the area at 8 o'clock, at which time the radiation intensity is measured at 20 R/hr. If the established allowable cumulative exposure is 40 R, how much time can the crew spend in the area? It is assumed that the radiation exposure is negligible until they enter the area. According to the measurement of the radiation intensity and the value of "established allowable dose," the ratio D/R = 40/20 = 2, and the time of entry is 2 hours after the blast, hence we find from Table 4.10 that the permissible exposure time in the area is 4 hours and six minutes.

The Soviet Civil Defense manuals do not discuss the basis for establishing permissible levels of radiation exposure, and the concept of ERD or the accumulative effect of radiation exposure are not introduced, although an instrument for measuring cumulative dose, a dosimeter, is described. However, if the duration of a work shift (or of exposure time) is restricted to no more and no less than four hours per day in a contaminated area, as implied in the Soviet manuals, then the entry times as given in Table 4.10 will never permit the maximum ERD to exceed 100 R. In the worst case, within the limits of Table 4.10, entry time will be at 24 hours after the blast, and the radiation fields will be decaying much more slowly than at earlier times. From Table 4.10 the four-hour exposure time corresponds to D/R = 3.6 (by interpolation), from which R = 11 R/hr, and  $R_o = 498 R/hr$ . If the same crew enters the same area at the same time every day for a four-hour exposure, their maximum ERD will be about 97 R at the thirteenth day, and it will decrease thereafter. The cumulative exposure at the time of peak ERD will be 109 R, and the cumulative exposures at 7 days, 1 month, and 4 months are, respectively 93 R, 129 R, and 155 R, all less than the maximum levels indicated in the "Penalty" Table (Table 4.9) for no medical care requirements.

If these examples are indicative of the Soviet policy towards exposure to fallout radiation, then we may conclude that, except for extreme emergency, the maximum permissible ERD for the Soviet emergency personnel is 100 R for a year, compared with 200 R indicated in the U.S.

RDPGG, and about 50 R ERD indicated (for astronauts) by the Space Radiation Study Panel (R. CF). In general, the Soviet policy towards exposure to radiation appears to be more cautious than the U.S. policy, exemplified by higher PFs for shelters and lower limits for exposure doses.

The penalty table may be used to construct a table similar to that of the Soviet's shown in Table 4.10 which will indicate the time after detonation when an area can be entered for a regular period of time each day, for "shift" work, for example, such that the exposure to radiation will not result in requirement for medical aid. In Table 4.11, the time at which regular exposures of specific duration may begin, counting from detonation time, are shown for three different unit-time reference dose rates, 1000, 3000, and 10,000 R/hr. It is assumed that during the periods between exposures the workers are either physically removed from the area of contamination or else they are housed in shelters which have such a high PF that the radiation dose received while inside them is negligible.

Suppose, for example, the unit-time reference dose rate is 1000 R/hr, and we wish to begin regular shift work of eight hours per day in the contaminated region. According to Table 4.11, if we started working at 3.36 days after the detonation with 8 hours' exposure every day, the total exposure in one week would meet the penalty table maximum of 150 R in one week, but if work were continued in the same area for 8 hours' every day for a month, the exposure would be 270 R, which exceeds the dose of 200 R specified by the penalty table for requiring no medical care. In order to meet the penalty table specifications for no medical care, regular shift work of 8 hours' duration per day in an area with 1000 R/hr unit-time reference dose rate should not begin until about  $5\frac{1}{2}$  days after the detonation. We assume here that in the postattack environment people vill work 7 days a week, without taking Saturdays and Sundays off.

Entry times for "shift" work in contaminated areas, based on the penalty table for no medical care requirements, are shown in graphical form for  $100 \le R_0 \le 1000$  in Fig. 4.7 and for  $1000 \le R_0 \le 10,000$  R/hr in

Permissible Entry Times and Doses for Daily Exposure to Fallout Radiation with No Medical Care Required Accoraing to the Penalty Table

Ro, 11-1 t- Time	H T T T T T	Time	t Duration of	Exp	osure in Roentg	* gens	Peak	Day of Occurrence of Peak ERD
Reference Dose Rate (R/hr)	After Pai Days	tonation Hours	Exposure Per Day (Hours)	Week N=7	Month N=30	Four Months N=120	EKD	(After Detonation)
1,000	1.37	32.88	4	150 ((145))	(220)((145))	((220) ((22))	160	1.5
1,000	1.60	38.40	4	(135)((130))	200 ((135))	(255) ((55))	150	16
1,000	1.10	26.40	4	(175)((170))	(245)((160))	300 ((30))	185	14
1,000	2.34	56.16	9	150 ((145))	(245)((170))	(325) ((70))	175	19
1,000	3.43	82.32	δ	((011))(011)	200 ((140))	(275) ((65))	145	22
1,000	2.81	67.44	9	(130)((130))	(225)((155))	300 ((65))	160	20
1,000	3.36	<del>1</del> ر 80	ø	150 ((150))	(270)((185))	(370) ((85))	190	21
1,000	5.58	133.92	8	((36)) (001)	200 ((345))	(295) ((70))	145	26
1,000	5.46	131.04	8	((001))(001)	(205)((145))	300 ((70))	145	26
3,000	5.50	132.00	4	150 ((150))	(305)((220))	(450)((110))	220	26
3,000	10.58	253.92	4	(85) ((80))	200 ((145))	(335) ((85))	145	33
3,000	13.00	312.00	4	((10)) (32)	<u>175</u> ((130))	300 ((80))	130	36

72

いたいしょう いちまたい たいちんいちょう たいちょう たいちょう たいちょう たいちょう

Table 4.11 (cont'd)

A DESCRIPTION OF THE PARTY OF T

a had a day of the second s

R o, Unit-Time	Entry	v Time	t Duration of	IXE	oosure in Roentg	sens *	Peak	Day of Occurrence of Peak ERD
Reference Dose-Rate (R/hr)	After D Days	etonation Hours	Exposure Per Day (Hours)	Week N=7	Month N=30	Four Months N=120	ERD	(After Detonation)
3,000	8.63	207.12	Q	((021)) <u>57</u>	(34')((250))	(250)((140))	250	31
3,000	18.13	435.12	6	((0/)) (0/)	200 ((150))	(375)((105))	155	41
3,000	26.67	640.08	6	((21) ((21))	(145)((110))	((06)) 000	120	47
3,000	11.63	279.12	8	150 ((150))	(370)((275))	((01))((110))	275	34
3,000	25.75	618.00	8	((2)) ((2))	200 ((150))	(410)((120))	160	47
3,000	41.67	1000.08	8	((07)) (07)	((130)((100))	300 ((95))	115	57
10,000	10.00	240.00	2	150 ((145))	(350)((255))	(270)((150))	255	32
10,000	21.00	504.00	2	((02)) (02)	200 ((150))	(385)((110))	155	43
10,000	32.00	768.00	2	((5)) ((7))	(140)((102))	300 ((90))	115	51
10,000	19.00	456.00	4	150 ((150))	(430)((325))	(815)((230))	335	41
10,000	46.00	1104.00	4	((22)) (09)	200 ((155))	(470)((150))	175	59
10,000	83.00	1992.00	4	((0E)) (0E)	(110) ((85))	300 ((105))	110	76
÷ Expo single par for the san	sures are entheses ne exposur	underlined are the expression of the expression	d which duplicat pusures for the rs are rounded c	te the penalty ( periods indicat off to the near	table for "no me ted. Numbers ir st multiple of	edical care req a double parent five.	uired. theses	' Numbers in give the ERD

74



×.

Fig. 4.8. For each specified daily exposure period there are three curves in each of these figures, corresponding to the week, month, and four-month dose accumulation periods specified by the penalty table. When the curves for a specific selected daily exposure period cross each other, the curve which gives the largest entry time should be used, if the penalty table requirements are to be met for all three dose accumulation periods.

The estimated entry times for shift work, as shown in Table 4.11, Figs. 4.7 and 4.8, are based on the assumption that the workers are exposed to negligible radiation during the daily periods between the workshift time in the contaminated area. This assumption implies either that the workers have access to fallout shelters which have a very high protection factor (over 1000), or that they can be transported to a radiation-free area for the time between shifts. These conditions may be unattainable in many locations, because the PF of most shelters will be under 200, and the distance to an uncontaminated area may be too far for practical commutation.

When the PF of shelters is taken into account, the number of possible conditions to consider becomes very large, hence we have investigated only one case as a representative example, viz, the fallout situation in Marshall, Minnesota, as discussed previously on page 44. In this case the unit-time reference dose rate at Marshall (the exposure rate which would have existed at Marshall at one hour after detonation if the fallout cloud had been transported instantly to Marshall) was assumed to be 2500 R/hr. Because of the assumed wind conditions, the fallout cloud does not arrive until around 13.5 hours (average) after the detonation.

We assume that the people of Marshall are in fallout shelters by the time the fallout arrives. Estimated exposures to radiation for people inside shelters of various protection factors are shown in Table 4.12 for various durations ranging from 4 days to 4 weeks. The numbers in Table 4.12 indicate that radiation exposure will be lethal to all occupants in shelters with PF of 5; 30-40% of those in shelters with PF of 10 will die from radiation exposure; all occupants in shelters with PF of 15 will require medical attention for radiation sickness according

The second second second second second

and the state of the state of the



Fig. 4.8 Entry Times into Contaminated Areas for 1000  $\leq$   $R_{0}$   $\leq$  10,000.

Exposures in Shelters of Low PF.  $R_0 = 2500$  R/hr, Fallout Arrival Time 13.5 Hours After Detonation

			Exposur	e (R)		
PF	Time in	Shelter,	Beginning	When Fallo	ut Arrives	Comments
	4 days	1 week	2 weeks	3 weeks	4 weeks	
5	508	602	710	770	808	100% lethal
10	254	301	355	385	404	30-40% lethal
15	169	201	237	263	269	100% radiation sickness
20	127	151	178	193	202	No medical attention required

Sec.

こと、このないと

and the second of the second second

to the Penalty Table; and no one will require medical attention for radiation sickness in shelters with PF of 20 or higher.

A PF of 10 is obtained by shielding with earth of about 12-in. thickness, or about 8 in. of concrete; and a PF of 100 is obtained with earth of about 24-in. thickness, or concrete of about 16-in. thickness.

In the shelter with PF of 20, no one can leave the shelter during the first four weeks without the possibility of requiring medical attention according to the Penalty Table.

Occupants in shelters of higher PF may leave the shelter at earlier times for daily work outside in the contaminated regions as indicated in Fig. 4.9. For example, occupants of shelters with PF of 25 may begin a 4-hour daily workshift outside as soon as 6 days after the detonations. If these people worked outside for 4 hours every day for a week, they would "use up" their "safe" (no medical care required) exposure of 150 R, and their subsequent exposures would have to be carefully monitored to keep their exposure below 200 R for the 4-week period, according to the Penalty Table. The functions of these early outside laborers would be first to map the existing radiation levels and then begin decontamination. Areas such as paved streets and buildings can be effectively decontaminated from fallout radiation by scrubbing and hosing, thus enabling others to come out and work with lower exposure to radiation.

It must be emphasized that these tables are presented for planning purposes only. In an actual situation the radiation decay rate may vary from the standard  $t^{-1.2}$  rate used for the calculations, and the radiation intensity will vary considerably from one location to another, whereas the calculation of the tables is based on the assumption of a uniform radiation intensity throughout the entire area in which people are working. A representative fraction of the people working in contaminated areas must be wearing dosimeters, and checks should be made and the accumulated dose on each instrument recorded several \*imes each day during the first week of work, and at the end of each day for the next few weeks thereafter.

For planning trips by truck drivers, buses, locomotives, etc., the protection factors listed in Table 4.13, taken from Burson (1974) may be



STARTING TIME, DAYS AFTER DETONATION

7**9** 

### Environmental Radiation Protection Factors Provided by Civilian Vehicles

Vehicle	Position	Protection Factor Range
Commercial bus (common type)	Throughout bus	1.5-2.0
Commercial bus (scenic cruiser type)	Throughout bus	1.5~2.0
School bus	Throughout bus	1.5-1.8
Passenger car	Passenger side (chest) Driver side	1.5-1.7 1.5-1.7
Pickup	Driver side	1.9-2.1
Crew cab	Driver side Back seat	1.8-2.0 1.8-2.0
Carryall	Driver side Rear side	1.7 <b>-1.9</b> 1.7 <b>-1.9</b>
2-1/2-ton truck	Driver side Center of bed	1.8-2.0 1.4-1.6
5-ton truck	Driver side Sleeper	2.0-2.2 1.9-2.1
Heavy Truck	Driver side Center of trailer	1.4-1.6 2.7-3.1
Fire truck	Driver side Standing area in back	2.7-3.1 1.6-1.8
Switch engine	Engineer's seat	3.0-3.5
Railway guaru car	Sleeping quarters Kitchen area Center area	2.2-2.6 2.4-2.8 2.0-2.4
Heavy lccomotive	Engineer's seat	3.0-3.5

PARAMON NAME AND A DESCRIPTION

SOURCE: Z. G. Burson, "Environmental and Fallout Gamma Radiation Protection Factors Provided by Civilian Vehicles," <u>Health</u> <u>Physics</u>, <u>26</u>, 41-44, 1974. used to estimate the starting time after detonation for trips through a radiation field of known average intensity, as determined by aerial monitoring first, and then by ground reconnaissance crews. For example, if the average radiation field over the trip area corresponds to a unittime reference dose rate of 3000 R/hr, and the protection factor is 1.5 for the driver of a heavy truck, then the starting time after the detonation can be estimated from Fig. 4.8 by using the ordinate value of 2000 R/hr, obtained by dividing the average unit-time reference dose rate of 3000 by the protection factor, 1.5. As an example, if a portion of a regular trip involves 4 hours' exposure to this radiation field, the truck driver could start these particular trips five days after the detonation, according to Fig. 4.8.

### 4.5 The Basis for Radiological Exposure Control Guidelines

Highly effective control guidelines against a specific hazard can be generated if the effects of hazard on man are definitive, so that the consequences of any specific action involving the hazard can be accurately predicted. Unfortunately, the effects of fallout-type radiation on man are not sufficiently definable to permit precise prediction of the consequences of exposure. The effects on mammals other than man have been exhaustively researched by experiment, but the extrapolation of the results to man remains in question. The most thorough studies of the effects of fallout-type radiation on man must fall back on the experience of the Hiroshima and Nagasaki victims, the Rongalap natives, and extrapolations of relatively few radiation accidents and exposures of patients under clinical conditions.

With these observations in mind, it should be apparent that radiological exposure control guidelines established at the present time provide only an initial structure, which radiological officers in a postattack situation may completely change or extensively supplement on the basis of their actual experience.

For emergency personnel, the radiation exposure levels are based on the expectations of early responses which would interfere with the performance of their mission. NCRP Report No. 42 divides the symptoms of exposure to radiation into five groups as follows:

<u>Group I Symptoms</u>. Less than half of this group will vomit within 24 hours after the onset of exposure. There are either no subsequent symptoms or, at most, weakness and easy fatigue. There is a decrease in the blood cell counts. Less than 5 percent will require medical care. All others can perform their customary tasks. Any deaths that occur are caused by complications. <u>Correlated Exposure</u>. Sickness of this type has been seen after brief, whole-body doses of gamma and Xradiation in the range of 50-200 R. An ERD of external gamma radiation of 50-200 R may have a similar effect.

Group II Symptoms. More than half of this group will vomit soon after the onset of exposure and are sick for a few days. This is followed by a period of 1-3 weeks when there are few or no symptoms. During the latent period, typical changes occur in the blood count and can be used for diagnosis. At the end of the latent period, epilation (loss of hair) is seen in more than half, and this is followed by a moderately severe illness due primarily to the damage to the bloodforming organs. Most of the people in this group require medical care and more than half survive. Correlated Exposure. Sickness of this type has been seen after brief, whole-body doses of gamma or Xradiation on the order of 200-450 R. An ERD of external gamma radiation of the same size will probably cause a similar illness.

<u>Group III Symptoms</u>. This is a more serious version of the sickness described as Group II. The initial period of illness is longer, the latent period is shorter, and the main episode of illness is characterized by extensive hemorrhages and complicating infections. People in this group need medical care and hospitalization. Less than half survive. <u>Correlated Exposure</u>. Sickness of this type has been seen after brief whole-body gamma radiation with doses in excess of 450 R.

<u>Group IV Symptoms</u>. This is an accelerated version of the sickness described as Group III. All in this group begin to vomit soon after the onset of exposure, and this continues for several days or until death. Damage to the gastrointestinal tract predominates, manifested by uncontrollable diarrhea, which becomes bloody. Changes in the blood court occur early. Death occurs before the appearance of hemorrhages or epilation. All in this group need care, and it is unlikely that many will survive. <u>Correlated Exposure</u>. Sickness of this type has been seen after brief, whole-body exposure to gamma radiation in excess of 600 R. During protracted exposure to external gamma radiation, it is not probable that an illness of this type would be the first evidence of injury.

<u>Group V Symptoms</u>. This is an extremely severe injury in which damage to the brain and nervous system predominates. Symptoms, signs, and rapid prostration come on almost as soon as the dose has been received. Death occurs within a few hours or a few days. <u>Correlated Exposure</u>. Sickness of this type has been seen after brief, whole-body exposure to gamma rays in excess of several thousand R, and to equivalent doses from neutrons.

Essentially the same five clinical levels of severity of acute radiation effects are described in the Radiological Defense Planning and Operations Guide.

The correlation of Group I and II symptoms with radiation exposure doses can be broken down into finer detail as a result of retrospective studies described by Lushbaugh (Tobias and Todd, 1974), in which a large volume of clinical data extracted from hospital charts of 2000 patients given therapeutic total-body irradiation were analyzed to give doseresponse relations for the sypmtoms and signs of the prodromal syndrome. These data have been "corrected" for normal man in RFMSF. The correlation of dose with symptoms in these two groups will be of great practical significance to emergency workers in radiologically contaminated areas. The statistically determined single exposures that can be expected to produce these symptoms in 50% of the patients so exposed are shown in Table 4.14. The corrected values for normal man, as adopted by NCRP Report No. 42, are also shown. The table also shows estimates by Lushbaugh of the increased levels of exposures required for the same incidence of response, i.e., 50% of the patients, when the exposure period is lengthened by either fractionating the exposures, or by protracting the dose accumulation by lowering the dose rates.

According to Table 4.14, when total-body exposure of patients occurs promptly in less than one day, the effective doses for 50% incidence of these responses ( $ED_{50}$ ) are anorexia (loss of appetite) 150 R; nausea, 210 R; vomiting, 275 R; and diarrhea, 350 R. According to Lushbaugh, et. al., (1966), the exposures for their 10% incidence would

and the state of the

# Accumulated Estimated Exposures<sup>a</sup> for 50% Incidence of Physiological Symptoms

Exposure for Normal Man (1 Day) (NCRP Report 42)	ngle Radiation 95% Confidence Exposure Range	180 150-210	260 220-290	320 290-310	280 230-310	360 310-410
1974)	8 Days Si (1083) (R)	600	750 <sup>b</sup>	4006	ე	υ
re, Duration, and cients (Lushbaugh	8 Days (103) (R)	300	400	750	, 400 <sup>b</sup>	800
Exposur Number of Pat	1 Day (504) (R)	150	210	275	225	348
	Symptom	Anorexia	Nausea	Vomiting	Fatigue	Diarrhea

<sup>a</sup>Midline upper abdominal dose, rad = 0.66 exposure <u>R</u>.

b"Guesstimate"; 20-30 R/day is the apparent threshold of dose rate. Longer emposures at lower rates are ineffective.

<sup>c</sup>Information lacking.

8۵

Literation and a second second

-

be about one-fourth of that for 50% incidenve. In other words, the approximate effective doses for 10% incidence of these responses (ED10) in patients are anorexia, 40 R; nausea, 55 R; vomiting, 70 R; and diarrhea, 90 R. Plans for the accomplishment of a mission by emergency personnel for which the outcome is critical for the survival of many people, such as the delivery of food supplies by truck, should probably take into consideration the  $ED_{10}$  rather than the  $ED_{50}$  as a basis for estimating maximum exposure during the mission. A truck driver who may require several days to complete his journey would be greatly impaired in performing his driving duty if nausea, vomiting, and diarrhea occurred during the second, third, or fourth day of the trip. Another factor which should be taken into consideration is the prediction, according to analysis by Lushbaugh (Tobias and Todd, 1974, p. 486), that a radiosensitive person who showed, for example, nausea at a low dose would be more likely to show other symptoms and signs of a greater damage per unit of radiation than a radioresistant person in whom nausea did not occur without a much greater exposure.

It is necessary to know at what levels of exposure fatalities will occur, and how quickly, for the purposes of triage and damage assessment. On the human lethal dose problem, Lushbaugh writes, (Tobias and Todd, p. 492): "There is a worldwide willingness to accept ... estimate that the exposure that will kill the unattended normal man wich 5 % certainty within 60 days of exposure  $(LD_{50,i50})$  is 450 R and that the mechanism of death is damage to his hemapoietic system and defense mechanisms against infection. The degree of acceptance of this 450 R value is surprisingly high in view of its history and its lack of valid support from reported human data." The origin of the 450 R estimate lies buried in the personal notes of some of the ten members of a distinguished committee of U.S. radiotherapists, radiation physicists, and pathologists who polled the U.S. community of practicing radiotherapists to determine what size of single total-body (photon) exposures was considered "safe" and "unsafe". Subsequently, there have been several attempts to check the 450 R estimate from human case histories after both accidental and intentional radiation exposures. These are summarized in Table 4.15 to show how all studies have produced values lower

### Some Clinical and Statistical Estimates of Human Total-Body Radiation Tolerance

	Exposure for <sup>LD</sup> 50/60 (Roentgens)
Normal men	
Warren and Bowers (1950)	450
Cronkite and Bond (1960)	528
Langham (1967)	430
Jablon et al., (1969)	614 <sup>a</sup>
Patients	
Mathe er al., (1964)	400
Langham (1967)	380
Lushbaugh et al., (1966)	370
Normal men + blast and burn trauma	
Lushbaugh and Auxier (1969)	394 <sup>b</sup>

<sup>a</sup>Using RBE of fission neutron component = 4.

<sup>b</sup>Using RBE of fission neutron component = 2.

From: <u>Space Radiation Biology and Related Topics</u>, p. 496. Doses given in rads and rem have been converted to exposure in roentgens by dividing by 2/3.

In a postattack situation, where there may be widespread shortage of doctors and medicine, particularly antibiotics, where the sanitary conditions may be unhealthy and morale may be poor, it may be more realistic to assume that 50% of the people exposed to a much lower radiation dose, say 350 R, will become fatalities. Under these conditions, the response to exposure to radiation may be closer to that of patients whose records were studied by Lushbaugh et al., (1966), for whom upper and lower dose-response relations for acute hemotologic syndrome is shown in Fig. 4.10. If we assume the reference dose to be 2/3 the exposure to photons in an average-sized man, then, according to Fig. 4.10, the mid-lethal dose is 250 rads or 375 R of whole-body exposure, the dose for lethality in 10% of those exposed is 75 rads, or about 115 R of whole-body exposure, and the dose for lethality in 90% of those exposed is 400 rads, or 600 R of whole-body exposure. 「「「ないないない」の

Man and other a incls appear to have a recovery capability from the harmful effects of expose to radiation, because it is evident from much research and experience that those exposed to periodic doses, or to a low dose rate over a long period of time can withstand a much greater cumulative dose than when the same dose is administered in a short time of a few days of less. The increases and decreases in the number of cellular eleme .s in the blood appear to be governed by bone-marrow and lymphocytic case recovery after total-body irradiation. Various mathematica' .odels have been proposed to predict the reparability of humans, as described by Steward (Tobias and Todd, 1974, pp. 523-564). One of the best known models, devised by Blair (1952, 1953, 1956), gives an equation for ERD, as previously discussed, which is the damage remaining unrepaired at some specific time after exposure. The theory and application of this model is thoroughly described, exemplified, and applied to the postattack fallout situation by Davidson (1957); however, the results are not useful for current planning even if the ERD concept were still in vogue, because the model is based on fallout radiation





decay according to  $t^{-1.33}$  rather than the current standard decay of  $t^{-1.2}$ . For want of models based on human data, this ERD concept has attained worldwide acceptance by planners for emergencies in which large segments of the population may be exposed to fallout radiations. Reviews of the imadequacies and limitations of this concept (Sacher, 1958; Sacher and Grahn, 1964; Storer, 1959; Langham, 1967; Steward, 1974) suggest its abandonment for human use. In this report, graphs and tables have been presented to indicate exposure guidelines based on <u>both</u> the ERD concept and the "Penalty" table, which is intended to replace the ERD concept.

The ultimate goal of radiation exposure guidelines is to present a general scheme by which RADEF officers and other emergency personnel can minimize the number of fatalities primarily in the first few weeks of a postattack situation, due to radiation hazards. If possible, exposures should be limited with regard to possible late responses such as life shortening, cataracts of the ocular lens, and leukemia. In many locations, these considerations would lead to requirements of shelters with much higher PF than presently considered. The estimated relationships of life-shortening probability and increased probability of leukemia to accumulated dose and intensity of whole-body radiation are shown in Figs. 4.11 and 4.12, respectively, as taken from RFMSF, pages 264 and 265.

4.6 Shelter Survival Conditions in a Hazardous Radiation Field

第二日の人主義をは、今天のおおかかり、 うちゅう たんない 見ためた かいたい ない

It is likely that many shelters in the immediate postattack environment will be found to have deficiencies of one kind or another. Some of these deficiencies may involve discomforts, but others may become lethal if not corrected. The latter category of deficiency will be considered here, and methods for coping with the various situations will be discussed.

The basic requirement of a shelter is, of course, that it provide protection from the harmful effects of radiation. The shelter should also be designed and stocked to provide adequate ventilation, water, and

0 0 INTERMEDIATE OR MIXED HIGH AND TENSITY LOW INTENSITY (\$ I RAD/DAY) ACCUMULATED DOSE (hundreds of RADS) Ø n HIGH INTENSITY (250 RADS/DAY) ¢ R) 2 O 8 0 24 22 20 4 2 0 0 Ø N **LIFE-SHORTENING (YEARS)** 

Space Radiation Study Panel, <u>Radiobiological Factors in Manned Space Flight</u>, Publication 1487, National Academy of Sciences, National Research SOURCE: Fig. 4.11 Life Shortening vs Radiation Exposure. Council, p 264, 1967.

Fig. 4.12 Incidence of Leukemia vs Radiation Exposure. SOURCE: Space Radiation Study Panel, <u>Radiobiological Factors in Manned Space Flight</u>, Publication 1487, National Academy of Sciences, National Research Council, p 265, 1967. 「ない」「「「「「」」」」」」」

and the second tead on a construction of a second



food. A deficiency in any one of these four requirements may result in the necessity for one or more occupants to leave the shelter in search for help or supplies. Such forays may not be necessarily hazardous if the people leaving the shelter know where to go or what to do.

The survival probability of occupants in a shelter can be greatly enhanced by having the following items on hand and in good working condition: a radiation survey meter and/or dosimeter, a telephone or two-way radio, and an AM radio receiver. The existence of these three survival items in the shelter in addition to its fulfillment of the four basic requirements listed above provide maximum survival probability for the occupants of the shelter through the shelter-phase of the postattack period.

If one of the basic requirements of the shelter, either the PF, ventilation, water or food supply is inadequate, it will become extremely urgent that some action be taken to prevent fatalities, if the shelter is situated in a potentially lethal field of fallout radiation. Suppose that while enroute the fallout radiation intensity threatens to become about ten times or more stronger than anticipated, due to a shift in wind direction. If the shelter has an AM radio and/or radic or telephone communication with the outside world, the occupants may possibly become informed of the possibility that the anticipated fallout will exceed their shelter's PF before the fallout arrives, thus enabling them to evacuate to a better shelter, or take steps to improve the PF of the'r shelter. The latter option would usually be open to those who had constructed an expedient shelter, but would not always be a possibility for those in large community shelters. de se de la construction de la serie de la serie de la construction de la serie de la serie de la serie de la s

語文は言語がないないであるとと、その語言は言語がないないないであるとないである。

As an example of the opposite extreme situation, suppose the shelter doesn't possess any of the three survival items--it has no radio or telephone connection to the outside world and no radiation survey or dosimeter instruments. In this situation, an extra heavy fallout situation, as hypothesized above, will arrive without detection or warning, and the first prodromal reaction to the radiation exposure in the shelter, i.e., nausea and vomiting, could easily be interpreted by the occupants as the symptoms of an infectious GI virus. The total dose to each occupant could exceed the lethal exposure before anyone might realize

that they were being over-exposed. This gruesome possibility points out the necessity for each shelter to have some kind of radiation detector. If an official CD survey meter or dosimeter is not on hand, then the occupants should construct an electroscope type of radiation detector, such as the Kearny Fallout Meter (Kearny, 1975). Instructions for constructing these meters should be widely disseminated, because there are not enough CD dosimeters and survey meters to supply all shelters, especially under the circumstances of a crisis relocation where many people will be improvising shelters or constructing expedient shelters. For the remainder of this discussion, we will assume that the shelters are equipped with some type of radiation detector.

Suppose that the PF of the shelter is adequate, but, through overcrowding and/or poor design, the ventilation is inadequate. Long before the level of carbon dioxide becomes lethal there will be a feeling of stuffiness and/or claustrophobia and, in the summer, overheating, among the occupants, to the extent that some will want to rush outside. In this situation the radiation meters may serve to convince the occupants that they should not leave the shelter. If two-way communication is available to the outside world, instructions may be obtained as to the availability of nearby shelter space to which some of the occupants may transfer; or, instructions may be received on how to improve the ventilation system of the shelter. In many large community s'uelters which are located inside large buildings, the radiation survey meters may be used to locate areas in the building where occupants may reside temporarily in order to relieve the burden on the ventilation system of the main shelter. Instructions for making improvised ventilating devices, such as the Kearny Air Pump, should be widely disseminated, and available to every shelter.

の文字を見たいとなるとなったのであるとなったのであると

If there is inadequate water or food in the shelter, radio or telephone contact may be used to arrange for an emergency delivery, or to determine the closest point of supply to which volunteers from the shelter will run or drive to get the necessary supplies. The level of radiation intensity in the vicinity of the shelter should be monitored so that a fairly accurate prediction can be made of the dose to which the emergency crew will be exposed.

93

「ないないないないないないない」というないできた。 ちょうちょう しょうちょう しょうちょう ないないないない しょうしょう

nter and and de and de and the second december of the second beats when an theory of the second december of the

al division and the solution of the solution of

### 5.1 AM Radio Broadcast

After a large nuclear attack on the United States, it will be essential that AM radio uppadcasting facilities continue to function to the extent that prover transmissions can be made which cover the entire area of the country. This capability is necessary to assure continuity of government, maintain morale, provide news and instructions, and alleviate the sense of isolation which may be prevalent, especially in areas where people are forced to stay in shelters for several weeks due to heavy fallout conditions. Every shelter should have at least one to table transistor radio. It may be necessary to run a wire to the cutside to which the radio antenna is attached in order to obtain the factor for force the shelt r.

There are over four thousand commercial AM broadcasting stations in th. U.S., and most of them are located outside target areas. The two greatest threats of nuclear war to the AM broadcast capability are (1) EM. (electromagnetic pulse) from the nuclear detonations and (2) lack of provision for operation under radioactive fallout conditions. Widespread electrical power failure is not a major threat to the AM broadcast capability because many key stations have been supplied by DCPA with emergency generators, with diesel oil storage for at least two weeks operation under full load. The effects of nuclear EMP on those AM radio broadcast stations which are a part of the EBS (Emergency Broadcast System) have been analyzed by Nelson (1971) and Barnes (1974). Considerable uncertainty exists as to how many stations will remain functional after the EMPs of a nuclear war, but the prospect appears good that there will be adequate AM broadcasting capability within a few hours after the attack, partly due to protective devices and measures taken by some of the stations, and partly due to the usual practice of retaining spare parts on hand which would enable station engineering personnel to make repairs. Many stations maintain two transmitters with provision for rapid switching between them, which reduces the possibility of EMP damage to the transmitter which is not in operation (Nelson, 1971).

In addition to emergency generators, DCPA has provided most EBS stations with fallout protection for the studio, and with radiation survey meters. During the crisis period preceding a potential nuclear attack the studio fallout shelters can be stocked with any provisions which may be lacking, and additional preparations may also be made to prevent or repair damage from EMP.

The EBS will provide much necessary and useful information during the crisis period and also after the attack. One important bit of information which should be given before an attack is the precaution necessary to protect two-way radios from damage by EMP. For example, when tactical warning is given that a nuclear attack is imminent, owners of radio receivers should disconnect or fold down or telescope the antennas on their equipment to make them as small as possible (Barnes, 1974). Typical portable AM receivers do not have a long antenna, and are therefore not vulnerable to EMP. However, if an external wire is hooked to the radio antenna to improve reception inside a shelter, that wire should be disconnected for several hours after tactical warning is given, until the threat of EMP from nuclear detonations has diminished.

### 5.2 Two-Way Communications

Although much of the U.S. land-line communications network has been hardened against blast and EMP, it is not the policy of the Federal Civil Defense Guide to rely completely on telephones for postattack communication. Telephones may continue to operate in non-target areas for local calls, and telephones installed in large community shelters will be extremely useful. Telephones in homes where basements are used for shelters may also be used for early postattack emergency messages by shelter occupants who feel that the risk of brief exposure to radiation is offset by the possible gain achieved by exchanging a message. Some telephone exchanges are built with shielding against radiation so that operators may continue to work in them after a nuclear attack. Instructions given on local AM broadcasts should give information on whether telephones are working, and what numbers to dial for information and assistance.

In recent years, partly because the breakthrough in transistor technology resulting in low costs and compact size, there has been a tremendous growth in the use of CB (Citizens Band) radios, and other types of transmitters, as shown in Figs. 5.1-5.3. Suitably distributed, there would be enough (about 4 million, 1973) to have one in every shelter (about 1.8 million).

The number of transmitters of all types show a sharp increase in 1970-1971, except those mobile units used by industry (see Fig. 5.2), which decreased because of the sharp decline in petroleum exploration by industry due to Congressional changes in the tax structure on the petroleum industry.

A breakdown of transmitters authorized by the FCC in 1973 is listed in Table 5.1 taken from the FCC 1973 Annual Report, p. 298.

It is estimated that about 80% of all semi-trailer trucks are equipped with CB radios, and about 56% of those with radios are licensed, according to a poll by FCC. Portable CB units can probably transmit to the outside from most shelters, and would be especially useful in expedient shelters. If the unit cannot transmit through the shelter walls, then an external wire may be mounted and hooked to the antenna. The same precautions should be taken concerning damage from EMP as described above for AM radios. During the crisis period preceding a potential nuclear attack, one or two sets of fresh batteries should be purchased for each portable radio device. An assessment should be made to determine whether enough batteries exist in retail outlets to fulfill this requirement.

and a state of the state of the
ORNL-DWG 75-14158 ÷ ţ MOBILE CITIZENS LAND PUBLIC SAFETY SERVICES (FIXED IN PLACE) MOBILE PUBLIC SAFETY SERVICES AMATEUR AND DISASTER SERVICES LAND (FIXED IN PLACE) 2200 2000 1800 1800 1600 

の日本の見たいというないというないというないであった。

Fig. 5.1 Citizens Band and Amateur Radio Transmitters, 1960-1973.



99

an to the state of the state o



100

22 124

ላይ የሚሰት የሚሰት የሚሰት የሚሰት የሚሰት በድንዳ የሚሰት የሚሰት የሚሰት የሚሰት የሚሰት የሆኑ የሚሰት የሳት የሚሰት የሳት የሚሰት የሚሰት የሚሰት የሚሰት የሚሰት የሚሰት 

と言いがないたちないないであるとなっていたない。

Table 5.1

Transmitters Authorized by FCC, 1973

Class of Station	Base or Fixed	Mobile	Total
Afreraft oroun		139.486	139.486
Accounties of the fixed around	9 987		9 987
APTOLIGALITICAL ANIA FILA BLOUP		5 5 5	
Aviation auxiliary group	116	د/د,د	0,250
Aviation radionavigation land	2,562		2,562
Civil air patrol	9,334	18,668	28,002
Operational fixed aviation service	61		61
Total aviation services	22.921	163.527	186.448
Police	47,401	1,114,904	1,162,305
Fire	25,487	340,521	366,008
Local government	29,489	307,939	337,428
Highway maintenance	24,531	208,688	233,199
Forestry conservation	40,093	104,849	144,942
Special emergency	8,458	61,852	70,310
State guard	388	1,114	1,502
Operational fixed public safety service	2,659		2,659
Matter and and a second and	705 91	239 061 C	0 318 272
TOLAL PULLES SALENY SELVICES	000,0011	100,501,2	C/C \$ OTC \$ 7
Special industrial	37,290	371,444	408,734
Business	100,176	877,539	977,715
Power	16,718	285,844	302,562
Petroleum	11,986	98,667	110,653
Manufacturers	3,731	68,422	72,153

A. 240 A. 10 MARCH

Table 5.1 (cont'd)

A BIT A STREET OF TASE

Class of Station	Base or Fixed	Mobile	Total
Manufacturers	3,731	68,422	72,153
Forest products	3,079	40,249	43,328
Taductrial vadialocation	18,200	107	18 316
IDAUSLITAL FAG.OLOCALION	10,203	1,598	1,844
Motion picture	246	1,598	1,844
Relay press	349	8,360	8,709
Telephone maintenance	2,393	89,898	92,291
Operational fixed industria! services	14,729		14,729
Total industrial services	208,906	1,842,128	2,051,034
Railroad	14,887	1,145,721	1,160,608
Taxicab	3.722	126,166	129.888
Automobile emergency	2,437	23,102	25,539
Interurban-passenger (motor carrier)	108		10,229
Incerurban property (mocor carrier)	2,130	42, 302	47,500
Urban passenger (motor carrier)	683	18, 873	19,556
Urban pronerty (motor carrier)	2,376	53, 244	55,620
Operational fixed land transportation services	2,509		2,509
Total land transportation services	28,860	1,422,589	1,451,449
Citizens: Class A Classes C and D	1,980 0	12,680 3,828,115	14,660 3,828,115
Total citizens services	1,980	3,840,795	3,842,775

Table 5.1 (cont'd)

「日本のため」と言語の言語を見てい

Class of Station	Base or Fixed	Mobile	Total
Amateur R.A.C.E.S. Disaster	270,694 8,625 186		270,694 8,625 186
Total amateur and disaster services	279,505	1	279,505
Total safety and special services stations	720,678	9,408,906	10,129,584

103

あえぞう

語語の見たりにないのかっているかがからないでいたがあったいでも

and the state of the

and a static structure of the state of the s

A PAR LAND IN THE WAY A

ì

#### 6. THE FOOD AND WATER SITUATION

and a second state of the second s

and and the state of the second state of the state of the second state of the state of the state of the second

a contraction in the second second

## 6.1 Water Supply

For most areas designated as reception areas in the U.S., there are abundant water sources. In many localities, local water purification equipment will be overloaded if evacuees and hosts attempt to use water at the pre-crisis rate of about 100 gal/day per person. Reduction in the use of purified water for bathing and washing will result in an adequate supply of drinking water for nearly all areas. In those few areas where the local water purification equipment will not be adequate to handle the overload, there will be other sources of water from streams and dams which can be purified by mobile emergency water purification units. Chlorine bleaches can be used to make many water sources safe for drinking. At worst, some people may use water from various sources for drinking without purification, and there is a danger that some of these people may consequently suffer from various dysenteries, typhoid, hepatitis, and other water-borne diseases. In general, there will be enough water such that people will not die of thirst in the postattack situation.

There may be a problem in some locations with soluble radioactive components of fallout, such as iodine, which may require special treatment of water by filtration, or distillation, or which may be counteracted by prophylactic measures. Radioactive iodine will not be present in well-water, but may be prevalent in lakes in fallout-contaminated areas (Brown, et al., 1968). If this water must be used it should be filtered. Filtration through about 5 in. of soil will remove the iodine, as determined by Kearny (Private Communication). Extensive research on expedient methods to remove radioactive contaminants from water has been performed by the Corps of Engineers.

If iodine is not removed from the water, doses of stable iodides should be taken orally to block thyroid uptake of the radioactive iodine (Ramsden, et al., 1967). The required doses range from 35 mg every 12 hours to 250 mg every 48 hours, depending on the size and activity of the subject.

PRECEDING PAGE BLANK-NOT FILMED

Calculations by Brown, et al., (1968) indicate that the radiation dose due to consumption of water which contains other soluble nuclides, such as strontium and cesium, would be negligible for a 1600 MT attack but could be harmful for larger attacks. Filtration of water through soil removes these nuclides as well as iodine.

# 6.2 The Food Situation

The problem of supplying the relocated population of the U.S. with food during the crisis period has been investigated by Billheimer et al., (1975). Of the stocks that might be available at regional and local levels for distribution under crisis relocation conditions, Billheimer et al., found that wholesalers have an average of three weeks of inventory, retail outlets have an average of two weeks inventory, and consumers have an average inventory of two weeks, measured against current consumption levels. Food stockpiles under federal control are insignificant. Wholesale stocks tend to be held in regional distribution centers located in the largest cities, in target areas. Nearly twothirds of the urbanized risk areas scheduled for evacuation under a crisis relocation strategy rely on metropolitan areas other than their own for at least 50% of their processed wholesale food supplies. Billheimer et al., concluded that the most effective strategy for food distribution under crisis relocation conditions is to allow agricultural output and major processing plants to follow normal distribution channels and to continue using risk area wholesale facilities to serve the evacuated population. Although this strategy may be the most effective, it will place a stress on transportation to distribute over the extra long lines, and retail outlets in the hosting areas will be strained to find adequate storage space and sales facilities. The transportation system will be further strained by the demand for extra supplies to prepare fallout shelters with reserve subsistence for at least two weeks' duration.

The kinds and relative quantities of food purchased by the typical American family is shown in Table 6.1, based on a survey taken by the U.S. Department of Agriculture.

# Table 6.1

# Distribution of the Food Dollar in Northeastern USA, 1965-66

Food Group by Urbanization	Spring	Summer	Fall	Winter
Meat, poultry, fish, eggs:				
All urbanizations	36	36	38	39
Urban	37	38	39	40
Rural nonfarm	34	33	36	37
Farm	34	30	33	38
Milk, cream, cheese:				
All urbanizations	13	12	14	13
Urban	12	12	13	13
Rural nonfarm	14	12	15	14
Farm	15	13	20	14
Grain products:				
All urbanizations	12	11	12	12
Urban	12	11	11	11
Rural nonfarm	13	12	13	13
Farm	12	13	11	12
Fresh vegetables (excluding potatoes):	c	c	1	1
All urbanizations	2	0	4	4
Urban Bunal aanfann	2	0	4	4
Rural nonlarm	4 5	0 10	4	4
rarm	5	12	0	4
Fresh fruit:				
All urbanizations	4	5	4	4
Urban	4	5	4	4
Rural nonfarm	4	5	4	4
Farm	6	?	5	5
Commercially processed vegetables and fruit:				
All urbanizations	4	3	4	4
Urban	4	3	4	4
Rural nonfarm	4	5	4	4
Farm	3	2	3	4
Patataon exceptions (freeh and pressoned)	•			
All urbanizations	• •	2	2	2
ALL ULDAHIZALIONS	2	2	2	2
Urban Purci nonform	2	2	2	2
Ruiai Nonialm Romm	2	5 4	2 7	2
ralm	J	-	4	2
Fats and oils:				
All urbanizations	3	3	3	3
Urban	3	3	3	3

in the location of the state of the 

a state and the second se

Food Group by Urbanization	Spring	Summer	Fall	Winter
Rural nonfarm	4	4	4	3
Farm	4	4	4	5
Other (beverages, sugar, sweets, juices, etc)	):			
All urbanizations	21	22	19	19
Urban	21	20	20	19
Rural nonfarm	20	20	18	19
Farm	18	15	16	16
Total	100	100	100	100

And the second second

Table	6.1 (	(cont'	d)
-------	-------	--------	----

------

If the CRP-2B attack takes place, wholesale food stocks in risk areas will most likely be destroyed, or become inaccessible for many weeks. Furthermore, according to studies by FPA (Federal Preparedness Agency) with the UNCLEX attack, which is similar to the CRP-2B attack, approximately 60% of the U.S. food processing plants will be destroyed by direct weapons effects, such as blast and fire. Most of the food in the form of livestock and poultry will become unavailable after the attack, either because much of the stock will be killed by fallout, for which little protection is being provided, or it will be impossible to process the livestock because of the lack of transport facilities and/or processing plants at convenient locations. Many of the surviving food processing plants will be unable to operate because of power outages or because of intense radiation from fallout.

The outcome of the cumulation of these effects on the food situation in the U.S. is uncertain. In most relocation areas, the total reserve of processed food will not exceed three or four weeks supply. Some of the surviving processing plants may be capable of tripling their output of products depending on grains, such as flour, corn meal, corn oil, margarine, and secondary products such as bread, cereals, noodles, spaghetti, etc., to the extent that lack of meat and dairy products can be tolerated in the areas which can be supplied by these processing plants. In many areas, the absence of adequate food processing plants may require that people turn to primitive methods of preparation of raw grains in order to survive. Many old recipes have been rediscovered as a result of the movement to communes in the U.S., and some of these and others have been tested and are described by Kearny (1975).

The innate capability of grain to be stored almost indefinitely with little care and without spoilage, and its suitability as a nearly complete food for human consumption, has drawn the attention of strategic planners throughout the entire known history of man. Currently, one of the principal slogans of Mao which is echoed throughout Red China is "Dig tunnels deep, store grain everywhere, prepare for war and disaster. . ." The Soviet Union is also preparing to store massive quantities of grain (USSR National Affairs, 1975), and is currently in

109

はないないないとないとうないというないないないないないないとうです。

artes ar artiste arbite artistic artistic and artistic artistic artistic artistic artistic artistic artistic art

the last five-years stage of a "shock construction" project to build storage facilities for 2.5 billion bushels of grain, enough to supply the Soviet population with 300 days of food.

The U. S. has consistently produced more grain annually than required by its population for human consumption. Since 1956, the total production of corn, wheat and soybeans annually in the U.S. has always been more than twice and occasionally three times the quantity needed annually by the U.S. population for adequate survival requirements, as shown in Fig. 6.1. The amount required per day for survival is estimated to be an average of two pounds of grain per person, based on a requirement of an average of 3000 calories per person per day (Garland, 1972). The extra quantity of grain produced is used primarily for feeding livestock and for exportation. The amount of grain on hand does not always exceed the annual minimum survival quantity, as shown in Fig. 6.2. From 1954 through 1965, the total quantity on hand always exceeded the minimum annual survival quantity, but it became less at times in 1966, 1971, 1973, and 1974, during certain quarters of the year. From Fig. 6.2, it is apparent that 30 to 40% of the annual U.S. grain production for the years 1965 through the present would be adequate for the minimum annual survival quantity.

のため、「ないないない」ので、こので、ないないないないで、このないないで、ないないないで、ない

and a state of the s

Nearly all the grain stored on farms and 50-70% of the grain stored off farms, depending on the season, would not be affected by a nuclear attack. Grains stored off farms is distributed among local town and county elevators, warehouses, processing plants, and large central terminals. Many processing plants and most of the large terminals would be destroyed by the nuclear attack, but the bulk of off-farm storage is in the local elevators in small towns which would not be affected by the blast and fire of nuclear weapons detonated on major targets.

The corn, wheat, and soybeans stocks since 1945 are shown for the January 1 and April 1 reporting dates to USDA in Fig. 6.3, and for the July 1 and October 1 reporting dates in Fig. 6.4. The quantities reported on these dates reflect the actual situation of two to three months preceding the date. Thus, Fig. 6.3 indicates that through the winter months, the stocks on farms alone are usually sufficient for the





undelite that as the second second second the product of the second second second

Fig. 6.2 Corn, Wheat, and Soybeans Stored On and Off Farms as Percent of Total Production.



and the second secon

manin , a 141 a have a firm

Fig. 6.3 Corn, Wheat, and Sovbean Stocks, January and April.



「中国」の「日本」」というのでは、「日本」の「日本」

1.6. States and a state of the state of the

2. 東京市 たんざい 空东区部市地方人口 計算



minimum annual survival quantity. The peak total on- and off-farm quantity of grains stored was 410 billion pounds, or about 6.8 billion bushels, in 1961 and again in 1972, and the peak on-farm quantity of grain stored was about 265 billion pounds, or about 4.4 billion bushels, in 1972. These peak numbers are an indication of the total grain storage capacity in the United States. The current trend for on-farm storage capacity (<u>Feedstuffs</u>, <u>47</u> (48), p i, November 24, 1975) is to increase it significantly, because farmers are becoming increasingly aware of the global market for grains with its associated large variation in price, and increased storage capacity enables them to hold their grain for the most favorable price. The price of wheat per bushel more than doubled in 1973, as shown in Fig. 6.5, primarily as a result of large sales to the Soviet Union.

Total grain stocks during summer and fall months, as shown in Fig. 6.4 have been adequate for a year's minimum survival for the years 1954-1966, but not during the years of 1974 and 1975. On-farm stocks by themselves have been adequate for at least six months' survival supply for the nation during the spring and summer months for all years since 1954. If a nuclear attack occurred during the spring or summer, the stocks on farms would be adequate for survival through September or later, and the harvest, although partially destroyed by fallout, would provide adequate food reserve through the next harvest.

Plant responses to irradiation, such as survival, growth inhibition or stimulation, and seed or grain yield, depend on dose, dose rate, plant species, type of radiation, and the developmental stage of the plant when irradiated. Killion and Constantin (1975) calculate that the dose required to reduce the yield to 50% is 2-4 kilorads in wheat and corn, and 8-12 kilorads in soybeans. Crops are planted at different times at different latitudes, hence if the attack occurs on a certain day, simultaneously as far as crops are concerned, crops which have just emerged irom the ground will be more severely affected than those which have attained a good stand, or those which have not yet emerged from the ground.

A complex computer program has been developed by Ryan, Garza, and Brown (1974) for the purpose of estimating damage to crops by beta and

115

and the second second second second second second second second is shown, when so is a contraction of the second second

うち コン・シャンプ なるななななな



116

and the second second second second second

gamma irradiation from fallout from nuclear weapons. Two runs of this program were made by the Federal Preparedness Agency of the General Services Administration, one for each of two different hypothetical attacks, each occurring on June 1, and with crop data corresponding to the year 1974. According to Brown and Pilz (1969) the effect of fallout on the total U.S. crop production should be the most severe for an attack in the month of June. The attacks are very similar to the CRP-2B attack in terms of total number and yields of weapons but differ in distribution of the weapons on the United States. The "UNCLEX-MIKE" attack concentrates on military targets, and "UNCLEX-CHARLIE" concentrates on civilian targets. The wind conditions assumed for distributing the fallout corresponded to a typical meteorological situation for the month of March.

A summary of the results of the computer calculations for estimating surviving yield of crops is shown in Table 6.2. Note that the total estimated surviving yield is in the vicinity of 50% of the total production for 1974. Soybeans and sorghums are shown to have a high estimated yield because of a combination of factors, including their high resistance to damage by irradiation (Ryan, Garza, and Brown, 1974, p 34), the time of their growing season, and the distribution of their planting in relation to location of fallout from these attacks. In order to investigate the sensitivity of the calculated surviving crop yield with changes in the agricultural model, we assumed a crop yield factor, Y, for each county, with values ranging between zero and unity, as a simplistic function of  $R_0$ , the unit time reference dose rate, varying according to

$$Y = \alpha - \beta R_0^t$$

The factors  $\alpha$  and  $\beta$  are calculated by assuming boundary values of  $R_{o}$  for Y = 0 and Y = 1. Under these conditions, the equation provides essentially the identical curve as Y =  $\alpha - \beta \ln R_{o}$  for t  $\leq 0.01$ .

Curves of Y vs R<sub>o</sub> are shown in Fig. 6.6 for the three boundary conditions shown in Table 6.3. Curve "B" shows Y = 0.5 at R<sub>o</sub> = 500 R/hr,

2
•
9
e
-
Д,
5
<u> </u>

,

A DISTURBANCE

A CALL AND A CALL AND

# Estimated Surviving Yield of U.S. Crops

Crop	Total Production	(1974)	Estimated Sur INCLEX-C	viving Yield HARLTE	Estimated Sur	viving Yield -MTKE
	(Bushels)		(Bushels)	(Percent)	(Bushels)	(Percent)
Barley	343,600		134,100	39.0	142,700	41.5
Corn	4,833,300		1,890,000	39.1	1,376,100	28.5
Irish Potatoes	413,100		187,100	45.3	175,700	42.5
Rye	28,500		8,100	28.5	7,800	27.3
Sorghums	908,800		851,400	93.6	749,500	82.4
Soybeans	1,269,800		1,269,800	100.0	1,269,800	100.0
Wheat	1,805,100		883,000	48.9	956,000	53.0
Totals	9,603,200		5,223,600	54.5	4,677,600	48.7

. . . . . .



\*\*\*\*

ć

むするなというときに、そうないないのですが、

corresponding to about 2000 roentgens exposure during a sixty-day growing time, at areas where the fallout arrives within one hour after the detonation. This exposure is approximately that amount required to reduce the yield of corn and wheat at maturity to 50% according to Killion and Constantin (1975). The yield would be greater than 50% at most places because the exposure to radiation would be less than 2000 R for these reasons:

- 1. The fallout will arrive <u>later</u> than one hour after detonation at most locations,
- 2. At northern latitudes, some seeds in the ground may still be ungerminated at the attack time, and thus less affected during the period of the most interse radiation,
- 3. In southern latitudes crops may be well developed at attack time and may be harvested before the period of 60 days is up, and
- 4. Rainfall may wash away significant amounts of fallout during the growing period.

The yield of grains for 1973 was estimated by determining  $R_0$  at the centroid of each county and then multiplying the production of grains for 1973 for that county by the value of Y corresponding to  $R_0$ . The results are shown in Table 6.3, giving 68% survival for the relatively insensitive case, 42.5% survival for the sensitive case, and 30.2% survival for the highly sensitive case.

For planning purposes, we will assume that grain stocks on farms and in rural elevators will be the major source of food required for survival in the postattack period. As we have seen from the data presented, this source is adequate to supply the relocated population until the following harvest, under most circumstances readily foreseen. We also assume that these grain stocks are located primarily in the counties in which they were produced. The location of grain-rich and grain-poor counties in the U.S. can be rapidly assessed by the maps shown in Figs. 6.7-6.13, which show the total quantities of all major grains produced in each county of the coterminous U.S. in 1973, according to USDA figures. A comparison with Figs. 3.5-3.11, which show the

and the first of the second second

Table	6.3	
-------	-----	--

# Total U.S. Grain Production for Three Different Sensitivities to Fallout Irradiation

The share were

Sensitivity	Y=l (100% Yield) at R <sub>o</sub> Listed Below (R/hr)	Y=O (No Yield) at R <sub>o</sub> Listed Below (R/hr)	t	Calculated Percent Survival (U.S.)
Relatively Insensitive	200	10,000	0.01	68.0
Sensitive	50	1,000	3/4	42.5
Highly Sensitive	50	500	3/4	30.2

a second a strand a second a second second



Fig. 6.7 Grain Production, 1973, Northeast, Based on Data Supplied by the Crop Reporting Board, USDA. In the legend, as an example, 1.0E6 indicates 1.0 x  $10^6$  pounds, or one million pounds.

ORNL DWG 76-16522



and the state of the

a Salaria

ORNL DWG 75-16518

ORNL DWG 75-16919



Grain Production, 1973, South ast, Based on Data Supplied by tie Crop Reporting Board, USDA. In the legend, as an example, 1.0E6 indicates 1.0 x  $10^6$  pounds, or one million pounds. Fig. 6.9

and the state of the



A STATE OF THE STA



Fig. 6.11 Grain Production, 1973, South Central, Based on Data Supplied by the Crop Reporting Board, USDA. In the legend, as an example, 1.0E6 indicates 1.0 x  $10^6$  pounds, or one million pounds.

126



はないたいないないないとなっていないできたのではないないできたのではないできたので、





のはななななない。



ないというないで

ORNL DWG 75-16524

distribution of relocated people by county according to the ADAGIO program, reveals that some regions, such as peninsular Florida, Northeastern U.S., and parts of California, are deficient in grain stocks but have a large number of people. This picture indicates that large quantities of grain may have to be shipped in order to avoid massive starvation. We assume that grain will be moved rather than the people, in order to avoid exposure of a large number of people to radiation from fallout. This situation could have been largely avoided if the evacuation plans had taken into account the location of grain stocks in addition to fallout avoidance and hosting capability, or if the grain had been moved and stored before the crisis, as recommended by Garland (1971).

In order to conserve fuel and to reduce exposure of operators to radiation, it would be desirable to minimize the total ton-miles in the shipments of grain. The solution of this problem involves linear programming of a typical transport problem, and will be presented in the next chapter on transportation.

W> have assumed that the privately owned stocks of grains by farmers can be obtained by the government for relief of potential starvation of large fractions of the U.S. population in a postattack situation. Some form of guarantee should be given by the government to assure the farmer of just compensation for his labor and to allay his feelings of anxiery for his own future. Such a guarantee may be difficult to produce in a postattack situation unless it is evident to the farmer that the federal structure of government remains firmly in power. AM radio broadcasts of news and information, with frequent reassuring messages from the President would be necessary to convey this information (see Chapter 10).

# 6.3 The Effects of Dust and Depletion of the Ozone Layer Due to Nuclear Attack

It has been calculated that a  $10^4$  MT nuclear exchange would inject  $10^7-10^8$  tons of dust into the stratosphere (NAS 1975), which is of the same order as the amount estimated to be injected into the stratosphere by the volcanic eruption of Krakatoa in 1883. The total volume of earth

and rock thrown into the sky by Krakotoa is estimated to be around 13 cubic miles. Thermonuclear explosions also produce NO, injecting approximately  $10^4$  tons of NC per megator into the stratosphere, (NAS 1975) which may have a climatic effect as a result of ozone depletion and also as a result of formation of NO<sub>2</sub> in the atmosphere.

It appears that volcanic injections of the magnitude of Krakatoa may lead to minor cooling on a hemispheric or global average scale, but this statement must be qualified because the global mean temperature shows a variety of fluctuations on different time scales, not all of which can be explained in terms of volcanic injections. According to the Committee to Study the Long-Term Worldwide Effects of Multiple Nuclear-Weapons Detonations (CSLWEMND) (NAS 1975), a deviation of  $0.5^{\circ}$  C from the average lasting for a few years might be expected from the stratospheric dust injection from a  $10^{4}$  MT nuclear exchange.

The effect of NO injection into the stratosphere was also investigated by CSLWEMND, with the conclusion that the detonation of  $10^4$  MT of nuclear weapons in the northern hemisphere would result in a maximum reduction of czone in the stratosphere in the range of 30 to 70%, beginning a few weeks after the nuclear exchange, and gradually restoring to within 10% of the normal ozone content within 5 to 7 years after. The model used for this calculation is highly limited, and data for verification do not yet exist in sufficient quantity or with sufficient quality, hence the estimated range of uncertainty for the depletion of ozone varies within a factor of 2 or 3.

If the ozone layer is substantially depleted, much of the solar radiation in the wavelengths from about 2500 A through 3300 A, which would normally be absorbed by the ozone, will now pass through to the earth's surface. A 5% decrease in the average ozone concentration would cause a 26% increase in uv-B radiation (2800-3150 A); a 50% reduction in ozone would produce a fivefold to tenfold increase in uv-B radiation.

Although reliable data are merger with regard to effects of uv-B radiation on biological organisms, and the responses are variable and often subtle, the biological implications of increased uv radiation at the earth's surface are considered to be far-reaching by CSLWEMND. They

130

A share the week is the briese of a real of the two has been a

「おちちょういうないないないないない」ないのないできょう

and the second states of the second of the second second second second second second second second second second

and a state of the state of the

conclude that some crope --corn, soybeans, barley, and alfalfa--would be affected only slightly by a fivefold to tenfold increase in uv-B radiation. Others, such as tomatoes, peas, beans, and chions would be severely "scalded" and even killed. The committee did not consider the synergistic effects of the combination of fallout radiation, increased uv radiation, and reduction in mean temperature, which could possibly result in much more severe reduction in crop yield than caused by any one of the effects alone.

The U.S. surplus capacity plus the ability to switch to the more resistant crops should enable this country to .eed itself despite these kinds of ecolog.cal upsets. The critical factors for the U.S. agriculture will be the supply of fuel, fertilizer, and pesticides. Principal casualties from the possible ecological disturbance will be in countries with marginal to inadequate agriculture, especially if they are dependent on U.S. exports. ないでするななななのないです。

#### 7. TRANSPORTATION FOR POSTATTACK SURVIVAL

# 7.1 Introduction

The U.S. has the greatest transport capability of any nation on earth. In 1970, the total volume of intercity traffic amounted to almost two trillion ton-miles, as shown in Fig. 7.1, of which the railroads transported about 40%, oil pipelines about 22.5%, motor vehicles 21.3%, inland waterways 15.7%, and aircraft (not shown in Fig. 7.1) about 0.5%.

The escimated number of privately and publicly owned transport units in the U.S. in 1972 is listed in Table 7.1. After a nuclear war, most of the motor vehicles and aircraft transport capability will remain intact, some rail and inland waterway transport capability will continue to exist, but oil pipelines will probably not be able to function at all for some weeks or months after the attack. In the following sections, each mode of transport will be briefly considered as ro its survivability and functional contribution to general survivability. Finally, we will discuss the transport model for redistributing grain stocks to prevent starvation.

### 7.2 Oil Pipelines

Pipelines in themselves, being mostly buried and passing through sparsely populated areas, are fairly invulnerable to nuclear attack. However, the terminals are usually located in target areas, most pumping and flow control stations are operated by remote electronic controls which are vulnerable to EMP (Stephens, 1973), and the pump motors are completely dependent upon commercially supplied electrical power for their operation. When pipeline terminals are destroyed by nuclear blast, there will be further damage caused by the fires from spilled petroleum.

In 1972 there was about 172,000 miles of pipeline in the U.S., and the total deliveries for the year amounted to almost nine billion barrels, with a total trunkline traffic of about 2.7 trillion barrel-miles (The Oil and Gas Journal, June 11, 1973). Much of the oil transported by

# 133

# PRECEDING PAGE BLANK-NOT FTIMED



のため、日本市は市政部では、日本市政部であるとのである。日本市政でので、日本市政部では、1994年1日

NA NUMBER OF STREET, ST

N. Contraction

Volume of Domestic Intercity Traffic (from *Statistical Abstract of the United States* 1972).

Fig. 7.1 Volume of Domestic Intercity Traffic.

Table 7.1

になると

5.00

Estimated Number of Privately and Publicly Owned Transport Units<sup>a</sup>

	1.950	1955	1960	1965	1970	1972
Motor Automobiles Commercial Buses <sup>b</sup> School & Other Buses Motor Trucks <sup>c</sup>	40, 339, 007 89, 000 134, 652 8, 598, 962	52,144,739 80,000 175,249 10,288,804	61,682,304 75,369 196,760 11,914,249	75,251,386 83,883 230,401 14,795,051	89,279,864 88,823 290,198 18,748,421	96,859,746 88,722 318,232 21,239,163
Rail horight Train Cars Passenger Train Cars Locomotives	2,032,966 43,585 42,951	2,018,398 36,894 33,533	1,984,639 28,396 31,178	1,816,426 21,516 30,061	1,799,342 11,378 29,122	1,716,937 7,763 29,338
<u>Air</u> Single Engine Twin Engine Four Engine	57,353 2,827 30 530	55,221 4,101 15 738	68,036 8,241 8	81,192 12,547 180 1,488	109,673 18,921 673 1,144	116,898 20,352 775 1,273

135

......

A STATE AND A STATE AND A STATE AND A STATE AND

のないのであるのであるというないできたが、こころのであるというであるという

<mark>terika di di</mark> kanan seberahan di kananan dan mananan dan kananan di kananan ana mananan mananan kana manang dan d

20**11 100 100 10**0

. million te. a. . . . . . . .
(P
cont
7.1 (
able

ć.

	1950	1955	1960	1965	1970	1972
Water						
Barges	13,805	14,875	16,777	17,033	19,809	22,245
Towboats	4,042	4,163	4,203	4,054	4,230	4,064
Ships	1,099	1,072	957	866	764	622

<sup>a</sup>Number of units fails to recognize greater capacity and utilization of newer equipment.

b<sub>Includes municipally owned transit units.</sub>

<sup>c</sup>The 1963, 1967, and 1972 Census of Transportation disclose, of the trucks surveyed, only a relatively small number were of the larger sizes, i.e.:

%	5.1@
Miscellaneous Sizes	615,825
%	6.1 7.6 7.6
Over 26,000 lbs.	736,750 1,089,720 1,500,000
%	10.4 5.2 4.2
20,000 to 26,090 lbs.	1,255,800 739,440 828,000
2	78.4 87.2 88.3
Under 20,000 1bs.	9,466,800 12,399,840 17,420,000
	963 967 972

SOURCE: <u>Transportation</u>, Eleventh Edition, Transportation Association of America, December 1974.

いいとれたのいないというというないないないないないとないというないないない

pipelines is used for heating, and may become an item necessary for survival in some locations if the nuclear attack occurs in the winter. Two pipelines, the Plantation and the Colonial, begin in Louisiana and serve the eastern seaboard, supplying about two-thirds of the oil for the New York-New Jersey area. If these two lines were severed by a nuclear strike, it would require from 200 to 250 T-2 tanker equivalents to move this liquid from Galveston, Texas, to New York, a thirteen-day round trip including loading and unloading, according to Stephens (1973). This operation "would use all of the U.S. tankers, and the ports at each erd, if they are not destroyed by the attack, would be so crowded that a complete traffic jam would result." It is evident from these comments that it would be desirable to construct bypass pipelines around major target areas, and to harden the control system to EMP.

Although oil may become necessary for survival of people in some situations, such as, for example, those in northern hospitals in the winter, this oil could be delivered by means other than pipelines, possibly by tanker trucks which delivered 28% of all petroleum in 1972 (Transportation Association of America, 1974). Healthy people in shelters may have provided an improvised wood-burning stove in their shelter during the crisis period, if it occurred in the winter. A design for an efficient improvised wood stove for cooking is described by Kearny (1975), as well as means for using newspaper wrapping around the body which will keep a person warm in subzero weather.

Other requirements for petroleum supplies, such as diesel fuel for trains and trucks to nove grain to avert starvation, will exist in sufficient quantity in reserve stocks, and will not require the oil pipelines to be in operation immediately. For economic recovery it will be essential to restore the pipelines to operation as quickly as possible, but we do not believe their operation will be necessary for survival in the first few weeks following a nuclear artack.

# 7.3 Inland Waterways

Tugs, barges, and freighters which ply the U.S. inland waterway systems can be moved out of high-risk areas during the crisis period.

However, if the CRP-2B attack occurs, these vessels will have limited use for several weeks, perhaps months after the attack, because of the destruction of critical locks and ports, and obstruction of waterways by fallen bridges. Except for Lake Michigan and Lake Superior, the Great Lakes will be isolated from each other, from canals leading to inland waterways, and from the St. Lawrence Seaway, as far as waterborne commerce is concerned. About 28 of the 67 major ports of the Great Lakes will be destroyed, and these 28 handled about 70% of the total commerce through the Great Lakes ports in 1971 (World Almanac, 1974). The Mississippi and the Ohio rivers will each be fragmented into seven pieces for water traffic by destruction of locks. Along the Atlantic and Gulf coasts, 65 of 71 major ports will be bombed according to the CRP-2B attack, and these 65 ports handled about 945 million tons out of 965 in 1971, or about 98% of the commerce to Atlantic and Gulf coast ports. We conclude that the inland waterway system will play a very minor role if any, in the early postattack survival period.

# 7.4 Railroads

Rolling stock can be moved out of high-risk areas during the crisis period. Most of the major switchyards, warehouses and repair facilities are located in major cities and will be either severely damaged or destroyed in the CRP-2B attack. Previous studies (Hamberg, 1969) indicate that "in no case would rail traffic be completely blocked" by destruction of "rail activity centers," although circuitous routing would be required to get around the damaged facilities in several cases. The major restriction to rail shipments in the early postattack situation will be destroyed bridges across major rivers such as the Mississippi, Missouri, Ohio, and Tennessee. A few temporary railroad bridge construction sets may exist for military use, and railroad ferries exist at a few river locations, but these will probably not be adequate to rely upon. Shipments of crucial supplies across rivers during the first few weeks or months after the attack may have to be accomplished by unloading boxcars at transfer points along the rivers, where the materials are then transported by trucks across the rivers on portable pontoon bridges.

The railroads are equipped to carry grain in large quantities, and loading and unloading equipment for grain in and out of boxcars exists throughout the grain-producing rural areas. The lower cost and greater efficiency of rail movement at about 1.62¢ per ton-mile, 200 ton-miles per gallon of fuel (Hirst, 1972) as compared with 8.24¢ per ton-mile and 58 ton-miles per gallon for trucks (1973 prices), also makes the railroads appear attractive for the major bulk of emergency shipments in the postattack situation. The railroads and the trucking industry will be required to play a major role in grain shipments in the postattack situation to avoid starvation in many areas.

# 7.5 Commercial Aircraft and Airports

In 1974, the United States operated more than 12,000 airports, of which 3480 had asphalt or concrete runways of 500 feet or longer, as shown in Fig. 7.2. The data base for this information was obtained from FAA. Runways of 8000 feet or longer are usually associated with SAC facilities, and would probably become prime targets. Of 126 runways in this category, shown by circles in Fig. 7.2, all but 33 are borbed by the CRP-2B attack, indicating a possible oversight in the attack planning. The number of runways in different multiples of 1000 feet is shown in Fig. 7.3, where it is shown, for example, that there were 1198 airports with runway lengths between 3000 and 4000 feet in 1974, in the coterminous U.S.

In 1971, there were about 134,000 active aircraft in the U.S. (Statistical Abstracts of the U.S., 1974), of which 2700 were air carriers, and about 109,000 were single-engine fixed-wing aircraft.

CAP (Civil Air Patrol) aircraft will play a crucial role for survival in the first few weeks after a nuclear attack, accomplishing aerial assessment of damage, checking highways for obstructions and destroyed bridges, and conducting aerial monitoring of radiation from fallout. As mentioned before, 1100 aerial monitoring kits have been distributed to the states for mapping radiation fields from fallout by CAP aircraft.

Aircraft belonging to CRAF (Civil Reserve Air Fleet) and WASP (War Air Services Program) can be used for evacuation of people from disaster

Children Chi





Service States

and the state of the state to be atting on the second state of the state of the state of the state of the state

n, **18, 18,10** (defen and frighting and Astronicki kierzani ožesti an 18,10 (1831)

141

See Store 1.18

State of the state

BARAND FRANKLASSING STATIS

and the spectrum and

areas, or from areas heavily affected by fallout, or they can be used for emergency airlift of supplies, or for aerial monitoring. Their capability for accomplishing these functions are briefly assessed in Appendix A.

Most of the airports have some fuel storage capacity, as shown in Figs. 7.4 and 7.5. About 95 million gallons capacity of the total of 139 million gallons exists on the 126 airports which have runways of 8000 feet or longer, and 91 of these airports are targeted by the CRP-2B attack. Because these runways are associated with SAC, we will assume that they are destroyed. The remaining airports have a storage capacity of about 44 million gallons. According to an inventory by the National Petroleum Council, 1974, the average quantity in the tanks is about 45% of full capacity, hence we might expect about 20 million gallons of aircraft fuel at airports to survive the nuclear attack. In 1971 the total fuel, gasoline and jet fuel, consumed for general aviation was 734 million gallons. At this rate of consumption, 20 million gallons would last for about 10 days; however, in the postattack situation, air travel and air shipments would be reduced to absolutely essential trips, and this fuel might be stretched to several weeks supply. After several weeks, fuel from other sources may be brought in by tanker trucks.

A priority listing of essential aircraft missions should be established for the postattack situation. The highest priorities should be given to light aircraft engaged in reconnaissance: (1) to assess damage; (2) to determine which surface transportation routes are open for traffic; and, (3) to monitor radiation intensities from fallout along transportation routes.

# 7.6 Trucks

There is a trend for truck terminals to be located along interstate highways outside of urban areas. Because of this trend, and because of the mobility of the trucks, we estimate that 60 to 80% of the current trucking capability would survive the CRP-2B attack, providing that adequate measures are taken during the crisis period to provide protection during the attack and after. ፦ 2. ራስት በዚህ የአስፈት ድፅ መድግ የሆኑ ስር በም ውስ እን

~ こうから、 なーズがあるいため

and the second second with the second s

10<sup>5</sup> ORNL-DWG 75-11293 П H (<u>0</u> ß E i N (12) ş **4**0**4** AIRCRAFT FUEL STORAGE CAPACITY PER AIRPORT (100 gal) (80) T ιΩ (112) Fig. 7.4 Distribution of Fuel Storage Capacity. ł 1 (105) N 103 (185) (193) ŝ 11 (810) 2 10<sup>2</sup> -<u>(503)</u> (336) ++-11 ß 11 N ō (06) ß ΠΠ (33) ີຄ 2 100 1000 500 <u>0</u>0 50 20 ę ഹ 200 N NUMBER OF AIRPORTS

143

223.51

adardara ta ta adalah kata kata ta ta

an. A statistick at the second states at the states of the states of the states and states at the states of the state A states of the states of the

うちょうできょうである

Saist int





Fig. 7.5 Cumulative Fuel Storage Capacity.

そうないないないであるとないないないできたいできょうであるというできょうできょうないないないないできょうとう しょうしょう

and the set of the second states with the second second states and a second second second

「「「「「「「「」」」」

Most truck terminals contain sleeping quarters for truck drivers which would provide very little fallout protection. During the crisis period it would be highly advisable to construct expedient shelters at truck terminals in which drivers can rest and be protected from fallout radiation.

In 1972, trucks consumed about 8.1 billion gallons of fuel in transporting 470 billion ton-miles of intercity freight, about 8% of the total motor fuel (105 billion gallons) consumed for all highway traffic (Transportation Association of America, 1974), at an average rate of 22.2 million gallons and 1290 million ton-miles per day for trucks. The average quantity of distillate fuel oil, including diesel fuel, stored in tanks in 1973 was about 5.67 billion gallons, and the average quantity of gasoline was about 7.39 billion gallons (National Petroleum Council, 1974). We estimate that about 60% of these fuels will be destroyed in the CRP-2B attack, which would leave about 2.27 billion gallons of discillate fuel oil and diesel and 2.96 billion gallons of gasoline surviving, if these numbers were representative of the fuels in storage at the time of the attack. If this total of 5.23 billion gallons were to be used only by trucks in the postattack situation at the same average rate as before the attack, i.e., at 22.2 million gallons per day, this supply would last for 236 days, which is probably adequate to carry the nation through the survival state into the recovery stage. Part of this fuel would be used by railroads and automobiles.

# 7.7 Postattack Shipment of Grain Stocks

We assume that within two or three weeks after the attack most of the food on hand for the relocated population will be consumed; that 60% of food processing plants and warehouses are destroyed; that grain stocks will have to be shipped to the relocated people where processing for human consumption will take place, either by surviving and operating processing plants or by emergency primitive techniques; that 60% of the year's production of grain remains available and undamaged on farms and in grain elevators located within or nearby the counties in which the grain was produced; and that 89.6 million people are relocated according いいいでの見たの

to the ADAGIO program and cannot readily relocate in the first few weeks after the attack because of fallout conditions, lack of a place to go, and lack of fuel for travelling.

The location of people by county is shown in Figs. 3.5-3.7, which may be compared with the distribution of grain produced per county, as shown in Figs. 6.7-6.13, for the year 1973. If, in addition to the assumptions above, we assume that two pounds of grain per person per day is adequate on the average for survival (Garland, 1972), an analysis of the data shows that 143 counties, containing a total of about 19.8 million people, mostly relocated population, have no grain stocks whatsoever; in other words, the number of food-days available from local grain stocks is zero, as shown in Fig. 7.6. In other locations, about 26 million people are located in counties in which the local grain stocks can supply zero to 5 days of food supply; about 8.7 million are in counties with 5 to 10 days food from grain stocks, and so forth, as shown in Fig. 7.6.

The problem of calculating the shipments of grain from surplus areas to deficit areas such that the total ton-miles is minimized is a typical problem of linear programming (Hillier and Lieberman, 1974). The particular problem at hand was solved with an unpublished program called TRANSPORT, developed by Brady Holcomb of the Computer Sciences Division of Union Carbide Nuclear Division, and modified by Gary Westley and Philip Coleman. A formal statement of the problem is given in Appendix A.

When counties are used as the basic area cell, the requirements for solution of the problem with the TRANSPORT program exceeds the memory capability of our large computers, hence the county information was consolidated into BEAs (Business Economic Areas) as defined by the Department of Commerce. There are 171 BEAs in coterminous U.S.A., compared with about 3300 county-type div\_sions. The distributions of people, as relocated by ADAGIO, and the production of grain in 1973, are shown by BEAs in Figs. 7.7 and 7.8. A BEA was considered to be a source if 60% of the 1973 grain production in that BEA provided more than 365 days of food for the relocated ADAGIO population in that BEA, and shipments from a source were terminated when the reserve was depleted to the



a managina ang kanang kanan

ORNL-DWG 75-12227R

a la ser ser a ser ser ser ser la station de de de de de la ser se ser de la ser de la ser de la ser ser de de

ORNL DWG 75-17537



148

Fig. 7.7 Distribution of People, Relocated According to ADAGIO, Shown by BEAs.

「「「「「「「「「「「」」」」」

ijų.





<mark>di zaka</mark>ti kirista zenediki disika disikan katika kenandi katuna untutu - muun anaa kirista <mark>kanan kanan kiri</mark>di

level of 365 food-days. Distances were computed along a great-circle on the earth's surface connecting the geographical centroids of the BEAs.

A solution of the grain shipment problem for the first six weeks is shown in Table 7.2. The total shipments amount to about 91,000 tons the first week, and increase to about 305,000 tons by the sixth week, as grain reserves in deficit BEAs are used up. The number of train carloads or truckloads or barges which would be required to ship the grain if only one mode of transportation were used is shown for each week in Table 7.2. The number of train carloads (or truckloads) for grain shipment increases from 1663 (or 5078) in the first week to 5785 (or 17,672) in the sixth-week, assuming that a closed or covered railroad car (adapted for emergency use) averages 55 tons per load, and a closed truck averages 18 tons per load. In comparison, the number of railroad carloads and truckloads per week in 1970, considering only those vehicles suitable for hauling grain, averaged about 266,000 and 308,000, respectively. The sixth week grain shipments in Table 7.2 would require about 2% of the 1970 railroad capability, or about 6% of the 1970 trucking capability, if only one mode of transportation were used. Due to inefficiencies in communication and control, a requirement for a much larger percentage of the U.S. transportation capability for shipping food supplies should be anticipated. Actually all three modes of transportation would be used to some extent, although the use of barges would probably be highly limited during the first few weeks because of the destruction to locks and port facilities, and obstructions by fallen bridges. Barges are extremely useful in hauling bulk cargo such as grain, averaging about 1100 tons per load, equivalent to about 20 average railroad carloads.

An analysis of alternative grain distribution systems has been made by Ladd and Lifferth (1975) for the purpose of optimizing the peacetime shipment problem. Their model could possibly be adapted to show changes which would improve the postattack shipment problem for the relocated population.

Note that in Table 7.2, the heaviest shipments are from the Baltimore BEA, #17, to the New England BEAs, in which 26.44 million people are

COME CALC 75 17631

TABLE 1 2 Possible in Crain Chipmerto en Seficer Prod Arone

on the set of the set of the

i. E

							1		1	1	Ì	3	•						: :	•				
			1     		1		3   	And Web		_	Piter Ver	1			Parents Mar				;		1	1		
BLA BAUTO			1	' <b>1</b>	, 2   2			Ē		1	:		ء 	;	F	1		:	i i		-	, <b>F</b>		
	1	(***		5	1	_	=	1	1	1	+ = 	• •••	-+	•	•	- •	•		•	•		ŀ	:	ł
	V Transfer Prioribura II	ž	-	\$	•			"Manatod		ţ	Tre Tahauan		-	fourt	Farmered			2	1					
		3		1	410		- -	7	•		3	÷	:		, ,	•	:			•		•	•	
In Jun one of the			ł		-					5	0.1		•	1		•			•	•			•	
143 Jucone, A2	144 Sao Biopo. Ct	2	2055	2	5				•								-	-			ł			
17 Battante. P	L Frim, M	÷								1	-		-	ź		-					ſ		-	
		â	11111							-			2	:			: : -			•	:			
			TIME			-			• -	-			1	-			10.11			•	E.			
				5			;			;			-			_	141			2	ŧ			
-		1	K I		[					5			-	9		-	5			•	ñ			
	1 banger, M	5	9			-	-		-	2							1.00			;	ş		-	
	11	2	•			*	<u>.</u>	1	2				Ĩ.					-						
	alson-femaled-free of		•	-	•					•••	2	;		×	142 - 143	<u>.</u>	-			~				
		1		-			•			•				•			1	ł		5	ĩ			
				-	•					•				•			د				111 041	167	•	:
	2 Byreewed, M	<u>.</u>	•	_			,						_	5			Intel			•	Q.			
41 Alberg, CA	31 Tampa-51 Pototoburg, PL	2 2	91151			=	ŝ							-						•				
•••	34 Mart. 7.	**	114		•	-				ł	_	-	•	-										
		W.	1111		1.4.1		5 10 10	0101 B	;	3	*	0102		\$	102 854		é.		• •	-	Į	2	-	;
				(11)	(3070)				•	8	2	:	2	8	· • • • • • • • • • • • • • • • • • • •		5	•	i	-	ð	:		
		1		-	-		1421 (441	(2428) (4.	1111	M.1.	ī	ş	•	•	1		111	2	÷		í	:	-	
110 Vec. TI	110 121101			-	-					1021	8	ĩ		2	15 12	•	110			-	240	-		-
te Columna, ca	st clertaburg. w	<u>.</u>	_			,							1.11	-			12.44	:	ŧ		ļ			
-	1 P.10101 1	2															•			•	1120			
	32 MINUMANANA . V. 51. 00	2					•						i			-			-	•	ł		-	
ter Presto. Co	114 Al. aquat que . W	ê											~'	1.1		- -		-	:					
		3											Ł	0 144	***) (120f	200		:	2	•	ŝ	-		-
10 Law Chaire, VI					~ ~								_ ***				(1011)		(10)	1 1 2021		2 + -		.'
J	44 Atlants CA	:	-							_	_		-	-	•		-	-		4	471) (MA	1.01 M	1	:
			-1			-							-i		! ,	_1	-1	- 4 -4	1	-				
The waterways connectly	ing source and destigation																							

() weekers in perontheses are totals

gerrout and a structure of the structure of the structure structure and a structure of the structure of the structure and the structure of the s

151

located according to the ADAGIO distribution. In the seventh week (not shown in Table 7.2) the grain reserve in the Baltimore BEA is exhausted, and shipments to the New England BEAs must come from other more distant BEAs, such as Norfolk. VA. The solution for the shipment problem beyond six weeks may be meaningless for two reasons: (1) people may begin to move out of the relocation areas, and, (2) the Mississippi River may become open to barge traffic, and unloading facilities for ships and barges along the Atlantic coast may begin to operate, and grain shipments from the grain belt may become feasible. A complete solution of the grain shipment problem, involving costs of each mode of transportation and actual distances of travel, is infeasible because the precise location and degree of damage to facilities and the limitations imposed by fallcut, debris and damaged bridges is unpredictable.

The solution of the grain shipment problem indicates that, within the assumptions described earlier, grain can be shipped to alleviate food shortages well within the capabilities of the surviving transportation facilities and petroleum.

のですというという。そのないというというないのですないできたのであるのであったとうないという

### 8. PETROLEUM

Of 224 refineries in the U.S. in 1973, as shown in Fig 8.1, 136 are destroyed in the CRP-2B attack, and these 136 had 79.7% of the total U.S. crude refining capacity in 1973, the total being about 14.6 million barrels per stream-day (Oil and Gas Journal, April 1, 1974). In Canada there were 42 refineries in 1973 with a total capacity of 2.1 million b/sd, which is 71% of the capacity of those refineries in the U.S. which are not attacked in the CRP-2B attack. Refineries which produce asphalt only, are not considered here.

Of the remaining 88 refineries in the U.S. which are not struck by the CRP-2B attack, many would be inoperable because of the lack of electricity and/or the lack of feedstock such as crude oil, and those existing inside a region of fallout would also become inoperable, because none of them are constructed to operate under fallout conditions (Stephens, 1973). If the Soviets wished to destroy these remaining refineries, they would have to divert about 93 weapons to reduce the total surviving U.S. refining capacity to about 3% of the 1973 capacity, or 114 weapons for 2%, or 169 weapons for 1% remaining capacity, assuming that the system reliability of their weapons is 66.7%. Methods for analysis of surviving refining capability are given in Appendix C.

The petroleum industry requires a substantial inventory for operations and some inventory is unavailable, such as the product at the bottoms of tanks. Minimum operating inventories for 1973 for the U.S. are shown in Table 8.1, (National Petroleum Council, 1974). Illustrative operating conditions of the trans-Alaska pipeline are shown in Fig. 8.2. Under severe emergency conditions threatening survival, it may become necessary to tap into the unavailable supplies, that is, stocks in tank bottoms and in pipelines, although it may be inevitable that damage will be done to the part of the system which is tapped into.





大学のないないないないないのないないないないないないない

1. 0.4 N. 100 C

Table 8.1

Minimum Operating Inventories, 1973 (Millions of Barrels)

	Total Inventory Reported to Bureau of Mines 9/30/73 (1)	Unavailable Inventory Reported to NPC 9/30/73 (2)	Minimum Operating Inventory (3)	Additional Inventory to Meet Seasonal Needs (4) (Percent of (3))
Crude Oil <sup>*</sup>	241	155	240	1 T N
Products				
<u>Gasolines (Including</u> Avgas)	214	74	200	12%
Kerosine/Kero Jet District I-IV District V	37 42	11 13 13	30 35 35	26%
Naphtha Jet Fuel	Ŋ	1	S	₩
Distillate Fuel Oil	199	37	100	108%
Residual Fuel 0il	62	11	50	

\*Includes producers' lease stocks.

tSpecific seasonal build-up levels are not shown for maphtha-type jet fuels. These are produced by blending certain components in the gasoline boiling range with components in the kerosine boiling range and seasonal fluctuations can be covered by inventories of these other products.



ORNL-DWG 75-15925

156

「大学がない」である

and the state of the second

# 9. THE MEDICAL LOAD

Our primary concern in this project is for the relocated population which has, in accordance with the assumptions given to us in our work statement, been adequately protected against fallout radiation, and has been removed from the area affected with blast and fire. If the shelters are also properly equipped, as discussed previously in section 4.6 and if the guidelines for exposure to radiation are properly observed, there should be little requirement for medical care for the relocated population beyond the normal situation. Under these particular conditions, the principal additional potential hazards will be the increased exposure to infectious disease brought about by the conditions of continued close proximity to other people in shelters, and, if the shelter is located in a cave, there is a high probability of histoplasmosis if bats have been using the cave.

A rough estimate of the maximum number of people who could become affected by a communicable disease in shelter can be made. The number of reported cases of infectious communicable diseases in the U.S. in 1974 is listed in Table 9.1. The total number of cases in 1972 was around 450,000. If we assume this number of cases occurs during the year in which relocation takes place, and also assume, very roughly, that the average duration of infectious stage is two weeks, and that 90% of the infected population is relocated into shelters, then it is probable that about 16,000 infected people enter shelters. In the worst case, each infected person will enter a different shelter. The average shelter occupancy is expected to be around 100 people, and if half of these contract the infectious diseases because of shelter conditions, then the total number of ill people in shelters may reach a maximum of about 800,000. These numbers do not include influenza, which may increase the number of ill people by as much as 25%, depending on the season.

As of December 31st, 1971, there were 316,545 medical doctors in the U.S., one doctor for an average of about 650 people. If doctors are distributed approximately according to the population, then 60-70% of

157

a a a dhuar a dhuar a dhuar a a dhuar a

いいないななない

Table 9.1

# Reported Cases of Specified Notifiable Diseases United States, 1972-1974

Disease	1972	1973	1974
Amebiasis	2,199	2,235	2,743
Anthrax	2	2	2
Chickenpox	164,114	182,927	141,495
Diphtheria	152	228	272
Encephalitis	1,302	1,967	
Hepatitis	63,476	59,200	59,340
Leprosy	130	146	118
Measles (rubella)	32,275	26,690	22,094
Meningococcal infections	1,323	1,378	1,346
Mumps	74,215	69,612	59,128
Pertussis (whooping cough)	3,287	1,759	2,402
Poliomyelitis, Lotal	31	80	7
Rheumatic fever, acute	2,614	2,560	2,431
Ruhella (German measles)	25,507	27,804	11,917
Salmoncllosis, excluding typhoid fever	22,151	23,818	21,980
Shigellosis	20,207	22,642	22,600
Tuberculosis (newly reported cases)	32,882	30,998	30,210
Typhoid fever	398	680	437
Typhus fever	541	700	780
Total	446,806	455,354	379,302

<sup>a</sup>Reports of cases of acute rheumatic fever were received from 36 states (see p. 8).

<sup>b</sup>Provisional figure for 1974.

SOURCE: <u>MORBIDITY & MORTALITY</u>, <u>23</u> (53) for year ending December 28, 1974, Center for Disease Control, U.S. Department of Health, Education, and Welfare.

158

2.21 - 1.8 m 427

these, or about 200,000 are located in the high risk-areas. If threefourths of these doctors relocate with the population during the crisis period, and the others stay with their patients in the city hospitals, then there would be approximately 250,000 doctors surviving in the host areas after the attack. If the average shelter occupancy is 100 people, there will be about 1.8 million shelters, with one doctor for every seven shelters, for 700 people.

From these observations, we conclude that, if the shelters provide adequate protection against fallout, and if they are properly equipped, the initial medical load after an attack will not be severe in the host areas.

A number of caves have been stocked as fallout shelters, and a great many more caves could probably be adapted for use as shelters, as suggested in an article in PARADE magazine (June 15, 1975). One of the hazards of caves which could increase the medical load if caves were used extensively without taking precautions is the prevalence of the fungus, histoplasmosis, (Lewis, 1974) which develops in the droppings of bats or birds which have used the caves as shelters.

It would be useful for planning for a crisis relocation to have a survey of caves suitable for human occupancy, including the prevalence of histoplasmosis in the caves.

If the shelters in the host areas do not provide adequate protection from fallout radiation and are not properly equipped, there will obviously be a great increase in the medical load. Because of the lowering of the number of white blood cells due to exposure to harmful radiation, the resistance to infectious diseases is lowered, which, combined with possibly reduced morale and nonideal sanitary conditions in the shelters, could lead to rampant spreading of disease and more serious reactions to them in shelters which do not provide adequate fallout protection. Large quantities of antibiotics may be required to cope with this situ-Medical supplies should be moved to the host region during the ation. crisis period. According to a study by Staackmann, et al., (1970), as much as 80% of the current drug manufacturing capability could be destroyed by an attack on the urbanized areas, but the surviving 20% could probably expand their operation in the postattack period and adequately meet the requirements.

「ちたいいちょうい

# 10. GOVERNMENT AND THE ECONOMY

It is generally accepted among civil defense researchers (Allnut, 1971; W. M. Brown, 1971; Chenault, et al., 1967; Chenault and Nordlie, 1971; Dresch and Ellis, 1968; Goen and White, 1974) that a strong continuing government is essential for economic recovery after a nuclear attack. It is not as widely recognized that a strong federal government may be necessary for survival of a large number of people in the first few weeks after a nuclear attack, especially if these people have been removed from target areas by relocation during the crisis period. W. M. Brown (1971) has developed a scenario in which the federal government and most of the state governments are incapacitated due primarily to a failure to relocate during the crisis period preceding a nuclear attack, although a partial unplanned evacuation of the urban population occurs. Brown states that "the concept of rescue or assistance to neighboring communities fails because of the extreme threat to survival prospects felt nearly everywhere (due to fallout radiation, and shortages of food and fuel), and because of the lack of a national authority with the capability to effect the required actions" (insert added).

ત્રમ અને જે દિવિધાક સંબંધ અને અવધ્યતા કે જે ક્લાક વધાર છોતો કે જે આ વ્યવસાય છે. આ દેશી કે કોર્ટ્સ બે બે બે જે જ

Our research indicates that a large-scale shipping program of grain may be necessary in the first few weeks after a nuclear attack, in order to avoid severe food shortages for 60-80 million people. This operation will require coordinated multi-state federal planning and supervision, as indicated by the grain shipments listed in Table 7.2.

The major supply of grain in the postattack situation will be in the hands of farmers and owners of rural elevators. Surrender of grain by these people for federal promissory notes will require their confidence and trust in the federal government. It is unlikely that sufficient federal law enforcement or military personnel will be available to confiscate food in face of widespread opposition by local authorities.

Federal authority will depend on: (1) the existence of a functional national leadership with the appearance of self confidence, and (2) the existence of a credible recovery program. The existence of each of these elements must be conveyed by convincing communications to the public. The President, or other emergent leaders, can make an enormous,

almost indispensable, contribution to survival and recovery as well as national unity by frequent morale-building speeches broadcast on AM radio. The national resolve for recovery and unity could be increased by the existence of external threats, possibly from other than the Soviet Union. A strong feeling for revenge may arise, which may unite the nation toward a common goal even more intensely than the spirit which pervaded the nation during World War II ("Remember Pearl Harbor").

One of the plans for management of the economy in the postattack situation specifies that all resource allocations of major importance are to be directed by government agencies, and that inflationary pressures are to be controlled by price and resource-use regulations rather than by monetary authorities (Sobin, 1969). Sobin suggests that an inflationary gap may arise from the use of this system which could threaten the effectiveness of money as a means of exchange. He proposes that greater efficiency in resource management would result from a system that would divide the economy into two sectors: one would be controlled in the manner currently planned for the entire economy, and the other would be controlled by a system that would leave much larger scope for private iritiative and allow prices to rise toward an equilibrium level appropriate to the supply and demand conditions existing in the postattack situation. The closely controlled sector would include all production for government and population survival use; the free sector would include all other production, including production of survival-type items in excess of survival needs.

The economic recovery measures instituted would have to be responsive to the conditions that actually prevailed after the attack. It is beyond the scope of this study to more than indicate that the economic problems would be national and require a strong, competent national government enjoying the confidence of the people for their solution.

### BIBLIOGRAPHY

# (Partially Annotated)

こうごう おんだいきちょうなん シット・シング ちょう

ことでないできょうないが、「ないない」というないないできょうというないできょうという

ようででないないないない

- Advisory Committee on Civil Defense, <u>Response to DCPA Questions on</u> <u>Fallout</u>, DCPA Research Report No. 20, National Academy of Sciences, May 1973.
- Bruce C. Allnutt, <u>A Study of Consensus on Psychological Factors Related</u> to Recovery from Nuclear Attack, Human Sciences Research, Inc., May 1971, (AD 730 360).

One hundred questions were sent to 30 panelists to study agreements that existed (in 1970) on the probable social and psychological consequences of nuclear war and the impact of such factors on the processes of national recovery. The panel was in general agre that after a nuclear attack the incidence of positive, adaptive social behavior would be likely to outweigh the incidence of negative, maladaptive antisocial behavior. Simple fear emerged as the greatest psychological barrier to recovery, and communications appeared to be the key to penetrating that barrier, since the immense needs for leadership, information, and instruction would all be dependent on the ability of the government to communicate with its surviving constituents.

- Charles G. Anderson, Assessment of Post-Attack Health Resources, 5X-11101-2421E-02, System Sciences, Inc., July 1, 1970.
- S. I. Auerback, P. B. Dunaway, and R. C. Dahlman, <u>Postattack Ecology</u>, Oak Ridge National Laboratory, December 1973, (AD 771 764).
- Robert U. Ayres, <u>On Damage Assessment Models</u>, Hudson Institute, Inc., December 30, 1966, (AD 644 492).
- Robert U. Ayres, Methodology for Postattack Research, Hudson Institute, Inc., September 1, 1966, (AD 639 751).
- Robert U. Ayres, <u>Models of the Postattack Economy</u>, Hudson Institute. Inc., August 1, 1966, (AD 639 713).

- Robert U. Ayres, <u>Volume Two Environmental Effects of Nuclear Weapons</u>, Hudson Institute, Inc., December 1, 1965 (AD 632 279).
- Robert U. Ayres, <u>Volume III Environmental Effects of Nuclear Weapons</u>, Hudson Institute, Inc., December 1, 1965, (AD 632 281).
- Robert U. Ayres, Implications of Environmental Damage Due to Nuclear Attack on the U.S., HI-518-RR, Hudson Institute, June 30, 1965.
- Robert U. Ayres, <u>Special Aspects of Environment Resulting from Various</u> <u>Kinds of Nuclear Wars</u>, Hudson Institute, Inc., November 30, 1964, (AD 458 406).

- Robert U. Ayres, <u>Part II The Use of Scenarios of Evaluating Post-</u> <u>attack Disutilities</u>, Hudson Institute, Inc., January 8, 1964, (AD 430 057).
- Robert U. Ayres, <u>Part II Annex III Application of Input-Output</u> <u>Analysis to a Homeostatic Ecosystem</u>, Hudson Institute, Inc., January 8, 1964, (AD 303 111).
- Robert U. Ayres, <u>Part II Annex IV Effects of Thermonuclear War on</u> <u>Weather and Climate</u>, Hudson Institute, Inc., January 8, 1964, (AD 430 095).
- Robert U. Ayres, <u>Historical Examples of Ecological Disaster</u>, Hudson Institute, Inc., June 5, 1963, (AD 409 888).
- Robert U. Ayres, <u>Special Aspects of Environment Resulting from Various</u> <u>Kinds of Nuclear Wars</u>, Hudson Institute, Inc., June 5, 1963, (AD 409 421).
- Stephen Baer, <u>Preliminary Estimates of Labor Demands in the Postattack</u> Economy, IDA, October 1968, (AD 678 633).
- Paul R. Barnes, <u>The Effects of Electromagnetic Pulse (EMP)</u> <u>is the and</u> <u>Local Radio Communications</u>, ORNL-4873, Oak Ridge National Laboratory, March 11, 1974.
- David W. Bensen and Arnold H. Sparrow, (eds.), <u>Survival of Food Crops</u> and <u>Livestock in the Event of Nuclear War</u>, U.S. Atomic Energy Commission, Demember 1971. Available as CONF-700909 from National Technical Information Service, U.S. Department of Commerce, Springfield, Virginia 22151.

and a substation of the state o

<mark>debil (h. k</mark>afarasi, n. kanavatanina da nashashada kikutanama are tara ana anad hikubén newasikana ki

- John W. Billheimer, Frank J. Jones, and Myron Myers, <u>Food System Support</u> of the Relocation Strategy, Volumes I and II, SYSTAN, Inc., September 1975.
- John W. Billheimer and Lacy G. Thomas, <u>Postattack Food Availability and</u> <u>Accessibility--Detroit, Michigan</u>, Stanford Research Institute, November 1970.
- Robert H, Black and William H. Van Horn, <u>Development of Procedures</u> for Assessment of Local Industrial Productive Capacity Following <u>Nuclear Attack</u>, URS Research Company, February 1970, (AD 709 644). Proposes an early postattack Orientation Survey to rapidly generate information on the productive capacity of a community, and later a Planning Survey to provide a more accurate picture of the industrial capacity available as a function of the degrees of repair provided.
- H. A. Blair, "A Formulation of the Relation Between Radiation Dose and Shortening of Life Span," <u>1st Int. Conf. Peaceful Uses At</u>. Energy, Vol. II (1956).

- H. A. Blair, The Shortening of Life Span by Injected Radium, Polonium, and Plutonium, The University of Rochester Rep. UR-274 (1953).
- H. A. Blair, <u>A Formulation of the Injury, Life Span, Dose Relations</u> for Ionizing Radiation, II, Application to the Guinea Pig, Rat, and Dog, University of Rochester Rep. UR-207 (1952b).
- H. A. Blair, <u>A Formulation of the Injury, Life Span, Dose Relations</u> for Ionizing Radiation, I, Application to the Mouse, University of Rochester Rep. UR-206 (1952a).
- E. Block, Characterization of Soviet Civil Defense Plans for Fatality Calculations, Stanford Research Institute, Huntsville, Alabama, October 1974.

Describes Soviet Civil Defense and postattack survival plans, based on translations (ORNL) of Soviet mannuals on Civil Defense. জিল্পাক মহাজানীয় জিল্পা জিল্পা প্ৰথম জান মাজ কৰে প্ৰথম জান নাম নাম জান জান প্ৰথম জিল্পা জিল্পা জান জান বা বিভা

And the second states and the features and the second states and the second of the second second second second

- Stephen L. Brown, H. Lee, J. L. Mackin, and K. D. Moll, <u>Agricultural</u> <u>Vulnerability to Nuclear War</u>, Stanford Research Institute, February 1973, (AD 765 725).
- Stephen L. Brown, <u>Agricultural Vulnerability in the National Entity</u> Survival Context, Stanford Research Institute, December 1970.
- Stephen L. Brown and Pamela G. Kruzic, <u>Agricultural Vulnerability in</u> <u>the National Entity Survival Context</u>, Stanford Research Institute, July 1970, (AD 716 613).
- Stephen L. Brown and Ulrich F. Pilz, U.S. Agriculture: Potential Vulnerabilities, Stanford Research Institute, January 1969, (AD 695 687).
- Stephen L. Brown, Hong Lee, and Oliver S. Yu, Postattack Food Production and Food and Water Contamination, Stanford Research Institute, June 1968, (AD 678 187).
- William M. Brown, <u>Recovery from a Nuclear Attack</u>, October 1971, (AD 732 499).

ないのはないないないというというと

This study is based upon a particular scenario which involves an international crisis which over a period of several months escalates to a 4000 MT nuclear attack on the U.S. Because of the long crisis period, a partial evacuation of the urban population occurs. The attack is assumed to incapacitate the Federal government and most of the State governments, resulting in fragmentation of the nation into thousands of autonomous entities, each most concerned with its own problems. The major difficulties during the first few weeks occur in the parts of the country with the most intense radioactivity and/or the greatest shortage of food and fuel. The concept of rescue or assistance to neighboring communities fails because of the extreme threat to survival prospects felt nearly everywhere and because of the lack of a national authority with the capability to effect the required actions. William M. Brown, Emergency Mobilization for Postattack Reorganization, HI-874/2-RR Hudson Institute, May 15, 1968, (AD 669 623).

Elwyn M. Bull and Heldon E. Adams. Postattack Resource Management, American Technical Assistance Corporation, May 1975.

Reviews Runout Production Evaluation (ROPE) model for simulating regulated resource management in the first ninety days following a nuclear attack, but does not provide results of any calculations with the model. The model does not deal with transportation constraints such as fallout barriers, destroyed bridges, petroleum shortages, hence the usefulness of the model as it is described in this document is doubtful.

- Elwyn M. Bull, <u>Postattack Capabilities of the Economy in the First</u> <u>Ninety Days (U) Final Report</u>, American Technical Assistance Corporation, June 1973, (AD 526 403).
- Elwyn M. Bull, <u>The Runout Production Evaluation (ROPE) Model: Structure</u> and <u>Methodology</u>, <u>General Research Corporation</u>, <u>June 1973</u>, (AD 763 810).
- Elwyn M. Bull, <u>Antibiotics Production Capacities in the Postattack</u> Economy, Research Analysis Corporation, February 1972 (AD 730 303).
- Elwyn M. Bull and Bernard Sobin, <u>Measurement of Critical Production</u> <u>Capacities for Models of the Postattack Economy</u>, Research Analysis Corporation, Technical Paper RAC-TP-387, February 1970, (AD 701 914).
- Z. G. Burson, "Environmental and Fallout Gamma Radiation Protection Factors Provided by Civilian Vehicle.," <u>Health Physics</u>, <u>26</u> 41-44, 1974.
- William W. Chenault, <u>Postattack Conditions and Recovery Management</u> Requirements, Human Sciences Research, Inc., February 1971.
- William W. Chenault, Richard E. Engler, and Peter G. Nordlie, <u>Social</u> and <u>Behavioral Factors in the Implementation of Local Survival</u> and <u>Recovery Activities</u>, Human Sciences Research, Inc., August 1967, (AD 663 811).
- William W. Chenault and Peter G. Nordlie, <u>Consumer Behavior and Worker</u> <u>Participation in Recovery Activities</u>, Human Sciences Research, Inc., February 1967, (AD 651 098).

Examines the problems of getting survivors to participate in a organized recovery effort.

C. V. Chester, G. A. Cristy, and C. M. Haaland, <u>Strategic Considerations</u> <u>in Planning a Counterevacuation</u>, ORNL-4888, Oak Ridge National Laboratory, December 1975.

- R. O. Chester, <u>Dose and Deposition from a Nuclear Reactor Core</u> Meltdown, <u>ORNL-4944</u> (1974).
- Donald E. Clark, Jr., <u>Techniques for Development of Postattack Recovery</u> <u>Management</u>, Stanford Research Institute, November 1969, (AD 701 099).

Donald E. Clark, Jr. and Carl F. Miller, <u>Postattack Recovery Management</u>: <u>Concepts and Techniques for Model Development</u>, SRI, March 1967, (AD 658 744).

Donald E. Clark, Jr., Carl F. Miller, and George D. Hopkins, <u>An</u> <u>Approach to Defining Postattack Recovery Management Concepts and</u> <u>Techniques</u>, Stanford Research Institute, November 1966, (AD 646 627).

Consumer and Food Economics Research Division, Agricultural Research Service, U.S. Department of Agriculture, Food Consumption of Households in the Northeast, Seasons and Year 1965-66, Washington, D.C., August 1972.

- G. A. Cristy, <u>Best Shelter for Critical Industry Workers</u>, ORNL-5022, Oak Ridge National Laboratory, August 1975.
- E. P. Cronkite and V. P. Bond, "Diagnosis of Radiation Injury and Analysis of the Human Lethal Dose of Radiation," <u>U.S. Armed</u> Forces Med. J., <u>11</u> 249-260 (1960).

- H. L. Crutcher, <u>Upper Wind Statistical Charts of the Northern Hemi-sphere</u>, NAV AER 50-1C-535, Vols. I, II, & III, Office of Chief of Naval Operations, August 1959.
- Harold O. Davidson, <u>Biological Effects of Whole-Body Gamma Radiation</u> on <u>Human Beings</u>, Operations Research Office, The Johns Hopkins University, The Johns Hopkins Press (1957).
- H. L. Dixon, D. G. Hanes, and P. S. Jones, <u>A Systems Analysis of the</u> <u>Effects of Nuclear Attack on Railroad Transportation in the Conti-</u> <u>nental United St.</u> Stanford Research Institute, Menlo Park, CA, 1960.
- Francis W. Dresch and H. 1 Ellis, <u>Criteria for Early Postattack Economic</u> <u>Viability of Local Areas</u>, Stanford Research Institute, June 1974, (AD 002 746).

Formulas are developed for assembling and analyzing lists of variables and factors that could affect viability or degrade potential output of SMSAs after a nuclear attack. Such formulas should assist in assessing the economic viability of an SMSA and in making decisions on abandoning a site temporarily. a su de contrates en la sus su compressiones de la subsection de contrates de la subsection de la subsection de

de la Monthelle Marthalistic de la constant la constant de la Subbrehanda.

- Francis W. Dresch, <u>Requirements for Comparative Evaluation of Counter-</u> <u>measures to Possible Postattack Fiscal Problems</u>, Stanford Research Institute, April 1969, (AD 695 641).
- Francis W. Dresch, Information Needs for Postattack Recovery Management, Stanford Research Institute, April 1968, (AD 668 692).
- Dunlap and Associates, Inc., <u>Training Requirements for Postattack</u> Adaptive Behavior - Final Report, December 1965, (AD 624 870).
- P. T. Egorov (Yegorov), I. A. Shlyakhov, and N. I. Alabin, <u>Civil</u> Defense, Moscow 1970, translation: ORNL-TR-2793, December 1973.
- Kay Franz, Factors to Consider in Maintaining Nutritional Adequacy in <u>a Long-Term Survival Situation</u>, ORNL-5068, Oak Ridge National Laboratory, June 1975.
- Clark David Garland, <u>Economic Alternatives and Policy Implications</u> of a Strategic Commodity Reserve for National Security Considerations, ORNL-TM-3741, Oak Ridge National Laboratory, March 1972.
- Raymond D. Gastil, <u>Scenario for Postattack Social Reorganization</u>, Hudson Institute, Inc., August 20, 1969, (AD 854 630).

A scenario was developed for evaluating social reorganization following a hypothetical nuclear attack of 2000 MT in 1970-1972, primarily against strategic forces, with a fatality level of about 15%. The following twelve important points emerged from the study: (1) general social breakdown is unlikely; (2) present policy (1969) to discourage evacuation can lead to serious problems in some scenarios (referring to general destruction of bureaucracy due to not evacuating); (3) interactions of nuclear effects will add to the disaster; (4) large-scale geographical isolation may occur; (5) trans-attack evacuation is possible and may be likely; (6) relative inactivity in shelter during disaster and aftermath poses special morale problem; (7) variations in military outcome are critical; (8) immediately after attack, civil defense will receive a top priority; (9) states are apt to have a primary role in the immediate postattack period; (10) relatively well-off states will have to be pressured to aid the more destroyed; (11) there will be cases of extreme food deficiency; and, (12) inflation need not be serious.

William G. Gay and William W. Chenault, <u>Crisis Relocation-Distributing</u> <u>Relocated Populations and Maintaining Organizational Viability</u>, Human Sciences Research, Inc., April 1974, (AD 730 114).

Relocation can exacerbate the problem of reorganization and reintegration following a nuclear attack because it may break down organized work groups and disrupt their routine patterns of interaction. "An efficient approach to relocation would appear to include the construction of 'hasty' shelters in the suburban fringe, where

Start Start Street Start and a second of a

hosting capacity is great (except in basement spaces) and where postattack workers could be housed relatively close to critical plants and families . . . Retail trade per capita appears to be a much sounder basis for deriving the relative hosting capacities of different communities than the oft-cited availability of basement spaces. The cost of providing improvised shelter in a community otherwise capable of expanding services will probably be far less than the costs of upgrading the organizational capacity of a community which happens to have many shelter spaces . . . The continuity of economic organizations is a salient factor in evacuation followed by attack, with the requirement for reorganizing the postattack economy to maximize production and distribution of essential goods and services."

- Richard L. Goen and William L. White, <u>Postattack Contingency Plans</u> for Undamaged Areas, Stanford Research Institute, July 1974.
- Richard L. Goen, <u>The Magnitude of Initial Postattack Recovery Activ</u>ities, Stanford Research Institute, December 1971, (AD 741 389).
- Richard L. Goen, Donald E. Clark, C. Alexander Kamradt, John W. Ryan, and Richard B. Bothun, <u>Critical Factors Affecting National</u> Survival, SRI Project, March 1969, (AD 693 877).
- Jeffrey K. Hadden and Edgar F. Borgatta, <u>Appendix I, A Study of the</u> <u>Demography of Nuclear War</u>, HSR-RR-66/14-Pr-App I, Human Sciences Research, Inc., May 1966.
- R. W. Hall and J. W. Billheimer, <u>Local Utilization of National Food</u> Resources, Stanford Research Institute, November 1973
- William A. Hamberg, <u>Vulnerability of a Zonal Transportation System</u>, Stanford Research Institute, August 1971.
- William A. Hamberg, <u>Transportation Vulnerability Research: Review</u> and Appraisal 1959-1969, Stanford Research Institute, January 1969.
- Ernest C. Harvey and Robert W. Hubenette, <u>Alternative Hosting and</u> <u>Protective Measures</u>, Stanford Research Institute, December 1968 (Confidential, declassified December 1974).
- Frederick S. Hillier and Gerald J. Lieberman, <u>Introduction to Operations</u> Research, Holden-Day, Inc., 1974.

de de la comparte de participada de la comparte de

the design of the

Eric Hirst, Energy Consumption for Transportation in the U.S., ORNL-NSF-EP-15, Oak Ridge National Laboratory, March 1972.

- Jean M. Ingersoll, <u>Historical Examples of Ecological Disaster Famine</u> <u>in Russia 1921-22 Famine in Bechuanaland 1965</u>, Hudson Institute, Inc., December 1965, (AD 629 887).
- Jean M. Ingersoll, <u>Special Aspects of Environment Resulting from Various</u> <u>Kinds of Nuclear Wars Part III Appendix 1-2 Historical Examples of</u> <u>Ecological Disaster (III)</u>, Hudson Institute, Inc., September 1, <u>1964</u>, (AD 450 798).
- Jean M. Ingersoll, Part II, Appendix 1-2- Historical Examples of Ecological Disaster (II), January 8, 1964, (AD 433 732).
- S. Jablon, S. Fujita, K. Fukushima, T. Ishimaru, and J. A. Auxier, "RBE of Neutrons in Japanese Survivors," in <u>Symp. Neutrons</u> <u>Radiobiol.</u>, November 1969, UT-AEC Agr. Res. Lab. and Oak Ridge National Laboratory, pp 547-579, USAEC Rep. Conf. 691106 (1969).
- John Karlik, Aspects of Postattack Economic Recovery, Hudson Institute, Inc., May 15, 1968.
- June H. Karlson and Ellen K. Langer, <u>Postattack Research Volume VII-</u> <u>Reviews and Abstracts of Research on Socio-Psychological Problems</u>, <u>MITRE Corporation</u>, August 1969, (AD 768 570).

Reviews seven studies of the social and psychological effects of nuclear war. Disaster research indicates that totally irrational or panic behavior is less likely than emotional shock or depression in a postattack situation. Communications, coordination and control, authority, and transportation are major factors in organizing an adequate emergency social system.

June H. Karlson, Ellen K. Langer, and Frederick J. Wells, <u>Postattack</u> <u>Research, Volume VI, Reviews and Abstracts of Research on Critical</u> <u>Postattack Resources and Industries</u>, The MITRE Corporation, <u>August 1969</u>.

Reviews and assesses research efforts up to 1969 on the survivability of critical industries, namely, food and agriculture, transportation, communications, electric power, petroleum products, natural gas, and iron and steel. Predictions seem to be unreasonably optimistic, especially for the 26,000 MT attack. ar a substitution of the states of the states of the states of the second states of the states of the

Nin Schwie

- June H. Karlson, Ellen Langer, and Frederick J. Wells, <u>Postattack</u> <u>Research- Volume V - Reviews and Abstracts of Research on Surviving</u> <u>Economic Production Potential</u>, The MITRE Corporation, February 1969, (AD 700 132).
- June H. Karlson and Frederick J. Wells, <u>Reviews and Abstracts of Research</u> on <u>Economic Recovery Management</u>, The MITRE Corporation, February 1969, (AD 700 131).

- June H. Karlson and Frederick J. Wells, <u>Reviews and Abstracts of Research</u> on the Demographic Effects of Nuclear War, The MITRE Corporation, February 1969, (AD 702 22...).
- June H. Karlson and Frederick J. Wells, <u>Reviews and Abstracts of Research</u> on Postattack Medic ' and Health Problems, Volume II, The MITRE Corporation, February 1969 (AD 684 341).

Reviews and assesses several documents on postattack medical and health problems. Some of the studies have considered the increased susceptibility to disease due to exposure to radiation, but only qualitatively. Inherent and perhaps unresolvable uncertainties which surround the possible threats of epidemics lead to the conclusion that further work in this area should be oriented toward preventative countermeasures rather than estimating the probabilities of such epidemics.

June H. Karlson, <u>A Context Study of Postattack Research</u>, Volume IV, The MITRE Corporation, February 1968, (AD 678 499).

Lists 222 documents relevant to postattack problems.

- C. H. Kearny, "Expedient Life-Support Equipment for Shelters," p 7 in <u>Health Physics Division Annual Progress Report (Period Ending</u> <u>June 30, 1975</u>, ORNL-5046, Oak Ridge National Laboratory, September 1975.
- Grace J. Kelleher, <u>Allocating Contested Space in a Regional Movement-</u> <u>to-Shelter System: A Case Study of the Central Gulf Coast Region</u>, IDA Research Paper P-310, Institute for Defense Analyses, January 1967.
- Robert B. King, Jr. and Anna M. Kleiner, <u>Social Institutions and Thermo-</u> <u>nuclear War -- A Case Study of Higher Education</u>, HBR-SINGER, Inc., January 1967, (AD 813 244).

のためになったのにないため

- Robert A. Krupka, <u>A Limited Postattack History of CDEX '68</u>, Hudson Institute, Inc., August 20, 1969.
- George W. Ladd and Dennis R. Lifferth, "An Analysis of Alternative Grain Distribution Systems," <u>American Journal of Agricultural</u> <u>Economics</u>, <u>57</u> (3) August 1975.
- Brian K. Lambert and Joseph E. Minor, <u>Vulnerability of Regional</u> and Local Electric Power Systems: <u>Nuclear Weapons Effects</u> and Civil Defense Actions, Defense Electric Power Administration, Department of the Interior, Washington, D.C., July 1975.
- W. H. Langham, (ed.), <u>Radiological Factors in Manned Space Flight</u>, Report of the Space Radiation Study Panel of the Life Sciences Committee, Space Science Board, National Academy of Sciences, National Research Council, Washington, D.C. (1967) (NRC Publ. 1487).

- Richard K. Laurino and Francis W. Dresch, <u>National Entity Survival</u>: <u>Measure and Countermeasure</u>, Stanford Research Institute, June 1971, (AD 730 946).
- Warren C. Lewis, M.D., "Histoplasmosis in Caves," <u>NSS News</u>, <u>32</u> (24) (National Speleological Society) (1974).
- Don C. Lindsten and Maurice Pressman, <u>Field Expedients for Decontami-</u> nating Water Containing Nuclear Bomb Debris, U.S. Army Engineer Research and Development Laboratories, Fort Belvoir, Virginia, Report-1904, USAMERDC, July 1967.
- C. C. Lushbaugh, "Human Radiation Tolerance," in <u>Space Radiation</u> <u>Biology and Related Topics</u>, (C.A. Tobias and P. Todd, ed.), pp 475-522, Academic Press 1974.
- C. C. Lushbaugh and J. A. Auxier, "Reestimation of Human LD<sub>50</sub> Radiation Levels at Hiroshima and Nagasaki," <u>Radiation Research</u>, <u>39</u> (526) (1969) (Abstract).
- C. C. Lushbaugh, F. Comas, E. L. Saenger, M. Jacobs, R. Hofstra, and G. A. Andrews, "Radiosensitivity of Man by Extrapolation from Studies of Total-Body Irradiation of Patients," <u>Radiation Research</u>, <u>27</u> (487-488) (1966) (Abstract).
- The Lynchburg College Research Conter, Federal Organization and Responsibilities for Emergency Preparedness and Resource Management Sections I, II, IV, November 30, 1968.

Catalogues responsibilities assigned (1968) to federal agencies by official documents, and recites official instructions which control the manner in which the reponsibilities are to be carried out. Oriented toward the postattack phase of the Five-City study, and therefore not very useful for CRP postattack planning.

- R. W. Manweiler, Effects of Nuclear Electromagnetic Pulse (EMP) on Synchronous Stability of the Electric Power System, ORNL-4919 Oak Ridge National Laboratory, November 1975.
- James H Marable, Paul R. Bornes, and David B. Nelson, <u>Power System</u> <u>EMP Protection</u>, ORNL-4958, Oak Ridge National Laboratory, May 1975.
- G. Mathé, J. L. Amiel, and L. Schwarzenberg, "Treatment of Acute Total-Body Irradiation Jnjury in Man," <u>Ann. N.Y. Acad. Sci.</u>, <u>114</u> (368-392) (1964).

Allah Barranah minarakan metanikatan keraja

Fred R. McFadden and Charles D. Bigelow, <u>Development of Rapid Shutdown</u> <u>Techniques for Critical Industries</u>, Stanford Research Institute, January 1966. and the second state of the second state of the second state of the second second second second second second s
- John W. McLanahan, Ph.D. and Rober<sup>+</sup> S. Hostetter, <u>Displacement: Social</u> and Psychological Problems, HBR-SINGER, August 1965, (AD 624 519).
- Carl F. Miller and Richard K. Laurino, <u>A Concept for Postattack Nuclear</u> Emergency Operations, The Dikewood Corporation, August 1973, (AD 774 162).
- Carl F. Miller, <u>Constraints on Civil Defense Operations in Physically</u> Damaged Areas, Center for Planning and Research, February 1973.
- K. Z. Morgan and J. E. Turner, <u>Principles of Radiation Protection</u>, Robert E. Krieger Publishing Co., 1973.
- National Academy of Sciences, Long-Term Worldwide Effects of Multiple Nuclear-Weapons Detonations National Research Council, 1975.
- National Council on Radiation Protection and Measurements, <u>Radiological</u> <u>Factors Affecting Decision-Making in a Nuclear Attack</u>, NCRP Report No. 42, November 15, 1974.
- National Petroleum Council, <u>Petroleum Storage Capacity</u>, September 10, 1974.
- National Petroleum Courcil, <u>Civil Defense and Emergency Planning for</u> the Petroleum and Gas Industries, March 19, 1964.
- Jiri Nehnevajsa, <u>Beravioral and Organizational Issues Connected with</u> <u>Crisis Relocation Flanning</u>, Stanford Research Institute, May 6, 1975.
- D. B. Nelson, Effects of Nuclear EMP on AM Radio Broadcast Stations in the Emergency Broadcast System, ORNL-TM-2830, Oak Ridge National Laboratory, July 1971.
- Peter G. Nordlie, <u>The Feasibility of Developing Standard Descriptions</u> of Postattack Situations, Human Sciences Research Inc., June 1967, (AD 655 649).
- John DeWitt Norton, <u>Studies for an Economic Model for Survival and</u> <u>Recovery of a Single City - Volume III</u>, National Planning Association, April 1969, (AD 689 486).
- John DeWitt Norton, <u>Studies for Economic Model for Survival and</u> <u>Recovery of a Single City - Volume II</u>, National Planning Association, November 1, 1968, (AD 689 486).
- John DeWitt Norton, <u>Studies for an Economic Model for Survival and</u> <u>Recovery of a Single City - Volume I</u>, National Planning Association, June 1, 1968, (AD 682 074).
- William W. Pendleton, Jeffery K. Hadden, and Edgar F. Borgatta, <u>A</u> Second Study of the Debography of Nuclear War, Human Sciences Research, Inc., August 1967, (A. 662 076).

Philipping and a second second second second

William W. Pendleton, <u>A Study of Personnel Demands and Availabilities</u> for Postattack Countermeasure Systems, Human Sciences Research, Inc., June 1966, (AD 537 833).

"Thirteen areas of need that would characterize the postattack period were examined: (1) distribution of food and water, (2) medical care, (3) housing and building, (4) provisions for trade, (5) transportation, (6) communication, (7) command and control, (8) restoring public utilities, (9) relocation of the population, (10) decontamination, (11) welfare services, (12) defense, and (13) maintaining morale."

- William W. Pendleton, <u>A Study of the Demography of Nuclear War</u>, Human Sciences Research, Inc., May 1966, (AD 647 802).
- M. Polan, <u>An Analysis of the Fallout Prediction Models Presented at</u> the USNRDL-DASA Fallout Symposium of September 1962, Volume 1: <u>Analysis, Comparison, and Classification of Models</u>, Ford Instrument Company Division of Sperry Rand Corporation, under contract to U.S. Naval Radiological Defense Laboratory, USNRDL-TRC-68, September 8, 1966.
- D. Ramsden, H. F. Passant, C. O. Peabody, and R. G. Speight, "Radioiodine Uptakes in the Thyroid. Studies of the Blocking and Subsequent Recovery of the Gland Following the Administration of Stable Iodine," <u>Health Physics</u> 13 (633) (1967).
- Frederick C. Rockett and William M. Brown, <u>Crisis Preparations for</u> <u>Postattack Economic Recovery</u>, Hudson Institute, Inc., July 15, 1966, (AD 639 357).
- Howard R. Ross, U.S. Passenger Transportation: An Inventory of Resources and an Analysis of Capabilities of Surface Modes, Stanford Research Institute, March 1967.
- John W. Ryan, Peggy A. Garza, and Stephen L. Brown, <u>A Damage Assessment</u> <u>Model for Agricultural Crops</u>, Stanford Research Institute, September 1974, (AD 002 291).
- G. A. Sacher and D. Grahn, "Survival of Mice Under Duration-of-Life Exposure to Gamma Rays. I. The Dosage-Survival Relation and the Lethality Function," J. Nat. Cancer Inst., 32: 277-321, 1964.
- G. A. Sacher, "Reparable and Irreparable Injury: A Survey of the Position in Experiment and Theory," in <u>Radiation Biology and</u> <u>Medicine</u>, (W.D. Claus, ed.), pp 283-313, Addison-Wesley, 1958.
- Abner Sachs, <u>Nuclear Emergency Operations Planning for Evacuation</u> of Urbanized Areas (Bravo NEOP), Volume I, IDA Paper P-959, Institute for Defense Analyses, October 1973.

2. (4. 2) 23. (4. 2) 44. (4. 1) 1) 1) 1, 10 1, 10 1, 10 1, 10 1, 10 1, 10 1, 10 1, 10 1, 10 1, 10 1, 10 1, 10 1

- Leo A. Schmidt, <u>The Use of the ADAGIO Computer Program in Strategic</u> <u>Evacuation Analysis</u>, IDA Paper P-1067, Institute for Defense Analyses, October 1974.
- Leo A. Schwidt, <u>Private Communication on Modified WSEG 10</u>, (NAS Fallout Model), Institute for Defense Analyses, December 1974.
- Monroe B. Snyder and Alfred J. Farina, <u>Methods and Techniques for</u> <u>Postattack Manpower Utilization</u>, Human Sciences Research, Inc., <u>August 1967</u>, (AD 661 336).

"The conclusion is reached that in general, the postattack manpower unit will be operating in an environment so different from the present that the present organization of functions and approaches to manpower utilization will probably be imappropriate and inadequate to handle the postattack problems."

Bernard Sobin, <u>Models of Economic Capability after Nuclear Attack</u>, Research Analysis Corporation, April 1969, (AD 688 182).

and a standard of the standard

- Bernard Sobin, <u>A Model of Technological Capacity to Support Survivors</u> of Nuclear Attack, Research Analysis Corporation, Technical Paper RAC-TP-313, September 1968, (AD 676 115).
- Bernard Sobin and David F Gates, <u>Economic Implications of High</u> <u>Population</u>, Research Analysis Corporation, July 1968, (AD 391 898).
- Space Radiation Study Panel, <u>Radiobiological Factors in Manned Space</u> <u>Flight</u>, Publication 1487, National Academy of Sciences, National Research Council, 1967.
- A. H. Sparrow, Susan S. Schwemmer, and P. J. Bottino, "The Effects of External Gamma Radiation from Radioactive Fallout on Plants with Special Reference to Crop Production," <u>Radiation Botany</u>, <u>11</u> pp 85-116, 1971.
- Milton Staackmann, William H. Van Horn, and Carl R. Foget, <u>Damage to</u> the Drug Industry from Nuclear Attack and Resulting Requirements for Repair and Reclamation, URS Research Company, July 1970.

"The drug industry has the potential to meet most demands for lifesaving drugs in the early postattack period by judicious conservation of available supplies and to meet the longer term demands by the expenditure of a modest repair and/or conversion effort." About 20% of the drug industry might remain unscathed following an intensive nuclear attack concentrating on SMSAs. Maynard M. Stephens, <u>Vulnerability of Total Petroleum Systems</u>, The Office of Oil and Gas, The Department of the Interior, May 1973.

Maynard M. Stephens, <u>Minimizing</u> Damage to Refineries, The Office of Oil and Gas, The Department of the Interior, February 1970. 

- Palmer Steward, "Mathematical Models of Mammalian Radiation Response for Space Applications," in <u>Space Radiation Biology and Related</u> <u>Topics</u>, (C. A. Tobias and P. Todd, ed.), pp 523-582, Academic Press, 1974.
- J. B. Storer, "Rate of Recovery from Radiation Damage and Its Possible Relationship to Life Shortening in Mice," <u>Radiat. Res.</u>, <u>10</u>: 180-196, 1959.
- Subcommittee on Energy of the Committee on Science and Astronautics, <u>Energy Facts</u>, U.S. House of Representatives, Ninety-Third Congress, First Session, by the Science Policy Research Service, Library of Congress, Serial H, U.S. Government Printing Office, 1973.
- Sanford B. Thayer and Willis W. Shaner, <u>The Effects of Nuclear Attacks</u> on the Petroleum Industry, Stanford Research Institute, July 1960.
- M. N. Titov, P. T. Yegorov (Egorov), B. A. Gayko, and others, <u>Civil</u> Defense, Moscow 1974, translation: ORNL-TR-2845, July 1975.

This manual of Civil Defense reviews and updates the more comprehensive Civil Defense manual of 1970.

- Cornelius A. Tobias and Paul Todd, Editors, <u>Space Radiation Biology and</u> Related Topics, Academic Press, 1974.
- Transportation Association of America, <u>Transportation Facts and</u> Trends, Eleventh Edition, Washington, D.C., December 1974.
- D. Turrentine, <u>A Context Study of Postattack Research Volume I</u> -Summary, The MITRE Corporation, February 1968, (AD 678 496).

The first three volumes on this four-volume set discuss a general model for systems analysis of postattack problems, but specific parameters and coefficients are not given.

D. Turrentine, <u>A Context Study of Postattack Research Volume II</u> -Approach, The MITRE Corporation, February 1968, (AD 678 497).

The first three volumes of this four-volume set discuss a general model for systems analysis of postattack problems, but specific parameters and coefficients are not given.

D. Turrentine, <u>A Context Study of Postattack Research Volume III</u> -<u>Model Development</u>, The MITRE Corporation, February 1968, (AD 678 498).

feldbenkähtereten der siestendetter

The first three volumes of this four-volume set discuss a general model for systems analysis of postattack problems, but specific parameters and coefficients are not given.

- U.S. Bureau of the Census, Census of Transportation, 1972, <u>Truck</u> <u>Inventory and Use Survey: U.S. Summary</u>, TC72-T52, U.S. Government Printing Office, Washington, D.C., 1973.
- U.S. Office of Emergency Planning, <u>Money, Credit and Banking in a</u> <u>Post-Attack Emergency</u>, Board of Governors of the Federal Reserve System, June 1965.
- William H. Van Horn and Kenneth Kaplan, <u>The Development of Postshelter</u> <u>Emergency Operations Planning at the Municipal Level: Phase II</u>, URS Research Company, April 1974.

A master check list is developed for use in recovery operations in local jurisdictions with damage from a nuclear attack.

William H. Van Horn, <u>Postattack Recovery and Operation Parameters</u> <u>Affecting Debris Estimation Procedures</u>, URS Research Company, June 1971.

Present debris prediction methods appear adequate for designing a debris removal/control system, but may not provide enough detail for some situations. An "ultimate" prediction method would (1) Describe the actual depth of debris across streets ("microcontours"); (2) Indicate the general composition of the debris, i.e., the fraction wood, steel, masonry, etc.; (3) Indicate the fire potential of the debris; and (4) Give the configuration of the debris in terms of interstices and voids for the purpose of estimating shielding effects from radiation from fallout.

F. E. Walker, Estimating Production and Repair Effort in Blast-Damaged Petroleum Refineries, Stanford Research Institute, July 1969.

- S. Warren and J. Z. Bowers, "The Acute Radiation Syndrome in Man," <u>Ann. Intern. Med.</u>, <u>32</u> 207-216 (1950).
- M. D. Wright, E. L. Hill, J. S. McKnight, and S. B. York, III, <u>Mine</u> Lighting and Ventilation in Crises, Final Report 43U-982-1, Research Triangle Institute, October 1975.

APPENDIX A TRANSPORT BY COMMERCIAL AIRLINES

FRECEDING PAGE BLANK NOT FILMED

all the second state and supply a supply and and an and a supply supply and and a supply supply and a supply s

かいたいのかいとうという

### APPENDIX A. TRANSPORT BY COMMERCIAL AIRLINES

### A.1 Introduction

It is assumed here that most land routes are closed and seaports are destroyed. A massive airlift would then be the alternative for bringing in supplies and evacuating people from regions heavily contaminated with fallout. As shown in recent history, air evacuation and supply is possible.

The Berlin Airlift and the partial evacuation of Danang, Vietnam, have demonstrated the feasibility. A Boeing 747, with FAA-approved seating capacity of 498, was altered to have a capacity of over 1000, although it was not used in the evacuation. In similar conditions, other aircraft seating capacities could be extended.

Tables A.1 and A.2 list the aircraft available for evacuation purposes in the Reserve Air Fleet (CRAF) and War Air Services Program (WASP) fleet. These tables list the seat-mile and cargo ton-mile capacities of the fleet, using averaged data pertaining to normal operating procedures. These are listed to give an idea of the order of magnitude that is of interest. Table A.3 lists aircraft from each major grouping and contains information pertaining to them. With this information and the use of Figs. A.1 and A.2, an idea can be obtained of the length of runway required to launch a specified aircraft. This information has been listed in Table A.4 for the aircraft mentioned, also listed are flight times and fuel consumption for 500- and 1000-mile flights.

Of prime interest is that in an actual emergency situation, i.e., a crisis situation preceding a potential nuclear attack, no plan exists for the relocation of our commercial air fleet from target areas. A detailed plan should be developed to meet this situation.

### A.2 Existing Programs

The CRAF and WASP were created in case of a national emergency by Executive Order 11490. They are comprised of all transport aircraft

Appendix A was written by Ronald R. Davis.

Table A.1

NAMES AND AND ADDRESS

λ¥,

## Annual Passenger Seat-Mile and Additional Cargo Ton-Mile Capacities of WASP Passenger Aircraft, Calendar Year 1975 International Fleet

Aircraft	Number in Certificated Carrier Fleet 12/31/74	International CRAF Fleet 12/31/74	International WASP Fleet 12/31/74	Annual Psgı. Seat-Mile Capacity Per Individual Aircraft (in thousands)	Anrual Additional Cargo Ton-Mile Capacity Per Individual Aircraft (in thousands)
TURBOFAN-4 ENGINE		ł			
707-300B	109	/	707	202,943	9,990
B-720B	7	I	7	216,741	5,464
B-747	104	69	35	657,679	39,758
DC-8-50	13	1	13	241,448	7,629
-61	34	I	34	340,530	11,119
-62	14	I	14	283,138	8,978
TURBOFAN-3 ENGINE					
B-727-100	68	ı	68	163,162	3,802
-200	130	ı	130	228,235	6,145
DC-10-10	22	ı	22	409,771	19,868
-30/40	24	14	10	430,875	32,587
L-1011	47	I	47	457,987	18,357
TURBOJET-4 ENGINE					
<b>B-707-300</b>	10	I	10	258,457	8,497
B-720	'n	1	e	220,752	8,059
DC-8-10/20	m	ı	e	244,035	6,847
-30	4	I	4	231,425	7,243

:

-
1. 2
Ρ
-
-
ີ
- <b>-</b>
0
- 65
$\sim$
• •
•
_ <
_
പ
_
0
- 12
- 5-4
•

A DELLA DELA PROVINCIA DE

					Annual
Aircraft	Number in Certificated Carrier Fleet 12/31/74	International CRAF Fleet 12/31/74	International WASP Fleet 12/31/74	Annual Psgr. Seat-Mile Capacity Per Individual Aircraft (in thousands)	Additional Cargo Ton-Mile Capacity Per Individual Aircraft (in thousands)
TURBOPROP-4 ENGINE L-188 A/C/F TOTALS	<u>1</u> 593	106	<u>503</u>	006,06	723

J. F. Laufer, <u>Civil Aeronautics Board War Air Service Program (WASP) Resource Report</u>, Bureau of Economics, Civil Aeronautics Board, 1975. SOURCE:

183

のないのないないであるというである

人はないないないないないがいたい

stadiski (prodintekatisti strikty system (phys.) (sta obrazoda i sa sa sistem o ya di sa fina shiki ki kasha s

יייי איייטעראיניגעיינעראינעראינעראינעראיינעראיייטערערערייי

:

Table A.2

# Annual Passenger Seat-Mile and Additional Cargo Ton-Mile Capacities of WASP Passenger Aircraft, Calendar Year 1975 Domestic Fleet

Aircraft	Number in Certificated Carrier Fieet 12/31/74	Domestic CRAF Fleet 12/31/74	Domestic WASP Fleet 12/31/74	Annual Psgr. Seat Mile Capacity Per Individual Aircraft (in thousands)	Annual Add'1. Cargo Ton-Mile Capacity Per Individual Aircraft (in thousands)	Total Annual Psgr. Seat- Mile Capacity of WASP Fleet (in thousands)	Total Annual Additional Ton-Mile Capacity of WASP Fleet (in thousands)
TURBOFAN-4 ENGINE	0	1	06	204.484	9,633	18,403,560	866,970
B-/U/-TUUD	2 4	I	16	200,268	5,870	3,204,288	93,920
B-/20B DC-8-61	14	ł	14	295,891	12,203	4,142,474	170,842
TURBOFAN-3 ENGINE		I	666	149,971	4.218	33,293,562	936,396
B-727-100	272		205	192,994	6,330	39,563,770	1,297,650
	07	1	50	403,077	21,196	20,153,850	1,059,800
L-1011	23	1	23	394,616	20,086	9,076,168	461,978
TURBOFAN-2 ENGINE					010	16 803 280	393 788
B-737-200	132	I	132	TZ/,290	61667	10,004,200	
BAC-1-11-200	48	1	48	77,395	1,498	3,/14,96U	11,904
	69	I	69	98,725	2,575	6,812,025	c/9,1/1
-30	246	ł	246	125,816	3,635	30,950,736	894,210
,							

184

Aircraft	Number in Certificated Carrier Fleet 12/31/74	Dorestic CRAF Fleet 12/31/74	Domestic WASP Fleet 12/31/74	Annual Psgr. Seat Mile Capacity Per Individual Aircraft (in thousands)	Annual Add'l. Cargo Ton-Mile Capacity Per Individual Aircraft (in thousands)	Total Annual Psgr. Seat- Mile Capacity of WASP Fleet (in thousands)	Total Annual Additional Ton-Mile Capacity of WASP Fleet
TURBOJET-4 ENGINE B-720 DC-8-10/20 -30 DC-8-50	6 11 28	1 I I	6 30 11 28	201,779 204,838 214,474 219,949	924 5,709 5,373	1,210,674 6,145,140 2,359,214 6,158,572	5,544 171,270 85,294 150,444
TURBOPROP-4 ENGINE L-188 A/C/F	19	i	19	81,585	742	1,550,115	14,098
<u>TURBOPROP-2 ENGINE</u> CV-580 -600 DHC-6 F-27 FH-227 YS-11	104 24 11 24 40 21 21	1111	164 24 11 21 21	42,566 31,536 8,705 29,492 33,014 44,457	1,129 1,025 116 24 293 	4,426,864 756,864 26,115 324,412 1,320,560 933,597	117,416 24,600 348 814 11,720
PISTON-2 ENGINE M-404 Totals	<u>12</u> 1,424	j	<u>12</u> 1,424	24,820	248	297,840 211,627,640	2,976 7,009,097

Table A.2 (cont'd)

\* \*\*\*

J. F. Laufer, <u>Civil Aeronautics Board War Air Service Program (WASP) Resource Report</u>, Bureau of Economics, Civil Aeronautics Board, 1975. SOURCE:

and the providence of the second

Table A.3

And a state of the state of the

### Characteristics of Representative Aircraft in the Commercial Fleet

Aircraft Manufacture	Boeing	Lockheed	McDonnell- Douglas	McDonnell- Douglas
Aircraft Model & Series	B-747-100	L-1011 TriStar	DC-9-30	DC-8-10/20
Number of Engines & Type	4 Engine Turbofan	3 Engine Turbofan	2 Engine Turbofan	4 Engine Turbojet
Quantity in U.S. Service	142	65	276	33
* Fuel Capacity (U.S. gal)	47,210	22,984	3,679	17,552
Passenger Limits	498	272	105	189
Cargo Wght. Limits (lbs)	174,689	85,500	26,700	34,500
Maximum Zero Fuel Weight	526,500	325,000	84,000	173,500
Maximum Landing Weight	564,000	358,000	98,100	199,500
Maximum Takeoff Weight	733,000	430,000	103,000	270,000
Wheel Track	36'1"	36 1 0"	16'6"	20110"
* Jet fuel weight/gallon =	= 6.7 lb/gal.			

186

1. 1. 2. 1.

のないないたいとうないでい

and the second

Reine Control of Contr



Fig. A.1 Field Length Requirements for the L-1011 Lockheed TriStar Aircraft. SOURCE: Lockheed-California Company, Lockheed 1011 Technical and Operations Summary, COA 1323, Burbank, California, June 1971.



SOURCE: Lockheed-California Company, Lockheed 1011 Technical and Operations Summary, COA 1323, Burbank, California, June 1971. 188

Table A.4

シアンというと言語

**\*** 

Performance Characteristics of Representative Airwraft

	B-747-100	L-1011	DC-9-30	DC-3-10/20
Maximum flight range from a 6,000 ft runway (Nautical Mile)	1,300	1,100	1,000	1,800
Maximum flight range from a 7,000 ft runway (Nautical Mile)	2,125	1,900	1,300	2,400
Runway length required for a 1,000 mile flight (ft) <sup>*</sup>	5,600	5,800	6,000	4,800
<ol> <li>Time of flight (hr)</li> <li>Fuel consumption (lb)</li> </ol>	2.4 58,000	2.4 40,000	2.4 15,800	2.4 33,000
Runway length required for a 500 mile flight (ft)	5,300	5,500	4,750	4.200
<ol> <li>Time of flight (hr)</li> <li>Fuel consumption (1b)</li> </ol>	1.4 33,000	1.3 22,000	1.3 8,800	1.3 19,000

\*These lengths are computed using the fuel (+ reserves) weight necessary to attain desired flight distance.

;

1 ALC: NO DECK ŝ

weighing over 12,500 lbs, and are controlled by the Department of Defense (CRAF) and Civil Aeronautics Board (WASP). The CAB publishes an annual resource report which lists the aircraft under its jurisdiction and data pertaining to the functions of the programs.

<u>CRAF</u> is composed of the most suitable U.S.-registered civil transport aircraft that are operationally capable of performing Department of Defense airlift. The Department of Defense is responsible for this program.

<u>WASP</u> is designed to provide for the maintenance of essential civil air routes and services. It also provides for the distribution and redistribution of air carrier aircraft among civil air transport carriers after the withdrawal of aircraft allocated to CRAF. The Civil Aeronautics Board is responsible for this program. ilite all have researed a state of the second of the second

Two thousand seventeen aircraft comprise the CRAF and WASP fleet; each program has definite functions and duties, and they are regulated independently. For massive air evacuation to occur, priority must be shifted to schedule a maximum effort for each program. Procedures need to be documented and placed under the duties of the program.

### A.3 Maintenance and Material Requirements

Information pertaining to the material requirements necessary to sustain operation of CRAF and WASP fleets were not available. Personal conversations with maintenance personnel associated with the aircraft were made to obtain an idea of the requirements necessary to maintain an aircraft in operation. They were asked to relate the operational stability of the aircraft and to estimate the parts and material requirewents necessary to maintain operational status. Eastern Airlines Maintenance, Atlanta, conversation concerning DC-9, ". . . if given 6 to 8 hours notice, we can carry enough maintenance materials with our aircraft to maintain 90% of our fleet for a period of 6 months, allowing the remaining 10% for cannibalizing. . ." Eastern Airlines Heavy Maintenance, Miami, concerning B-727, ". . . most dependable aircraft we have . . . estimate a material requirement of 1000 lbs. per aircraft to maintain operation for 90 days." To substantiate the opinions of these maintenance personnel, a check of Delta Airlines Maintenance Department was made which verified the estimates.

Estimates of support equipment necessary to maintain aircraft were given which ranged from 900-1200 lbs. In stating the conditions necessary to maintain the aircraft, the relocation of the aircraft from the target areas was proposed which implied the need to carry parts with them.

### A.4 Description of Aircraft

For the purposes of this report only one aircraft is described in detail from each grouping of jet classification in Tables A.1 and A.2. This information is listed in Tables A.3 and A.4. The representative aircraft are Boeing's 747, Lockheed's L-1011, and McDonnell-Douglas's DC-8 and DC-9.

and a start of the start of the

### A.5 Alternate Airfields

Certain alternatives to formal runways could exist. To use dry lake and river beds, i.e., Utah Salt Flats, little would have to be done in preparing their surfaces for the accommodation of aircraft mentioned in this report. Also, portions of interstate highways could be used to accomodate certain selected aircraft that meet the desired criteria.

Wheel track, landing and takeoff distances, weights, and turning radii must be such as to meet the limitations of each specific aircraft. Careful planning is necessary in this respect.

The Department of Transportation, Concrete Division, supplied information concerning the concrete thickness of the interstate highway system. The interstates are relatively constant in thickness. Unreinforced highways are 10-11" in thickness. Reinforced highways are 9" thick. Figure A.3 shows the thickness of concrete necessary to support certain aircraft at various weight loads.



いたがあいようないないが言語とれてたのとうと

and a second state of the second state of the

192

APPENDIX B MINIMIZATION OF TRANSPORT OF GRAIN

ないないであるが、それであるというできたなななないであるというです。

and a subscription of the subscription of the

### APPENDIX B. MINIMIZATION OF TRANSPORT OF GRAIN

The problem is to minimize the total ton-miles required to ship grain from BEAs (Business Economic Areas) which have a surplus to those which have a deficit.

### Let Z = total ton-miles, the objective function,

- x<sub>ij</sub> = tons shipped from BEA <u>i</u> to BEA <u>j</u>,
- d = distance from BEA i to BEA j (straight-line, centroid to centroid),
- $x_i = initial$  quanticy of grain in BEA <u>i</u>,
- R = quantity of grain reserve in BEA i after shipments are completed (assume shipments are completed in zero time, at least for the first look at the problem),
- $P_i = number of people in BEA i,$
- S = number of days of grain reserve in grain-rich BEAs,
- D = number of days of grain reserve to be built up in grainpoor BEAs.

Minimize  $Z = \sum_{i=j}^{\Sigma} d_{ij} x_{ij}$  subject to these constraints:

 $\sum_{j=1}^{L} x_{j} \leq X_{j} - R_{j} \geq 0$  (Total shipments from <u>i</u> do not deplete reserve in i) .

 $\sum_{i=1}^{\sum_{j=1}^{n}} x_{j} \geq 0$  (Sum of shipments to j are equal to reserve required in j minus initial quantity).

For this particular problem we work with the distribution of the relocated U.S. population according to the ADAGIO program. We assume the grain reserve per BEA is 60% of the 1973 grain production.

We do not allow shipments of grain FROM counties in which  $X_i < S \ge 0.001 \ge P_i$ , ( $X_i$  in tons); nor do we allow shipments TO counties in which  $X_j > D \ge 0.001 \ge P_j$ . We choose S = 365, and define  $R_i = S \times 0.001 \times P_i$  and  $R_j = D \times 0.001 \times P_j$ . Values are calculated for S and for D in mulitples of 7 days, up to 30 weeks.

Computer output lists source and deficit BEAs, distances, tons and ton-miles shipped.

and the second second second second second second second second

APPENDIX C SURVIVAL OF PETROLEUM REFINERIES

### APPENDIX C. SURVIVAL OF PETROLEUM REFINERIES

We wish to determine the number of weapons of reliability, r, required to reduce the U.S. POL refining capacity to some fraction, k, of its present capacity.

Let  $c_j$  = crude capacity of the j-th refinery in barrels per stream day (b/sd), and

 $T = \sum_{i}^{N} c_{i}$ , the total capacity of Nrefineries.

Refineries which produce only asphalt will not be considered.

Let S = kT represent the expected surviving refining cappcity, given by

$$S = \sum_{j=1}^{n} (1-r)^{n_j} c_j$$
, (1)

or and a distribution of a static static

where n<sub>j</sub> is the number of weapons delivered on the j-th refinery.

We assume the refineries are targeted so the expected surviving capacities are all equal, i.e.,

$$(1-r)^{n_{j}} c_{j} = (1-r)^{n_{j+1}} c_{j+1}$$
, etc. (2)

In this case,

$$S = N(1-r)^{n_j} c_j$$
, (3)

and the number of weapons to be delivered on the j-th refinery is given by

$$n_{j} = ln(\frac{S}{Nc_{j}})/ln (1-r) . \qquad (4)$$

NATURAL STREET, STREET

nter definitionen structur ander Brunn anderstanderstanden attractionen in enter externationen andere service a

a to she was the second of the second s

The total number of weapons is then obtained by summing over  $n_j$ . The application of this formulation to U.S. and Canadian refineries results in the curves shown in Fig. C.l, in which k = 0.03, the reliability, r, has values of 0.5, 0.6, and 0.8, and the bombing list has been rank-o.dered by decreasing capacity. If r = 0.6 (60% reliability), 800 weapons would be required to achieve an expectation of 3% survival

of the capacity of U.S. and Canadian refineries.

We assume that a finite CEP has negligible effect on these calculations because we are dealing with megaton yield weapons and targets of high vulnerability.





A TOMPY I TO THE

ORNL-5041 UC-41 - Health and Safety a Constituted (N

### Internal Distribution

1-3.	Central Research Library	18.	F. L. Culler
4.	ORNL-Y-12 Technical Library	19.	R. R. Davis
	Document Reference Section	20-119.	Emergency Technology Library
5-7.	Laboratory Records Department	120.	M. E. Fish
8.	Laboratory Records, ORNL R.C.	121.	R. E. Funderlic
9.	S. I. Auerbach	122.	C. M. Haaland
10.	J. A. Auxier	123.	R. F. Hibbs
11.	P. R. Barnes	124.	C. F. Holoway
12.	W. J. Boegly	125.	C. H. Kearny
13.	R. B. Burditt	126.	L. E. McNeese
14.	T. J. Burnett	127.	H. Postma
15.	C. V. Chester	128.	C. R. Richmond
16.	P. R. Coleman	129.	G. W. Westley
17.	G. A. Cristy		

### External Distribution

130. AFOSR Library, 1400 Wilson Boulevard, Arlington, VA 22209

- 131. Air University, Attn: Library, Maxweli Air Force Base, AL 26052
- 132. Assistant Secretary of the Army (R&D), Attn: Assistant for Research Washington, DC 20310
- 133. Ballistic Research Laboratory, Attn: Librarian, Aberdeen Proving Ground, MD 21005
- 134. Raymond J. Barbuti, Deputy Director, Office of Natural Disaster and Civil Defense, N.Y. State Dept. of Transportation, Bldg. 22, State Office Bldg. Campus, Albany, NY 12226
- 135. David W. Benson, RE(HV), Defense Civil Preparedness Agency, Washington, DC 20301
- 136. John E. Bex, DCPA Regional Director, Region 2, Federal Regional Center, Olney, MD 20832
- 137. John Billheimer, Systan, Inc., 343 Second Street, P. O. Box U, Los Altos, CA 94022
- 138. George F. Bing, Lawrence Livermore Laboratory, P. O. Box 808, Livermore, CA 94550
- 139. Bruce Bishop, DCPA Regional Director, Region 4, Federal Center, Battle Creek, MI 49016

140. Sarah A. Brown, 217-B Bim Street Apt., Carrboro, NC 27510

- 141. Stephen Brown, Stanford Research Institute, 333 Ravenswood Ave., Menlo Park, CA 94025
- 142. William M. Brown, Research Consultant, 19709 West Horseshoe Drive, Topanga, CA 90290
- 143. Arthur Broyles, Department of Physics, University of Florida, Gainesville, FL 32601
- 144. James O. Buchanan, Deputy Director for Research and Engineerning, DCPA, Washington, DC 20301
- 145. Edwin Bull, American Technical Assistance Corporation, 7655 Old Springhouse Rcad, McLean, VA 22101
- 146. Zolin Burson, Division of Operational Safety, USERDA, Washington, DC 20545
- 147. T. J. Byram, Data Services Branch, Survey Division, Statistical Reporting Service, U.S. Department of Agriculture, Washington, DC 20250
- 148. Arlin Carlson, State Division of Emergency Service, B-5 State Capital, St. Paul, MN 55155
- 149. Center for Naval Analysis, Attn: Head Studies Mgmt Group, 1401 Wilson Boulevard, Arlington, VA 22209
- 150. William W. Chenault, Human Sciences Research, Inc., Westgate Research Park, 7710 Old Springhouse Road, McLean, VA 22101
- 151. Chief, Joint Civil Defense Support Group, Office of Chief of Engineers, Department of the Army, Forrestal Building, 1F035, Washington, DC 20314
- 152. Chief of Naval Research, Washington, DC 20360
- 153. William K. Chipman, Deputy Assistant Director, Plans PO(DP), Defense Civil Preparedness Agency, Washington, DC 20301
- 154. John Christiansen, Dept. of Sociology, Brigham Young University, Provo, UT 84601
- 155 Civil Defense Technical Services Center, College of Engineering, Department of Engineering, University of Florida, Gainesville, FL 32601
- 156. Commander, Naval Facilities Engineering Command, Research and Development (Code 0322C), Department of the Navy, Washington, DC 20390

and and an and the statistic before the set of the statistical design of the set of the

the District Street and the set of the state of the

A REAL AREA STATE AND A STATE

- 158. Commander NMCSSC, Pentagon, Rm BE-685, Washington, DC 20310
- 159. Alvin M. Cruze, Research Triangle Institute, P. C. Box 12194, Research Triangle Park, NC 27709
- 160. James A. Davis, Emergency Preparedness Division, Agricultural Stabilization and Conservation Service, U.S. Department of Agriculture, South Building, Room 6628, Washington, DC 20250
- 161. L. J. Deal, Division of Operational Safety, USERDA, Washington, DC 20545
- 162-211. Defense Civil Preparedness Agency, Attn: Administrative Officer, Research and Engineering, Washington, DC 20301
- 212-223. Defense Documentation Center, Cameron Station, Alexandria, VA 22314
  - 224. Defense Intelligence Agency, Attn: DS-4A, Washington, DC 20301
  - 225. Defense Nuclear Agency, Commander Field Command, Sandia Base, Albuquerque, NM 87115
  - 226. Defense Nuclear Agency, Attn: Librarian, Washington, DC 20310
  - 227. Frances K. Dias, DCPA Regional Director, Region 6, Federal Regional Center, Building 710, Denver, CO 30225
  - 228. Henry Dickson, Dept. of Transportation, Federal Aviation Agency, National Flight Data Center, Room 431, 800 Independence Avenue, Washington, DC 20003
  - 229. Dikewood Corporation, 1009 Bradbury Drive, S.E., University Research Park, Albuquerque, NM 87106
  - 230. Director Development Center, Marine Corps Development and Education Command, Quantico, VA 22134
  - 231. Director Disaster and Defense Services Staff, Agricultural Stabilization & Conservation Service, U.S. Department of Agriculture, Washington, DC 20250
  - 232. Disaster Research Center, Ohio State University, 127 W. 10th Avenue, Columbus, OH 43210
  - 233. Rudolph J. Engelmann, Energy Research and Development Administration (DBER), Washington, DC 20545
  - 234. Charles R. Fisher, Oak Ridge Operations, U.S. Energy Research and Development Administration, P.O. Box E, Oak Ridge, TN 37830

- 235. William J. Flathau, Chief, Weapons Effects Laboratory, Waterways Experiment Station, U.S. Corps of Engineers, P.O. Box 631, Vicksburg, MS 39180
- 236. Roger Gibbons, System Development Corporation, 5827 Columbia Pike, Falls Church, VA 22041
- 237. Leon Goure, Director, Center for Advanced International Studies, P.O. Box 8123, University of Miami, Coral Gables, FL 33124
- 238. Cornelius Hall, President, Chemtree Corporation, Central Valley, NY 10917
- 239. William E. Hanzen, DCPA Regional Director, Region 8, Federal Regional Center, Bothell, WA 98011
- 240. David G. Harrison, DCPA Regional Director, Region 6, Federal Regional Center, Building 710, Denver, CO 80225
- 241. Edward L. Hill, Research Triangle Institute, P.O. Box 12194, Research Triangle Park, NC 27709
- 242. Jack Hirschleifer, Department of Economics, University of California, Los Angeles, CA 90024
- 243. Institute for Defense Analysis, 400 Army Navy Drive, Arlington, VA 22202
- 244. Institute for Energy Analysis, P.O. Box 117, Oak Ridge, TN 37830
- 245. Don Johnston, Research Triangle Institute, P.O. Box 12194, Research Triangle Park, NC 27709
- 246. T. K. Jones, Manager, Program & Product Evaluations, Product Development, Boeing Aerospace Co., P.O. Box 3999, Seattle, WA 98124
- 247. Reverend Daniel Jordahl, 402 Hill Street, Marshall, MN 56258
- 248. Herman Kahn, Hudson Institute, Croton-on-Hudson, NY 10520
- 249. Charles D. Kepple, 6912 Floyd Avenue, Springfield, VA 22150
- 250. Harold Knapp, Weapons Systems Evaluation Group, 400 Army-Navy Drive, Arlington, VA 22202
- 251. Albert Latter, R&D Associates, P.O. Box 3580, Santa Monica, CA 90405
- 252. Richard K. Laurino, Center for Planning & Research, 750 Welch Road, Stanford Frofessional Center, Palo Alto, CA 94304

「「「ないないないないないない」というないないできたいのできた」というとうというないないないです。

- 253. A. Longinow, IIT Research Institute, 10 West 35th Street, Chicago, IL 60616
- 254. Clarence C. Lushbau, A, Oak Ridge Associated Universities, P.O. Box 117, Oak Ridge, TN 37830
- 255. Clarence R. Mehl, Sandia Corporation, P.O. Box 5800, Albuquerque, NM 87115

and the second second

- 256. Carl F. Miller, Center for Planning and Research, Inc., P.O. Box 553, Moriarty, NM 87035
- 257. Peter Moulthrop, Lawrence Livermore Laboratory, P.O. Box 808, Livermore, CA 94550
- 258. William G. McMillan, McMillan Science Associates, Suite 901, Westwood Center Building, 1100 Glendon Avenue, West Los Angeles, CA 90024
- 259. Walter Murphey, Editor, JOURNAL OF CIVIL DEFENSE, P.O. Box 910, Starke, FL 32091
- 260. Jiri Nehnevajsa, Department of Sociology, University of Pittsburgh, 3117 Cathedral of Learning, Pittsburgh, PA 15213
- 261. National War College, Attn: Librarian, Fort Leslie J. McNair, Washington, DC 20315
- 262. John W. Nocita, Office of Preparedness, General Service Administration, Room 4229, ATGC, Washington, DC 20405
- 263. Peter G. Nordlie, Human Sciences Research, Westgate Industrial Park, P.O. Box 370, McLean, VA 22101
- 264. Edgar Parsons, System Sciences, Inc., P.O. Box 2345, Chapel Hill, NC 27514
- 265. President, Naval War College, Attn: Code 1212, Newport, RI 02840
- 266. James Pettee, Federal Disaster Assistance Administration, Room B-133, Department of Housing and Urban Development, 451 - 7th Street, S.W., Washington, DC 20410
- 267. Project Director, Engineer Strategic Studies Group, Office of Chief of Engineers, 6500 Brooks Lane, N.W., Washington, DC 20315
- 268. Ren Read, Assistant Director, Technical Services, Defense Civil Preparedness Agency, Washington, DC 20301
- 269. Research and Technical Support Division, ERDA, ORO, Oak Ridge, TN 37830

- 270. Bernice Rideout, Office of Civil Defense, Statehouse, Augusta, ME 04330
- 271. Robert M. Rodden, Stanford Research Institute, 333 Ravenswood Ave., Menlo Park, CA 94025
- 272. Murray Rosenthal, System Development Corporation, 2500 Colorado Ave., Santa Monica, CA 90406
- 273. Harvey G. Ryland, Mission Research Corporation, P.O. Drawer 719, Santa Barbara, CA 93102
- 274. Abner Sachs, Science Application Inc., 1651 Old Meadow Road, McLean, VA 22101
- 275. Leo Schmidt, Institute for Defense Analyses, 400 Army-Navy Drive, Arlington, VA 22202
- 276. W. W. Schroebel, Analysis and Evaluation Branch, Division of Biom.dical and Environmental Research, U.S. Energy Research and Development Administration, Washington, DC 20545
- 277. Bernard Shore, Lawrence Livermore Laboratories, P.O. Box 808 Livermore, CA 94550
- 278. George N. Sisson, Shelter Research Division, Defense Civil Preparedness Agency, Washington, DC 20301
- 279. Lewis V. Spencer, Center for Radiation Research, Radiation Theory Section, National Bureau of Standards, Washington, DC 20234
- 280. Burke Stannard, National Emergency Planning Establishment, Third Floor, Power B, Pearson Building, 125 Sussex, Ottawa 4, Ontario, Canada
- 281. H. A. Strack, Northrop Corporation, 1791 N. Fort Myer Drive, Arlington, VA 22209

- 282. Walmer E. Strope, Stanford Research Institute, 1611 North Kent Street, Arlington, VA 22209
- 283. C. J. Sullivan, Director, Civil Defense Department, Administration Bldg, Basement, 64 N. Union, Montgomery, AL 36104
- 284. Frank P. Szabo, Defense Research Establishment, Ottawa, Ontario, Canada KIA OZ 4

- 285. Lauriston Taylor, National Academy of Sciences, 2101 Constitution Avenue, N.W., Washington, DC 20418
- 286. Technical Library, USA-MERDC, Building 315, Fort Belvoir, VA 22060

- 287. Robert Thibodeau, Code 212, Transportation Systems Center, U.S. Department of Transportation, Kendall Square, Cambridge, MA 02142
- 288. Claude B. Thompson, DCPA Regional Director, Region 2, Federal Regional Center, Thomasville, GA 31792
- 289. Kyle Thompson, DCPA Regional Director, Region 5, Federal Regional Center, Denton, TX 76201
- 290. U.S. Army War College, Attn: Library, Carlisle Barracks, PA 17013
- 291. U.S. Energy Research and Development Administration, Technical Library, Washington, DC 20545
- 292. William H. Van Horn, URS Research Company, 155 Bovet Road, San Mateo, CA 94402
- 293. Luke Vortman, Sandia Corporation, P.O. Box 5800, Albuquerque, NM 87115
- 294. Lee Webster, Advanced Ballistic Missile Defense Agency, Huntsville Office, ABH-S, P.O. Box 1500, Huntsville, AL 35807
- 295. Alvin M. Weinburg, Institute for Energy Analysis, P.O. Box 117, Oak Ridge, TN 37830
- 296. Iram Weinstein, Stanford Research Institute, 333 Ravenswood Ave., Menlo Park, CA 94025
- 297. William J. Werner, Office of Policy Development & Research, Room 8158, Department of Housing and Urban Development, Washington, DC 20410
- 298. William White, Stanford Research Institute, 333 Ravenswood Ave., Menlo Park, CA 94025
- 299. Eugene P. Wigner, 8 Ober Road, Princeton, NJ 08540

and a second second second and a second s

- 300. Daniel Willard, Office of Operations Research, Office of Under Secretary of Army, Room 2E729, Pentagon, Washington, DC 20310
- 301. Walter Wood, Dikewood Corporation, 1009 Bradbury Drive, S.E., University Research Park, Albuquerque, NM 87106
- 302. Milton Wright, P.O. Box 12194, Research Triangle Institute, Research Triangle Institute Park, NC 27709
- 303. Allan R. Zenowitz, DCPA Regianal Director, Region 1, Federal Regional Center, Maynard, MA 01754
- 304-525. Given distribution as shown in TID-4500 under Health & Safety category (25 copies --- NTIS)

+ U.S. GOVERNMENT PRINTING OFFICE: 1976-748-189/376

and a start of the second s

REPORT DOCUMENTATION PAGE       END INSTRUCTIONS         REPORE COMPLETING FORM         INSTRUCTIONS         OBVE ACCESSION NO       3 RECIPIENTS CATALOG NUMBER         ORTACLESSION NO         INTLE LING FORM         INTLE LING FORM         SURVIVAL OF THE RELOCATED POPULATION OF         INTLE LING FORM         SURVIVAL OF THE RELOCATED POPULATION OF         INTLE CONSTRUCTIONS         INTLE CONSTRUCTION OF         INTLE CONSTRUCTION OF REPORT A PERIOD CONSTRUCTION         INTLE CONSTRUCTION OF REPORT DATE         CONTACLE ON CONSTRUCTION OF CONSTRUCTION         CONTACLE ON CONSTRUCTION OF CONSTRUCTION         INTLE CONSTRUCTION OF CONSTRUCTION         INTLE CONSTRUCTION OF CONSTRUCTION         OBSTRUCTION OF CONSTRUCTION OF CONSTRUCTION         OBSTRUCTION STATEMENT FORMERS         DEFENSE CON	Unclassified SECURITY CLASSIFICATION OF THIS PAGE (When When Entered)						
THEORY NUMBER       1       0 OUT ACCESSION NO. 2       RECEPTENTS CATALOG NUMBER         ORNL-5041       1       0 OUT ACCESSION NO. 2       RECEPTENTS CATALOG NUMBER         SURVIVAL OF THE RELOCATED POPULATION OF THE U.S. AFTER A NUCLEAR ATTACK       5       TYPE OF REPORT NUMBER         7       AUTHORY,       5       FINAL       5         Carsten M. Haaland, Conrad V. Choster, and Eugene P. Wigner       0 CONTACT OR CARLY STREET, TAKE       6         7       PERFORMING ORG REPORT NUMBER       10       PERFORMING ORG REPORT NUMBER         7       PERFORMING ORG REPORT NUMBER       10       PERFORMING ORG REPORT NUMBER         7       PERFORMING ORG ACCESSION NO.       10       CONTACT OR CARLY TRUET, TAKE         0ak Ridge NA CONTACL NAME AND ADDRESS       10       PERFORMING ORG NOT CONTACT.       10         0ak Ridge NT 37830       Work Unit 3539A       11       CONTACLEG TAKE AND ADDRESS       12       Recent PORTACT.         0ak Ridge NT D. C. 20301       12       Recent PORTACT.       NUMBER OF PARES       12         14       MONITORING AGENCY NAME & ADDRESS (If different hom Controlling Office)       15. SECURITY CLASS (of this report)         15       SUBPLEMENT OF FARES       10       12. SECURITY CLASS (of this report)         16       DISTRIBUTION STATEMENT (of the ADDRESS (	REPORT DOCUMENTATION PAGE	READ INSTRUCTIONS					
<ul> <li>TYLE INTRE INTERIOR SUBJECT OF THE RELOCATED POPULATION OF THAL</li> <li>SURVIVAL OF THE RELOCATED POPULATION OF THAL</li> <li>SURVIVAL OF THE RELOCATED POPULATION OF THAL</li> <li>SURVIVAL OF THE RELOCATED POPULATION OF THAL</li> <li>PERFORMED ONG REPORT NUMBER</li> <li>AUTHORIN,</li> <li>AUTHORIN,</li> <li>AUTHORIN,</li> <li>AUTHORIN,</li> <li>CARSTEN M. Haaland, Conrad V. Chester, and Eugene P. Wigner</li> <li>OK REGARN LUMINY FUNCTION THE AND ADDRESS</li> <li>OK REGARN LUMINY FUNCTION THE AND ADDRESS</li> <li>OK REGARN LUMINY FUNCTION THE AND ADDRESS</li> <li>OK RUGE, TH. 37830</li> <li>CONTRACT ON GRANT CUMPERT, TAIK</li> <li>ORA RIGGE, TN. 37830</li> <li>CONTRACT OFFICE HARE AND ADDRESS</li> <li>OK RUTH 13539A</li> <li>CONTRACT OFFICE HARE AND ADDRESS</li> <li>UNCLASS (of this report)</li> <li>Approved for public release; distribution unlimited.</li> <li>DISTRIBUTION STATEMENT (of the aburest antered in Block 20, if different ham Report)</li> <li>Nuclear War Transportation Postattack Fallout Survival Fallout Survival Fallout Surviva a nuclear attack of 6600 megatons (1444 weapons, current Soviet capability) if 90 million Americans would survive a nuclear attack of 6600 megatons (1444 weapons, current Soviet capability) if 90 million were evacuated from high-risk areas during the crisis period preceding the attack. This report presents solutions to problems of continuing survival of the surviving population of the United States in the face of threats from postattack</li></ul>	1 REPORT NUMBER 1 2 GOVT ACCESSION NO ORNL-5041	3 RECIPIENT'S CATALOG NUMBER					
<ul> <li>AUTHORIA, Carsten M. Haaland, Conrad V. Chester, and Eugene P. Wigner</li> <li>Correction Consultation Hume AND ADDRESS</li> <li>PERFORMATION HUME AND ADDRESS</li> <li>Oak Ridge National Laboratory / P.O. Box X</li> <li>Dak Ridge, TN 37830</li> <li>Work Unit 3539A</li> <li>Controlling Office HAME AND ADDRESS</li> <li>Befense Civil Preparedness Agency</li> <li>Washington, D.C. 20301</li> <li>REDAT DATE</li> <li>Monitorning Addensity different hom Controlling Office;</li> <li>Security CLASS for this report</li> <li>Unclassified</li> <li>Distribution Statement for this Report;</li> <li>Approved for public release; distribution unlimited.</li> <li>Distribution Statement for the abstract entered in Black 20, if different hom Report;</li> <li>Nuclear War</li> <li>Fallout</li> <li>Supplementaria on severe and if necessary and identify by block number;</li> <li>Nuclear State (100 million Americans would survive a nuclear attack of 6600 megators (1444 wapons, current Soviet capability) if 90</li> <li>Anstract 190 million Americans would survive a nuclear attack of 6600 megators (1444 wapons, current Soviet capability) if 90</li> <li>Anstract Continuing survival of the surviving population of the United States in the face of threats from postattack food shortages and survival provide for high-risk areas during the crisis period preceding the attack. This report presents solutions to problems of continuing survival of the surviving population of the United States in the face of threats from postattack food shortages and survival survival provide for high-risk areas during the crisis period preceding the attack. This report presents solutions to problems of continuing survival of the surviving population of the United States in the face of threats from postattack food shortages and presents solutions to problems</li> </ul>	SURVIVAL OF THE RELOCATED POPULATION OF THE U.S. AFTER A NUCLEAR ATTACK	S TYPE OF REPORT & PERIOD COVERED FINAL 6 PERFORMING ORG REPORT NUMBER					
<ul> <li>17. DISTRIBUTION STATEMENT (of the obsiract entered in Block 20, If different from Report)</li> <li>18 SUPPLEMENTARY NOTES</li> <li>19 KEY WORDS (Continue on reverse and if necessary and identify by block number) Nuclear War Transportation Postattack Fallout Survival Fogd</li> <li>20. ABSTRICT (Continue on reverse alde if necessary and identify by block number) At least 190 million Americans Would survive a nuclear attack of 6600 megatons (1444 weapons, current Soviet capability) if 90 million were evacuated from high-risk areas during the crisis period preceding the attack. This report presents solutions to problems of continuing survival of the surviving population of the United States in the face of threats from postattack food shortages and</li> </ul>	<ul> <li>7 AUTHOR(*,</li> <li>Carsten M. Haaland, Conrad V. Chester, and Eugene P. Wigner</li> <li>9 PERFORMING ORGANIZATION NAME AND ADDRESS Oak Ridge National Laboratory / P.O. Box X Oak Ridge. TN 37830</li> <li><sup>11</sup> CONTROLLING OFFICE NAME AND ADDRESS</li> <li>Defense Civil Preparedness Agency Washington, D.C. 20301</li> <li><sup>14</sup> MONITORING AGENCY NAME &amp; ADDRESS(II different from Controlling Office)</li> <li><sup>16</sup> DISTRIBUTION STATEMENT (of this Report)</li> <li>Approved for public release; distribution</li> </ul>	<ul> <li>CONTRACT OR GRANT NUMBER(s) Interagency Agreement No. AEC 40-31-64 and DCPA 01-74-C-0227</li> <li>PROGRAM ELEMENT PROJECT. TASK ARE 4 &amp; WORK UNIT NUMBERS</li> <li>WORK Unit 3539A</li> <li>REPORT DATE March 1976</li> <li>NUMBER OF PAGES 220</li> <li>SECURITY CLASS (of this report) Unclassified</li> <li>DECLASSIFICATION DOWNGRADING SCHEDULE</li> <li>tion unlimited.</li> </ul>					
<ul> <li><sup>19</sup> SUPPLEMENTARY NOTES</li> <li><sup>19</sup> KEY WORDS (Continue on reverse side if necessary and identify by block number) Nuclear War Transportation Postattack Fallout Survival Food</li> <li><sup>20</sup> ABSTRECT (Continue on reverse side if necessary and identify by block number) - At least 190 million Americans would survive a nuclear attack of 6600 megatons (1444 weapons, current Soviet capability) if 90 million were evacuated from high-risk areas during the crisis period preceding the attack. This report presents solutions to problems of continuing survival of the surviving population of the United States in the face of threats from postattack food shortages and -</li> </ul>	17. DISTRIBUTION STATEMENT (of the obstract entered in Block 20, if different from Report)						
<ul> <li><sup>19</sup> KEY WORDS (Continue on reverse side if necessary and identify by block number) Nuclear War Transportation Postattack Fallout Survival Foqd</li> <li><sup>20</sup> ABSTRACT (Continue on reverse side if necessary and identify by block number)         <ul> <li>ABSTRACT (Continue on reverse side if necessary and identify by block number)</li> <li><sup>20</sup> ABSTRACT (Continue on reverse side if necessary and identify by block number)</li> <li><sup>20</sup> ABSTRACT (Continue on reverse side if necessary and identify by block number)</li> <li><sup>20</sup> ABSTRACT (Continue on reverse side if necessary and identify by block number)</li> <li><sup>20</sup> At least 190 million Americans would survive a nuclear attack of 6600 megatons (1444 weapons, current Soviet capability) if 90 million were evacuated from high-risk areas during the crisis period preceding the attack. This report presents solutions to problems of continuing survival of the surviving population of the United States in the face of threats from postattack food shortages and</li> </ul> </li> </ul>	18 SUPPLEMENTARY NOTES						
of continuing survival of the surviving population of the United States in the face of threats from postattack food shortages and	<ul> <li><sup>19</sup> KEY WORDS (Continue on reverse aide if necessary and identify by block number, Nuclear War Transport Postattack Fallout Survival Food</li> <li><sup>20</sup> ABS TRACT (Continue on reverse aide if necessary and identify by block number) At least 190 million Americans would s of 6600 megatons (1444 weapons, current Sov million were evacuated from high-risk areas preceding the attack. This report presents</li> </ul>	ation urvive a nuclear attack iet capability) if 90 during the crisis period solutions to problems					
	of continuing survival of the surviving pop States in the face of threats from postatta	ulation of the United ck food shortages and					

,

Unclassified SECURITY CLASSIFICATION OF THIS PAGE (When Date Entered)

Unclassified

Contra and

APPROX.I

SECURITY CLASSIFICATION OF THIS PAGE(When Date Entered)

BLOCK - 20 (continued)

fallout radiation hazards. Reserves of food, transportation capacity, and fuel would survive the attack to provide more than adequate capability to feed the entire population until the first harvest after the attack.

### Unclassified SECURITY CLASSIFICATION OF T.: PAGE (When Data Entered)

うためたいとないない

SURVIVAL OF THE RELOCATED POPULATION OF THE U.S. AFTER A NUCLEAR ATTACK - Unclassified - OAK RIDGE AATIONAL LABORATORY - March 1976 - 220 pages - Interagency Agree- ment AEC 40-31-64 and DCPA 01-74-C-0227, Work Unit 3539A	Ar lessr 190 million Americans could continue .	At least 190 million Americans would survive a nuclear attack of 6600 megatons (1444 weapons, current Soviet capability) if 90 million were evacuated from high-risk areas during the crisis period preceding the attack. This report presents solutions to problems of continuing <u>sur- vival</u> of the surviving population of the United States in the face of threats from <u>postattack food</u> shortages and <u>fallout</u> radiation hazards. Reserves of food, transpor- tation capacity, and fuel would survive the attack to provide more than adequate capability to feed the entire population until the first harvest after the attack.	SURVIVAL OF THE RELOCATED POPULATION OF THE U.S. AFTER A NUCLEAR ATTACK - Unclassified - OAK RIDGE NATIONAL LABORATORY - March 1976 - 220 pages - Interagency Agree- ment AEC 40-31-64 and DCPA 01-74-C-0227, Work Unit 3539A	At least 190 million Americans would survive a nuclear attack of 6600 megatons (1444 weapuns, current Soviet capability) if 90 million were evacuated from high-risk areas during the crisis period preceding the attack. This report presents solutions to problems of continuing <u>sur- vival</u> of the surviving population of the United States in the face of threats from postattack food shortages and <u>fallout</u> radiation hazards. Reserves of food, <u>transpor- tation</u> capacity, and fuel would survive the attack to provide more than adequate capability to feed the entire population until the first harvest after the attack.
SURVIVAL OF THE RELOCATED POPULATION OF THE U.S. AFTER A NUCLEAR ATTACK - Unclassified - OAK RIDGE NATIONAL LABORATORY - March 1976 - 220 pages - Interagency Agree- ment AEC 40-31-64 and DCPA 01-74-C-0227, Work Unit 3539A	At least 190 million Americans would survive a nuclear	At teact 100 million where the average a nuclear attack of 6600 megatons (1444 weapons, current Soviet capability) if 90 million were evacuated from high-risk areas during the crisis period preceding the attack. This report presents solutions to problems of continuing survival of the surviving population of the United States in the face of threats from postattack food shortages and fallout radiation hazards. Reserves of food, transportation capacity, and fuel would survive the attack to provide more than adequate capability to feed the entire population until the first harvest after the attack.	SURVIVAL OF THE RELOCATED POPULATION OF THE U.S. AFTER A NUCLEAR ATTACK - Unclassified - OAK RIDGE NATIONAL LABORATORY - March 1976 - 220 pages - Interagency Agree- ment AEC 40-31-64 and DCPA 01-74-C-0227, Work Unit 3539A	At least 190 million Americans would survive a nuclear attack of 6600 megatons (1444 weapons, current Soviet capability) 1f 90 million were evacuated from high-risk report presents solutions to problems of continuing sur- vival of the surviving population of the United States in the face of threats from postattack food shortages and fallout radiation hazards. Reserves of foud, transpor- tation capacity, and fuel would survive the attack to provide more than adequate capability to feed the entire population until the first harvest after the attack.

•

T